

## LUND UNIVERSITY

### **Essays on Institutions and Institutional Change**

Khalid, Usman

2016

#### Link to publication

*Citation for published version (APA):* Khalid, U. (2016). *Essays on Institutions and Institutional Change*. [Doctoral Thesis (compilation), Department of Economics].

Total number of authors: 1

#### General rights

Unless other specific re-use rights are stated the following general rights apply:

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights. • Users may download and print one copy of any publication from the public portal for the purpose of private study

or research.

You may not further distribute the material or use it for any profit-making activity or commercial gain
You may freely distribute the URL identifying the publication in the public portal

Read more about Creative commons licenses: https://creativecommons.org/licenses/

#### Take down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

LUND UNIVERSITY

**PO Box 117** 221 00 Lund +46 46-222 00 00

## Essays on Institutions and Institutional Change

Usman Khalid

Lund Economic Studies



Number 191

Essays on Institutions and Institutional Change

# Essays on Institutions and Institutional Change

Usman Khalid



DOCTORAL DISSERTATION by due permission of the School of Economics and Management, Lund University, Sweden. To be defended at Holger Crafoord EC3 210, Lund on February 2, 2016 at 10:15.

Faculty opponent

Markus Taube, The University of Duisburg-Essen

| Organization  | Document name           |  |
|---|-------------------------|--|
| LUND UNIVERSITY   | DOCTORAL DISSERTATION   |  |
| Department of Economics, P.O. Box 7082                              | Date of issue           |  |
| S 220 07 Lund, Sweden   | February 2, 2016        |  |
| Author(s): Usman Khalid   | Sponsoring organization |  |
| Title and subtitle: Essays on Institutions and Institutional Change |                         |  |

on Institutions and Institutional Change

Abstract: This thesis consists of three essays discussing institutional change and its effects. It distinguishes between economic and political institutions, and highlights the need for broader empirical efforts that consider interactions among different types of institutions.

The first essay titled Effect of Trade and Political Institutions on Economic Institutions explores the relationship between trade and economic institutions under different political regimes. The findings suggest that the effect of trade on economic institutions reduces significantly in the presence of extractive political institutions. In contrast, there is a positive and significant effect of trade on economic institutions in countries with relatively less extractive and democratic institutions.

The second essay Catch-up in Institutional Quality: An Empirical Assessment, studies the process of catchup in institutional quality by asking whether a catch-up in institutional quality has occurred across countries. The results indicate that a catch-up in economic institutional quality has occurred and that most countries with weak economic institutions have a higher rate of change than that of countries with strong economic institutions. In contrast, for political institutions, the catch-up process lasts only a few vears.

The third essay Firm Ownership and Provincial CO<sub>2</sub> emissions in China, investigates whether China's move towards a market economy has affected its CO2 emissions by considering the effect of firm ownership on provincial emissions. The hypothesis is that private firms are economically more efficient than non-private firms resulting in fewer CO2 emissions by the private firms. The results suggest that capital growth is the main driver of emissions growth and that private firm capital is more energyefficient than capital employed in non-private firms.

Key words: Institutions, Institutional change, Economic institutions, Political institutions, Firm ownership

| Classification system and/or index terms (if any): P14, P16, P12, P21, P28, E02, F14, F13, O43, |  |
|---|--|
|---|--|

| Supplementary bibliographical information                   |                         | Language English  |
|---|-------------------------|---|
| ISSN and key title: 0460-0029 Lund Economic Studies no. 191 |                         | ISBN:978-91-7623-638-3 (print)<br>978-91-7623-639-0 (pdf) |
| Recipient's notes   | Number of pages 127     | Price   |
|   | Security classification | ·   |

I, the undersigned, being the copyright owner of the abstract of the above-mentioned dissertation, hereby grant to all reference sourcespermission to publish and disseminate the abstract of the above-mentioned dissertation.

Signature

Usman Khalid

Date 16-12-2015

# Essays on Institutions and Institutional Change

Usman Khalid



Copyright Usman Khalid

School of Economics and Management Department of Economics

ISBN 978-91-7623-638-3 (print) ISBN 978-91-7623-639-0 (pdf) ISSN 0460-0029

Printed in Sweden by Media-Tryck, Lund University Lund 2015



To my parents for their love, endless support and encouragement.

## Content

| Acknowledgement   | 11       |
|---|----------|
| Chapter 1 Introduction  | 13       |
| Overview of the Chapters<br>Trade, Political Institutions and Economic Institutions<br>Catch-up in Institutional Quality: An Empirical Assessment | 15<br>16 |
| Firm Ownership and Provincial CO <sub>2</sub> emissions in China  |          |
| References  | . 19     |
| Chapter 2 Trade, Political Regimes and Economic Institutions  | 23       |
| Introduction  | 23       |
| Trade and political institutions  | 24       |
| Methodology   | 27       |
| Data  | 30       |
| Dependent variable  | 30       |
| Explanatory variables   |          |
| Control variables   | 31       |
| Results   | 33       |
| Robustness checks   | 39       |
| Conclusions   | 44       |
| References  | 46       |
| Appendix  | 50       |
| Chapter 3 Catch-up in Institutional Quality: An Empirical Assessment  | 53       |
| Introduction  | 53       |
| Theoretical background  | . 56     |
| Catch-up Perspective  | . 56     |
| Lagging-behind Perspective  | 57       |

| Mixed Account  | 58  |
|--|-----|
| Methodology  | 60  |
| Data   | 63  |
| Descriptive statistics   | 65  |
| Results and discussion   | 68  |
| Economic institutions  | 68  |
| Political Institutions   | 76  |
| Conclusion   | 81  |
| References   | 83  |
| Appendix   | 89  |
| Chapter 4 Firm Ownership and Provincial CO2 emissions in China | 95  |
| Introduction   | 95  |
| Firm Ownership and the Carbon Footprint                        | 97  |
| Firm Ownership and the Carbon Footprint: A Theoretical         |     |
| Background   |     |
| Firm Ownership and the Carbon Footprint: The Case of China     | 99  |
| Measuring the Size of the Private Sector                       | 101 |
| Empirical Analysis   | 102 |
| Econometric Model  |     |
| Data Sources and Descriptive Statistics                        | 105 |
| Regression Results and Policy Conclusions                      | 112 |
| Conclusion   | 117 |
| References   | 119 |
| Appendix   | 125 |
| Appendix A: Total Factor Productivity Estimation               |     |
| Appendix B: Variable Description and Data Sources              |     |

# Acknowledgement

Doing a Ph.D. is a journey that cannot be completed without the support of several people, who lend a hand when the going gets tough, who help relieve the occasional boredom with their warmth and wit, and who continue to believe in you when self-doubt creeps in. My journey to Ph.D. has been made easier by the support of several people who I have been fortunate enough to meet along the way over the last five years. Mentioning all of those people here would require reliving the whole five years through a narration that could take up several pages. Therefore, I would express my heartfelt gratitude to everyone who has touched my life over the years.

That being said, a few people deserve special gratitude without whom this thesis could not have been completed. I am extremely grateful and indebted to my main supervisor, Sonja Opper, for her continuous encouragement and invaluable comments and feedbacks. You are an inspiration to me since the very beginning, and your lectures in Development Economics course developed my interest in institutional economics. Although the journey was not easy and often frustrating, the motivation and mentorship provided by you helped me to overcome the struggle I went through during the course of my Ph.D. I am forever thankful for your patience, understanding, and support.

I am also very thankful and indebted to Fredrik N.G. Andersson, my assistant supervisor, for his continuous support and timely guidance. Thank you for being always reachable for consultation. Your expert advice on numerous issues including technical econometric problems has immensely helped me in writing the thesis. It is always an experience to talk with you about any academic and nonacademic subject. I also appreciate the chance Sonja and Fredrik provided me to co-author a paper with them. I have learned a great deal from this experience.

In addition to my official supervisors, I would also like to thank Andreas Bergh for acting as a discussant on my final seminar and providing valuable comments. My colleagues who are also supervised by Sonja and Fredrik: <sup>17</sup>asia, Ross, and Karl also contributed with their valuable comments during our regular group meetings. A special thanks to Ali Gibran Siddiqui and Hassan Mazhar Goreja for reading part of my work every now and then and giving me valuable feedback.

I have benefited a lot from the social and academic life at the Department of Economics. My frequent interactions with other faculty members at the Department have been very fruitful. In particular, I would like to thank David Edgerton, Christer Gunnarsson, Klas Fregert, Jerker Holm, Pontus Hanson, Tommy Andersson, Fredrik Sjöholm, Hans Falck, Farrukh Javed, Lina Maria Ellegård and Lu Liu. I would also like to thank the administrative staff at the Department especially, Nathalie Stenbeck, Jenny Wate, and Mariana Anton, for helping me in administrative matters. Thank you to all the participants of the weekly badminton games especially Jens Gudmundsson, Erik Wengström, Anders Vilhelmsson and Carl Hampus Lyttkens.

I would also thank all my fellow Ph.D. students with whom I had a wonderful time at the Department. I am especially grateful to Milda for being an understanding roommate and a friend; to Valeriia for all the stress relieving chats we had over the years. I have enjoyed many friendly chats with Anna Welander, Veronika Lunina, Hassan Sabzevari, Björn Thor Arnarson, Claes Ek, Alemu Chala, Yana Pryymachenko, Mehmet Caglar Kaya, Thomas Hofmarcher, Simon Reese, Caren Nielsen, and Margaret Samahita.

Finally, I would like to express my greatest gratitude to my family. My caring and loving parents and my siblings who have always supported and encouraged me throughout this journey. Without their support and encouragement, I would not have been able to achieve this milestone. I am deeply grateful to my wife, Asma, who stood strong with me through this thesis, and provided much support.

Lund, Sweden, December 2015

Usman Khalid

# Chapter 1 Introduction

The New Institutional Economics (NIE) perspective highlights the importance of institutions in shaping economic outcomes. The basic premise of NIE is that economic outcomes realized by a society are also a function of the institutions put in place, which create incentives and impose constraints that in turn motivate agent's behavior. Numerous empirical studies have established that institutions are critical for economic development (e.g., see Acemoglu, Johnson, & Robinson, 2001; Easterly & Levine, 2003; Hall & Jones, 1999; Knack & Keefer, 1995; Mauro, 1995; Rodrik, Subramanian, & Trebbi, 2004). A large body of literature has explored the relevance of institutions in determining international trade, foreign direct investments, financial development, and efficiency (Adkins, Moomaw, & Savvides, 2002; Busse & Hefeker, 2007; Levchenko, 2007; La Porta, Lopez-de-Silanes, Shleifer, & Vishny, 1998). Still further, institutions have been identified as substantive predictors of macroeconomic stability, and entrepreneurship (Acemoglu, Johnson, Robinson, & Thaicharoen, 2003; Aidis, Estrin, & Mickiewicz, 2010; Simón-Moya, Revuelto-Taboada, & Guerrero, 2014; Stephen, Urbano, & van Hemmen, 2005).

The quality of these institutions varies significantly across countries. Shirley (2005) highlights four explanations for these differences. The first is a legacy of poor institutions inherited from colonizers (La Porta, Lopez-de-Silanes, Shleifer, & Vishny, 1999; North, 1990). The second explanation refers to differences in initial resource endowments that led to the development of distinct institutional development pathways (Acemoglu & Johnson, 2002; Acemoglu, Johnson, & Robinson, 2005b; Acemoglu & Robinson, 2012). The third is that a lack of political competition may have encouraged leaders to build institutions to serve their own interests rather than the interests of the people (Nugent & Robinson, 2010). The final explanation is that distinct cultural and ideological beliefs and norms may have discouraged the development of anonymous markets and market-supporting institutions such as property rights, contract laws and investor protection (Knack & Keefer, 1997; North, 1994, 2006). Although these

theoretical explanations help explain the deeper causes of institutional differences, they offer limited guidance to improve local institutions in the short and medium terms.

Another strand in the literature interprets institutional quality as a by-product of economic development and shifting policy demands (Chang, 2011). The modernization hypothesis assumes that the relative benefits of institutions increase with the level of economic development (e.g., see Acemoglu, Johnson, Robinson, & Yared, 2009; Chang, 2011; Lipset, 1959). Along this reasoning, the causal chain predicts that accelerating economic development and increased wealth create corresponding demands for high-quality institutions (e.g., demands for political institutions with greater transparency and accountability), nurture agents who introduce change dynamics, and finally provide resources to introduce and maintain institutional change (Chang, 2011). Similarly, the grand transition perspective interprets "development" as a process of steady economic growth leading to broad institutional transitions (Paldam & Gundlach, 2008). Naturally, other economic factors, such as economic openness and human capital, also contribute and shape these complex processes of institutional change (Al-Marhubi, 2005; Alonso & Garcimartín, 2013; Bergh, Mirkina, & Nilsson, 2014; Bhattacharyya, 2012; Islam & Montenegro, 2002; Nicolini & Paccagnini, 2011).

Most empirical studies aiming to document institutional change do not distinguish between different types of institutions. Rather, they treat different institutions as alternative proxies for overall institutional quality (Acemoglu & Johnson, 2005). However, North's (1990) definition of institutions as "rules of the game" that "structure incentives in human exchange, whether political, social, or economic" distinguishes between different types of institutions that are unlikely to be equally open to change. Economic and social institutions determine the incentives and constraints of economic and social actors, and political institutions determine the incentives and constraints of key actors in the political sphere. In non-democratic systems, these key actors often have ultimate control over the distribution of political rights (Acemoglu, Johnson, & Robinson, 2005a).

This taxonomy provides a useful approach to study the complex dynamics and contingencies of institutional change. For instance, how do modifications of one distinct institution affect other institutions? Does spillover have positive effects or counterbalancing tendencies? Which institutions deserve priority? Finally, how can contingency effects be exploited to advance institutional change? Inspired by these open questions, this thesis distinguishes between polirical and economic institutions as two types of institutions that likely differ in their malleability and are also likely to mutually affect related change-dynamics. Political institutions form the basis of economic institutions (Acemoglu et al., 2005a; Acemoglu & 14

Robinson, 2012; North, 1990, 1991, 1994). Thus, Acemoglu et al. (2005a) argue that political institutions define the constraints limiting the use of political power, and they condition the distribution of resources (and rents) within the economy, thereby determining which group holds the de jure political and economic power in a given society. Political institutions play a crucial part in the institutional system. Any change in political institutions can modify the distribution of economic resources and the structure of economic institutions (Acemoglu et al., 2005a). Various historical illustrations demonstrate how the unequal distribution of de facto economic power combined with extractive de jure political institutions can lead to the persistence of socially inefficient economic institutions over time (Acemoglu & Robinson, 2012).

The theoretical and conceptual framework underscores the importance of distinguishing between several types of institutions and the interactions among them. Relying on North's (1990) broader definition of institutions as "rules of the game" provides little insight on the process of institutional change. This basic definition helps identify the importance of institutions in the economic system, but it requires further development to understand their specific functions in relation to different parts of an economy. In particular, institutions should be delineated in terms of their specific effects on economic outcomes to establish how individual institutions (or institutional groupings) cause (either directly or indirectly) or hinder growth. Rodrik (2000, p. 2) summarizes the question as follows: "Which institutions matter and how does one acquire them?" This question highlights the need for broader empirical efforts that consider interactions among different types of institutions. Any discourse on institutions and institutional change should also consider the mutually reinforcing and dynamic relations of well measured, distinct political and economic institutions. As Acemoglu and Johnson (2005) state, it is necessary to "unbundle" institutions to explain how and why they develop.

### Overview of the Chapters

#### Chapter 2: Trade, Political Institutions and Economic Institutions

This chapter contributes to the growing literature on identifying the determinants of economic institutions. Recent empirical studies confirm a positive relation between international trade and the quality of domestic economic institutions (Bergh et al., 2014; Bhattacharyya, 2012; Nicolini & Paccagnini, 2011).

However, the effect of trade on economic institutions is likely to vary with different political regimes. From a short-term perspective, trade regimes and domestic rules of international trade participation reflect the priorities of the ruling class and, in this sense, they are political constructs. In an authoritarian regime, the political elite can effectively restrict broader participation of the public in trade and can use regulations, quotas, and licenses to manipulate the market structure and impede market access (Aidt & Gassebner, 2010). Increased trade is therefore not necessarily associated with better economic institutions. I argue that institutional effects stemming from international trade are unlikely to be independent of the prevailing political regime. Centralized power in the hands of a small group of political elites is likely to have moderating effects on the assumed link between trade and economic institutional quality. Thus, the effect of international trade on economic institutions will be attenuated in countries with more extractive political institutions compared to countries with democratic institutions.

Using panel data from 138 countries from the year 1984 to 2010, this chapter reports a positive link between the level of international trade and the quality of economic institutions. However, in the presence of extractive political institutions, the effect of trade on economic institutions decreases significantly and even becomes negative. The results are in line with the theoretical predictions and reinforce the conditional relationship between trade and economic institutions. These results appear to confirm that authoritarian regimes effectively restrict broader participation of the public in trade and may also resort to use of regulations, quotas, and licenses to manipulate the market structure and impede market access, reducing the effect of trade openness on economic institutions. The findings underscore the role played by political institutions in determining economic institutions and calls for the inclusion of political institutions in the analysis of economic institutions.

#### Chapter 3: Catch-up in Institutional Quality: An Empirical Assessment

This chapter contributes to the literature studying the dynamics of institutional change and the differences in change patterns among political and economic institutions. This chapter deviates from examining the determinants of institutions and instead focuses on institutional change. The central question is whether there is a process of catch-up in institutional quality across countries and whether there are any differences in how economic and political institutions change. To answer this question, this chapter investigates the evolution of these institutions over time. A key objective of reforms aimed at promoting economic 16

growth in developing countries has been to improve institutional quality, as dysfunctional and ineffective institutions and weak governance "is increasingly seen to be at the heart of the economic development challenge" and "building effective and accountable institutions is arguably the core challenge for sustainable poverty reduction" (World Bank, 2000, p. 1). However, institutional development is a challenging endeavor, as new institutions must comply with the existing social norms and face opposition from elites (Acemoglu et al., 2005a; Evans, 2004). Moreover, economic institutions are more likely to change, compared to political institutions, as the interests of elites align with changing economic institutions vis-à-vis political institutions (Nye, 2011; Olson, 1993). Therefore, it is fitting to ask whether a catch-up in institutional quality has occurred and whether the catch-up process differs between economic and political institutions.

The chapter uses a novel testing method to test for catch-up in institutional quality across countries. The study uses data on 81 countries from 1985 to 2010 and three different measures of institutional quality that capture both political and economic dimensions of institutions. The results suggest that economic institutions in most countries tend to change in similar directions. Moreover, evidence supports a catch-up in institutional quality, as most countries with weak institutions have a higher rate of change than that in countries with strong institutions. In contrast, for political institutions, the catch-up process was short-lived, lasting only a few years. The results indicate that the pattern of change for economic and political institutions is different and underscores the need to incorporate the differentiation between economic and political institutions in future studies.

#### Chapter 4: Firm Ownership and Provincial CO2 emissions in China

(Co-authored with Fredrik N. G. Andersson and Sonja Opper)

Chapter 4 contributes to the literature on the effects of public and private ownership of firms on environmental quality by studying the effects of regional differences in ownership structure on overall provincial carbon dioxide ( $CO_2$ ) emissions in China. Since the 1980s, China has emerged as one of the world's largest economies and the main emitter of  $CO_2$ . Many contend that China's leadership has prioritized economic growth over environmental concerns throughout most of its reform, causing severe air, water, and and pollution (He, Lu, Mol, & Beckers, 2012). However, China has also experienced a gradual capitalist transformation from a fully state-owned economy to a hybrid economy largely relying on private production and mixed ownership forms (Nee & Opper, 2012). The question remains whether and to what extent the country's openness to capitalist forms of production has reinforced or possibly mitigated accumulating environmental costs.

To model the effect of this capitalist transformation on environmental quality, the chapter uses Chinese provincial data from 1992 to 2012. The approach followed here is to decompose the short-run and long-run cause of emissions into scale, energy, and carbon intensity components. The results show that capital growth is the main driver of emissions growth and that private-firm capital is more energy efficient than capital employed in non-private firms. This translates into emission growth between 3 and 4 percent per year in the long-term, had private firms been as inefficient as non-private firms. This result emphasizes that marketoriented reforms and general firm efficiency can decrease future CO<sub>2</sub> emission growth. The competitive pressure on private firms, as compared to non-private firms, is likely to be the main driver of the negative long-run correlation between private enterprises and emissions. The results also show that continued structural changes from an agricultural economy to a modern industrialized economy, coupled with an active environmental policy (i.e. regulation of carbon price) may reduce overall emissions growth close to zero, despite economic growth in excess of 5 percent per year. These insights offer valuable lessons for many developing countries, which continue to rely on large state-owned sectors. Economic development and environmental concerns can be coupled into a win-win situation through reforms that directly enhance productivity and competition and indirectly decrease environmental degradation.

### References

- Acemoglu, D., & Johnson, S. (2002). Reversal of Fortune: Geography and Institutions in the Making of the Modern World Income Distribution. *The Quarterly Journal of Economics*, *117*, 4.
- Acemoglu, D., & Johnson, S. (2005). Unbundling Institutions. *journal of Political Economy*, 113(5).
- Acemoglu, D., Johnson, S., & Robinson, J. (2001). The colonial origins of comparative development: an empirical investigation. *American Economic Review*, 91(5), 1369-1401.
- Acemoglu, D., Johnson, S., & Robinson, J. (2005a). Institutions as a Fundamental Cause of Long-Run Growth. In A. Philippe & N. D. Steven (Eds.), *Handbook of economic growth* (Vol. Volume 1, Part A, pp. 385-472): Elsevier.
- Acemoglu, D., Johnson, S., & Robinson, J. (2005b). The Rise of Europe: Atlantic Trade, Institutional Change, and Economic Growth. *American Economic Review*, 95(3), 546-579.
- Acemoglu, D., Johnson, S., Robinson, J., & Thaicharoen, Y. (2003). Institutional causes, macroeconomic symptoms: volatility, crises and growth. *Journal of Monetary Economics*, 50(1), 49-123.
- Acemoglu, D., Johnson, S., Robinson, J. A., & Yared, P. (2009). Reevaluating the modernization hypothesis. *Journal of Monetary Economics*, 56(8), 1043-1058.
- Acemoglu, D., & Robinson, J. (2012). Why Nations Fail: The Origins of Power, Prosperity, and Poverty. *Crown Business, New York NY*.
- Adkins, L. C., Moomaw, R. L., & Savvides, A. (2002). Institutions, freedom, and technical efficiency. *Southern Economic Journal*, 92-108.
- Aidis, R., Estrin, S., & Mickiewicz, T. (2010). Institutions, finance and the level of development: the impact on entrepreneurship in transition. *Review of economics and institutions, 1*(1).
- Aidt, T. S., & Gassebner, M. (2010). Do Autocratic States Trade Less? *The World Bank Economic Review, 24*(1), 38-76.
- Al-Marhubi, F. A. (2005). Openness and governance: Evidence across countries. Oxford Development Studies, 33(3-4), 453-471.
- Alonso, J. A., & Garcimartín, C. (2013). The determinants of institutional quality. More on the debate. *Journal of International Development*, 25(2), 206-226.
- Bergh, A., Mirkina, I., & Nilsson, T. (2014). Globalization and Institutional Quality— A Panel Data Analysis. *Oxford Development Studies*, *42*(3), 365-394.

- Bhattacharyya, S. (2012). Trade liberalization and institutional development. *Journal of Policy Modeling*, *34*(2), 253-269.
- Busse, M., & Hefeker, C. (2007). Political risk, institutions and foreign direct investment. *European journal of political economy*, 23(2), 397-415.
- Chang, H.-J. (2011). Institutions and economic development: theory, policy and history. *Journal of Institutional Economics*, 7(04), 473-498.
- Easterly, W., & Levine, R. (2003). Tropics, germs, and crops: how endowments influence economic development. *Journal of Monetary Economics*, 50(1), 3-39.
- Evans, P. (2004). Development as institutional change: the pitfalls of monocropping and the potentials of deliberation. *Studies in comparative international development*, *38*(4), 30-52.
- Hall, R. E., & Jones, C. I. (1999). Why do some countries produce so much more output per worker than others? *The Quarterly Journal of Economics*, 114(1), 83-116.
- He, G., Lu, Y., Mol, A. P., & Beckers, T. (2012). Changes and challenges: China's environmental management in transition. *Environmental Development, 3*, 25-38.
- Islam, R., & Montenegro, C. E. (2002). What determines the quality of institutions? World Bank Policy Research Working Paper(2764).
- Knack, S., & Keefer, P. (1995). Institutions and economic performance: cross-country tests using alternative institutional measures. *Economics & Politics*, 7(3), 207-227.
- Knack, S., & Keefer, P. (1997). Does social capital have an economic payoff? A crosscountry investigation. *The Quarterly Journal of Economics*, 1251-1288.
- La Porta, R., Lopez-de-Silanes, F., Shleifer, A., & Vishny, R. W. (1998). Law and Finance. *The Journal of Political Economy, 106(6)*, 1113-1155.
- La Porta, R., Lopez-de-Silanes, F., Shleifer, A., & Vishny, R. (1999). The quality of government. *Journal of Law, Economics, and organization, 15*(1), 222-279.
- Levchenko, A. A. (2007). Institutional quality and international trade. *The Review of Economic Studies*, 74(3), 791-819.
- Lipset, S. M. (1959). Some social requisites of democracy: Economic development and political legitimacy. *American political science review*, *53*(01), 69-105.
- Mauro, P. (1995). Corruption and growth. *The Quarterly Journal of Economics*, 681-712.
- Nee, V., & Opper, S. (2012). *Capitalism from below: Markets and institutional change in China:* Harvard University Press.
- Nicolini, M., & Paccagnini, A. (2011). Does Trade Foster Institutions? *Review of economics and institutions*, 2(2).

- North, D. C. (1990). *Institutions, institutional change and economic performance:* Cambridge university press.
- North, D. C. (1991). Institutions. The Journal of Economic Perspectives, 5(1), 97-112.
- North, D. C. (1994). The historical evolution of polities. *International Review of Law* and Economics, 14(4), 381-391.
- North, D. C. (2006). *Understanding the process of economic change*: Academic Foundation.
- Nugent, J. B., & Robinson, J. A. (2010). Are factor endowments fate? *Revista de Historia Economica, 28*(1), 45.
- Nye, J. V. (2011). Why Do Elites Permit Reform? *The Annual Proceedings of the Wealth and Well-Being of Nations*, 53.
- Olson, M. (1993). Dictatorship, Democracy, and Development. *American political science review*, 87(03), 567-576.
- Paldam, M., & Gundlach, E. (2008). Two views on institutions and development: the grand transition vs the primacy of institutions. *Kyklos, 61*(1), 65-100.
- Rodrik, D. (2000). Institutions for high-quality growth: what they are and how to acquire them. *Studies in Comparative International Development*, *35*(3), 3-31.
- Rodrik, D., Subramanian, A., & Trebbi, F. (2004). Institutions rule: the primacy of institutions over geography and integration in economic development. *Journal of economic growth*, 9(2), 131-165.
- Shirley, M. (2005). Institutions and Development. In C. Menard & M. Shirley (Eds.), Handbook of New Institutional Economics (pp. 611-638): Springer US.
- Simón-Moya, V., Revuelto-Taboada, L., & Guerrero, R. F. (2014). Institutional and economic drivers of entrepreneurship: An international perspective. *Journal of Business Research*, *67*(5), 715-721.
- Stephen, F. H., Urbano, D., & van Hemmen, S. (2005). The impact of institutions on entrepreneurial activity. *Managerial and Decision Economics*, 26(7), 413-419.
- World Bank. (2000). Reforming Public Institutions and Strengthening Governance. A World Bank Strategy. Washington, DC: World Bank.

# Chapter 2 Trade, Political Regimes and Economic Institutions

### Introduction

Explaining institutional change is a challenging endeavor that has only recently gained broader academic attention. In economics, changing relative prices and the specific role of trade have long been regarded as crucial drivers of institution building (North, 1990). In their study of Atlantic trade after 1500, Acemoglu, Johnson, and Robinson (2005b) show how the growth of trade has empowered merchant groups to push for institutional changes to constrain monarchies and protect traders' property rights. Others confirm a positive relation between trade openness and the quality of domestic economic institutions (Bergh, Mirkina, & Nilsson, 2014; Bhattacharyya, 2012; Levchenko, 2012; Nicolini & Paccagnini, 2011). A related strand of the literature also identifies a link between trade openness and the quality of governance (Al-Marhubi, 2005; Bonaglia, Braga de Macedo, & Bussolo, 2001; Wei, 2000), the level of corruption, and rent seeking (Ades & Di Tella, 1999; Gatti, 2004; Neeman, Paserman, & Simhon, 2004; Treisman, 2000).

An isolated analysis of trade openness per se, however, may grossly simplify the mechanisms at work. From a short-term perspective, trade regimes and domestic rules of international trade participation clearly reflect the priorities of the ruling class and are, in this sense, political constructs. It is, therefore, an open question whether higher trade flows<sup>1</sup> (henceforth, trade) will – independently of the political system – lead to an improvement in domestic economic institutions. This study shifts attention from trade openness as a construct of institutional quality to the realized trade and corresponding quality of economic institutions under

<sup>&</sup>lt;sup>1</sup> "Trade flow" is defined as the sum of exports and imports as a share of GDP.

different political institutions. The likely interaction effect between trade and political structures is far from trivial. Not only is this interaction important for understanding and predicting the direction of institutional change, but the assumed interaction effect should also help policymakers devise context-specific policies. Ultimately, such analysis is crucial to understanding whether, in the short-term, bilateral trade with authoritarian regimes is actually helping to improve their institutional quality or merely increasing the survival chances of dictatorial rulers.

To analyze the relation between trade and economic institutional quality under different political regimes, this study uses a longitudinal data set covering 138 countries during the period from 1984 to 2010. The observation period thus includes a time of intense global trade liberalization followed by an average growth in world exports of approximately 6 percent between 1990 and 2008 (World Trade Organization [WTO], 2011). Instrumented estimations yield two robust results: First, the results confirm a positive and significant effect of trade on the quality of economic institutions independent of the domestic political regime. Second, the effect of trade on economic institutional quality is considerably smaller in the presence of authoritarian political institutions. These results are robust to changes in the measurement of political institutions as well as to different identification strategies, and the overall conclusions remain similar.

The remainder of the paper is organized as follows: Section 2 reviews the theoretical arguments detailing central causal mechanisms by which trade may influence a country's economic institutions and highlights the interplay between political institutions and the effects of trade on economic institutions. Section 3 introduces the empirical model; section 4 describes the data, and section 5 presents the results. The final section concludes the paper.

## Trade and political institutions

Trade has featured prominently in the recent theoretical literature on the determinants of institutional change. The link between trade and the quality of domestic economic institutions rests on various mutually reinforcing transmission channels.

First, trade introduces a competition effect. Because domestic producers must compete both nationally and internationally with other producers, high transaction costs stemming from weak domestic institutions take a toll on

domestic production. Domestic economic agents, therefore, have incentives to lobby for and invest in better economic institutions to survive in the international marketplace (Islam & Montenegro, 2002). Simultaneously, competition has the potential to decrease corruption and rent-seeking activities and, thereby, improve bureaucratic efficiency (Ades & Di Tella, 1999; Bardhan, 1997; Treisman, 2000).

Trade also brings about a learning effect. Frequent cross-border interactions with international business partners increase knowledge and lead to a more informed citizenry (Al-Marhubi, 2005; Islam & Montenegro, 2002). Global flows of information improve and serve as alternative sources of knowledge and ideas. Such information spillovers help to improve domestic institutions because citizens will treat international practice as a benchmark and increasingly request changes that are consistent with international standards in their home institutions (Al-Marhubi, 2005).

Third, trade brings about a technology effect, which can have second-order effects on the quality of domestic economic institutions. Technological changes or technological shocks are likely to influence the distribution of domestic economic and political power (Coe & Helpman, 1995; Grossman & Helpman, 1991, 1995; Litwin, 1998; Spilimbergo, Londoño, & Székely, 1999). These shifts in relative societal power may subsequently affect the quality of economic institutions (Acemoglu, Johnson, & Robinson, 2005a).

Finally, trade and exposure to international market fluctuations trigger a certain insurance effect. Strong trade dependence can leave the economy more prone to external shocks such as sharp interest rate fluctuations, abrupt changes in the terms of trade, or a sudden capital outflow (Al-Marhubi, 2005). To cope with these fluctuations, countries seek to improve their institutional environments. In parallel, trade with foreign business partners requires a more sophisticated contract law and legal arbitration in case of business conflicts. This may push governments to legal reforms if they wish to reap the benefits of international trade (Islam & Montenegro, 2002). Particularly, membership or aspiration to gain membership in international organizations such as the WTO requires the signatory to bring domestic rules and regulations into compliance with international standards. In sum, the close link between trade and institutional qualities suggests the following hypothesis:

Hypothesis 1: The higher a country's trade is, the better the quality of domestic economic institutions.

The link between trade and the quality of economic institutions, however, is likely to vary with political regimes. If a small elite not subject to an effective system of political checks and balances holds political power, leaders are likely to establish extractive rules to pursue their own self-interest (North, 1990; North & Weingast, 1989).

The mechanisms impeding positive institutional effects from trade can be twofold: On the one hand, authoritarian regimes applying extractive approaches are likely to have trade regimes in place that do not allow the broader participation of the general populace. Key elements are typically large shares of state-ownership in trading and manufacturing companies, trade monopolies, and monopsonies (Acemoglu et al., 2005a). Many countries, for instance, rely on governmentowned agro-food trading companies that buy cash crops from domestic farmers at prices well below the world market prices to generate substantial monopsony rents when trading these products on the world market. In fact, from 1959 to 1960, when cocoa prices dropped by £50 a ton, instead of subsidizing their farmers to maintain a stable price, the governments of Ghana and Nigeria passed the burden of the drop in prices on to them (Acemoglu et al., 2005a).

Similarly, the political elite may use regulations, quotas, and licenses to manipulate the market structure and impede market access (Aidt & Gassebner, 2010). Typically, the goal is to retain control over the bulk of the resources for a few, often politically connected key players. The export boom in the meat and cotton industries in Central America in the 1960s is an example that shows how political institutions were used to retain gains from growing trade for the ruling elite of large landholders (Do & Levchenko, 2006). To capture the economic benefits of a rapidly growing cotton and meat export market in the 1960s and 1970s, large landholders with political power first evicted smallholders from their land by sharply increasing their rents. Politically well-connected landholders then used their power to obtain formal land titles. Domestic rules cemented these landholders' dominant market position by imposing restrictions on the number of firms involved in meat packaging and their production capacity. This regulation effectively instituted market entry barriers that restricted competition and increased economic profits for the elite (Do & Levchenko, 2006).

Evidently, extractive governments can rely on a wide portfolio of policy measures to reduce the expected competition effect associated with trade. Extractive political institutions will dampen the corrective influence of trade on corruption; in fact, the participation of the politically well-connected elite in trading activities may increase corruption levels given the great financial rents at stake (Banlaoi, 2004; Hutchcroft, 1997). Soft budget constraints and frequent state subsidies sponsoring the state's participation in foreign trade reinforce the problem (Broadman & Recanatini, 2001). Similarly, because of the potential restriction of foreign trade to the privileged, politically well-connected elite, information spillovers are limited and direct learning effects may remain modest. The same is 26 true for technology effects. If the political leadership is in control of those economic sectors that benefit the most from new technologies and innovations, shifts in the societal power balance may be relatively limited. Clearly, political efforts to tightly control new communication technologies and social media tools represent one example of how restrictive political regimes seek to cement their power by controlling access to new technologies (Knutsen, 2015).

In addition, how a country responds to the rules that a trading agreement or an international organization (such as WTO) implements depends largely on the political regime (Mansfield, Milner, & Rosendorff, 2002). A democratic government's willingness to implement the rules of a trade agreement is subject to the voters' preferences, whereas an oligarchic state will first realize the vested interests of the ruling elite (Mansfield et al., 2002). Because institutional reforms generally have a direct redistributive effect from small groups of the political elite towards the general public, external pressure for reforms is rarely successful unless domestic governments are forced to respond for economic and financial reasons (Andrews, 2013).

Thus, institutional effects stemming from trade are unlikely to be independent of the prevailing political regime. Centralized power in the hands of the political elite is likely to have moderating effects on the assumed link between trade and economic institutional quality. This leads to the second hypothesis:

Hypothesis 2: Trade under extractive political regimes will lead to smaller improvements in economic institutional quality than that in democratic systems.

### Methodology

The empirical model broadly follows models commonly used in the literature on institutional development (see Bhattacharyya, 2012; La Porta, Lopez-de-Silanes, Shleifer, and Vishny, 1999); however, it differs in two respects. First, to test hypothesis 2, an interaction term is introduced to capture the effect of trade under different political regimes. Second, the dependent variable is bounded from above and below. Therefore, estimation is performed using a two-way Tobit model and the following regression equation is estimated:

$$eci_{it} = \alpha + \beta_1 trade_{it} + \beta_2 (trade_{it} \times poi_{it}) + \beta_3 poi_{it} + \gamma Z_{it} + \varepsilon_{it} \quad (1)$$

in which  $eci_{it}$  is a measure of the quality of economic institutions for country *i* in period *t*,  $\alpha$  is the constant, *trade*<sub>it</sub> is the log of trade share, *poi*<sub>it</sub> is a measure of country *i*'s quality of political institutions, an  $Z_{it}$  is the matrix of other control variables that contains population, legal origin, geography, and GDP per capita. The model is estimated using annual observations. The reason for using annual observations instead of five-year averages is that political institutions experience discontinuous change in large steps, for instance in the form of a sudden political regime change (Roland, 2004).

There may exist a causal link between the quality of economic institutions and trade (e.g., see Jansen & Nordås, 2004; Levchenko, 2007; Linders, de Groot, & Rietveld, 2005), which could introduce bias into the estimates. Two methods are used to identify the causal relationship: an instrumental variable technique and identification through the heteroscedasticity method. The instrumental variable (IV) used for trade is constructed using the approach that Frankel and Romer (1999) suggested. This approach involves estimating a gravity model for bilateral trade shares (relative to GDP) using the geographic and demographic characteristics of a country and the corresponding trading partners as explanatory variables<sup>2</sup>. The predicted values from the gravity equation are then used to estimate a "geographic component of countries' overall trade," sometimes referred to as 'natural openness' (Wei, 2000). This measure of 'natural openness' is then used as an instrument for trade. Because Frankel and Romer (1999) approach is cross-sectional in nature, a year by year estimation of the gravity model is performed in order to capture the time-varying influences of the variables suitable for the panel data estimation. The gravity model used in the construction of the IV is estimated using OLS and is given below:

$$\ln(\frac{trade_{ijt}}{GDP_{it}}) = \frac{\beta_{0t} + \beta_{1t}\ln(pop_{it}) + \beta_{2t}\ln(pop_{jt}) + \beta_{4t}\ln(distance_{ij})}{+ \beta_{5t}contiguity_{ij} + \varphi_t(Geography_j) + \delta_t (Geography_i)} + \epsilon_{ijt}$$
(2)

<sup>&</sup>lt;sup>2</sup> Data on trade with partners comes from the data set by Head, Mayer, and Ries (2010), which is only available up to the year 2006 and restricts the analysis when using instrumental variables. This data set also provides information on the GDP and population for the domestic country and its trading partner, the distance between each pair of countries, and whether they have a common language. Data on the country's other geographical features, such as its area, a dummy indicating whether it is landlocked, and its longitude and latitude, come from Mayer and Zignago (2011).

in which *trade*<sub>*iit*</sub> is bilateral trade from country *i* to country *j* in period *t* and  $GDP_{it}$  is the GDP of country *i* in period *t*. *Pop*<sub>*it*</sub> and *pop*<sub>*jt*</sub> represent country *i*'s and country *j*'s populations, respectively; *distance*<sub>*ij*</sub> is the distance between country *i* and country *j*; and *contiguity*<sub>*ij*</sub> is a dummy variable indicating whether country *i* and country *j* share a border. *Geography*<sub>*i*</sub> is a matrix containing a set of other geographical features of country *i*, which includes its area, a dummy indicating whether it is landlocked, its longitude and latitude, and a dummy indicating its continent. *Geography*<sub>*j*</sub> is a matrix containing whether it is landlocked, and a dummy indicating whether it is landlocked, and a dummy indicating whether it is landlocked, and a dummy indicating the area, a dummy indicating whether it is landlocked, and a dummy indicating the area, a dummy indicating whether it is landlocked, and a dummy indicating its area, a dummy indicating whether it is landlocked, and a dummy indicating its area, a dummy indicating whether it is landlocked, and a dummy indicating its continent. Note that the time-varying coefficients indicate that estimation is performed on a single year's data at a time.

The predicted values from equation (2) are then used to determine the predicted value of the bilateral trade share for each pair of countries. The predicted value of bilateral trade shares for each country compared with all other countries are then aggregated to obtain a measure of natural openness of the country denoted by  $Natopen_{tt}$ .

$$\widehat{Natopen_{it}} = \sum_{j} exp(predicted value of ln(\frac{trade_{ijt}}{GDP_{it}}))$$
(3)

The second estimation technique used to resolve the endogeneity problem is the identification through heteroscedasticity (IH) method. The IH method estimates the causal relation between variables by exploiting differences in the variance of the error terms across different sub-samples of the data. Rigobon (2003) shows that the identification problem can be solved if the data can be split into separate groups such that the variance of the structural error term differs while the parameters remain stable across groups. Each additional data split adds more equations to the system than unknowns, which allows the estimation of the structural parameters of interest. In principle, it is sufficient to identify the model if the data can be split into two groups with different variances in the structural error term.

Two different splits of data are used to consistently and robustly estimate the model. The first split – the Euro-colony split – distinguishes between countries that European powers colonized and countries that they did not colonize. The assumption underlying this split is that non-colonized countries may have had more heterogeneous outcomes in institutional arrangenicits than colonized countries; hence, the variance in structural shocks for non-colonized countries will be higher than that for colonized countries (Rigobon & Rodrik, 2005). The

second split – the development split – differentiates between countries based on their levels of development as measured by their per capita GDP.<sup>3</sup> Compared to that in low-income countries, there is a greater variance in per capita income in high-income countries. This implies that there will be relatively greater variance in investments related to institutional building in the high-income group.<sup>4</sup>

### Data

Data is compiled from several sources and covers a total of 138 countries for the period 1984-2010. The data is unbalanced because not all the countries have data from 1984 until 2010. The total number of observations, therefore, differs across the various model specifications.

#### Dependent variable

Data on the quality of economic institutions comes from the Investment Profile Index (IP Index) and the International Country Risk Guide (ICRG).<sup>5</sup> The score on the IP index is assigned on the basis of subjective analysis of the available information regarding the investment climate in a country. This index is composed of three sub-indices, which present a country's investment profile when combined. These sub-indices are Contract Viability and Expropriation, Profits Repatriation, and Payment Delays. The three sub-indices capture dimensions of economic institutions such as the protection of property rights and contract enforcement. For the purpose of this analysis, a central advantage of these

<sup>&</sup>lt;sup>3</sup> Countries with a GDP per capita of \$5,000 (in current prices) or more in the year 2000 are placed in one group and the rest are placed in the other group.

<sup>&</sup>lt;sup>4</sup> For the development split variance in IP index for developed countries is 6.08 and for developing countries it is 4.04. Similarly, for the Euro-colony split countries that were colonized have a variance of 5.60 on IP index whereas, countries that were not colonized have a variance of 6.87. The difference in variance between the groups in both of the splits is statistically significant.

<sup>&</sup>lt;sup>5</sup> Several measures to capture changes in economic institutions are highlighted in the literature. These include the measure of Political Freedom and Civil Liberties from Freedom House (see, for example, La Porta et al., 1999; Nicolini & Paccagnini, 2011; Scully, 1988). Other measures include Expropriation Risk, Rule of Law, and Repudiation of Contracts by Government, which are available from ICRG (see, for example, Acemoglu, Johnson, & Robinson, 2001; Bhattacharyya, 2012; Knack & Keefer, 1995), and the Executive Constraint Index from the Polity IV data set (see, for example, Acemoglu et al., 2001; Bhattacharyya, 2012). These measures either partially reflect the variation in economic institutions and overlap with political institutions or cover a limited time period and are therefore not appropriate for the analysis.

measures is that they do not include any political aspects. This index has been used in similar contexts in a variety of empirical studies (for example, Harms & An de Meulen, 2013; Nsouli, Atoian, & Mourmouras, 2004; Rajan & Subramanian, 2007). The index varies from 0 to 12, and a higher value represents less risk of investment.

### Explanatory variables

The explanatory variables include the measure of trade and the measure of political institutions. Trade is measured by the trade to GDP ratio, taken from the Penn World Table 7.0 (PWT 7.0).<sup>6</sup> It is one of the most widely used measures of trade in the literature (see, for example, Dollar & Kraay, 2003, 2004; Frankel & Romer, 1999; Rodrik, Subramanian, & Trebbi, 2004).

To proxy for the quality of political institutions, data is taken from the ICRG's Democratic Accountability Index (DA Index). This measure of political institutions captures political accountability – the extent to which policymakers are restricted from following the interests of the elite in establishing extractive economic institutions. The DA Index measures the responsiveness of government to its citizens and the checks and balances on the executive that are in place. This index varies from 0 to 6 and differentiates between five types of governance: alternating democracy, dominated democracy, de facto one-party state, de jure one-party state, and autarchy. In general, alternating democracies receive the most points and autarchies received the fewest points. For the ease of interpretation of the results, the index is reversed so that democracies receive the fewest points and autarchies receive the most points.

#### **Control** variables

Following the literature (e.g., see Bhattacharyya, 2012; La Porta et al., 1999; Levchenko, 2007, 2012), the estimation model includes a set of control variables that can influence economic institutions and that are potentially correlated with the trade. First, it is highly likely that rich countries can afford better institutions as well as they tend to trade more (Acemoglu et al., 2001; Alonso & Garcimartín, 2013; Frankel & Romer, 1999; La Porta et al., 1999; Rigobon & Rodrik, 2005). To control for the level of development, the average of log of PPP-converted GDP per capita (in current prices) is included. In addition to this, yearly growth rate in

<sup>&</sup>lt;sup>6</sup> The log of trade to GDP ratio is taken.

PPP-converted GDP per capita (in current prices) is included to control for the confounding effect arising from income growth. The log of total population is added to control for the demographic structure of the country because the demographic structure affects the security of property rights (Harms & An de Meulen, 2013). La Porta et al. (1999) observed that a socialist legal system negatively affected institutional development in comparison to common law countries; therefore, legal origins are included as a control in the analysis. The quality of institutions and trade can also be influenced by geographical conditions such as a country's location, natural endowments, and disease environment (Easterly & Levine, 2003; Frankel & Romer, 1999). Hence, geographical conditions are controlled for using the distance from the equator as measured by the absolute value of latitude.

Table A1 in the appendix provides descriptive statistics of the variables used in the analysis. The first two columns report the mean and standard deviations of the full sample. The average IP score is 7.2, and the average trade to GDP ratio in the sample is 78.41. The latter is approximately the same as Switzerland's mean trade to GDP ratio, which is 78.75. Moreover, the average values for the DA index and the PR index are 2.25 and 3.49, respectively. The rest of the table reports the mean and standard deviations when the data is split into sub-samples based on the quartile values of the trade to GDP ratio. A clear pattern is evident in the first two rows of the table; trade and the IP index are positively correlated. The increase in the average trade to GDP ratio across sub-samples is substantial compared to the increase in the IP index, especially when the last two columns are compared. This shows an increase in the average trade to GDP ratio of approximately 1.25 standard deviations. However, the average IP index only increases marginally by only 0.2 standard deviations. This supports the hypothesis that the effect of trade on economic institutions may depend on the quality of political institutions.

Table A2 in the appendix provides correlations between dependent and explanatory variables. The correlation between the IP index and the log of trade to GDP ratio is positive, suggesting that countries with high trade to GDP ratios have better economic institutions. The correlation of the IP index to the DA index is negative, which indicates that countries with better political institutions have better quality economic institutions.

### Results

Table 1 presents the OLS and Tobit model estimation results of the effect of trade and political institutions on the quality of economic institutions. Columns 1-3 present the unconditional effect of trade on the quality of economic institutions. Whereas Column 1 shows the results of the OLS model, Columns 2 and 3 show the results of the random effect Tobit model and IV estimation using Tobit. The results suggest that there is a positive and significant effect of trade on economic institutional quality for all the specifications. Moreover, the results hold up when the IV is used. These results are consistent with hypothesis 1, which suggests that larger trade volumes improves economic institutional quality and also confirms the results of previous studies (see Bergh et al., 2014; Bhattacharyya, 2012; Levchenko, 2012). The effect of the DA index is negative and significant for all specifications. This accords with the theory that predicts a positive relation between the quality of political institutions and quality of economic institutions (see, for example, Acemoglu et al., 2005a; North, 1990; North & Weingast, 1989).
|                     |                                 | Тс                              | bit                             |                                 | To                              | bit                            |
|---------------------|---------------------------------|---------------------------------|---------------------------------|---------------------------------|---------------------------------|--------------------------------|
| Variables           | OLS                             | Random<br>Effect                | $IV^{\S}$                       | OLS                             | Random<br>Effect                | IV <sup>§§</sup>               |
| log (trade)         | 1.08***                         | 1.40***                         | 2.37***                         | 1.36***                         | 2.87***                         | 2.71***                        |
|                     | (0.06)                          | (0.13)                          | (0.30)                          | (0.12)                          | (0.21)                          | (0.36)                         |
| log (trade) × DA    |                                 |                                 |                                 | -0.09***                        | -0.43***                        | -0.38***                       |
| index               |                                 |                                 |                                 | (0.03)                          | (0.05)                          | (0.11)                         |
| DA index            | -0.46***                        | -0.51***                        | -0.36***                        | -0.09                           | 1.29***                         | 1.17***                        |
|                     | (0.02)                          | (0.03)                          | (0.03)                          | (0.14)                          | (0.20)                          | (0.44)                         |
| log (Avg. GDP       | 0.55***                         | 2.69***                         | 0.35***                         | 0.57***                         | 2.58***                         | 0.54***                        |
| per capita)         | (0.04)                          | (0.46)                          | (0.05)                          | (0.04)                          | (0.44)                          | (0.06)                         |
| GDP per capita      | 3.03***                         | 1.61***                         | 2.81***                         | 3.04***                         | 1.69***                         | 3.38***                        |
| growth              | (0.44)                          | (0.39)                          | (0.57)                          | (0.44)                          | (0.38)                          | (0.53)                         |
| log (Population)    | 0.18***                         | 3.49***                         | 0.41***                         | 0.21***                         | 3.30***                         | 0.37***                        |
|                     | (0.03)                          | (0.32)                          | (0.06)                          | (0.03)                          | (0.34)                          | (0.05)                         |
| Constant            | -2.63***                        | -42.92***                       | -8.70***                        | -4.22***                        | -46.9***                        | -11.1***                       |
|                     | (0.54)                          | (5.25)                          | (1.59)                          | (0.80)                          | (5.23)                          | (1.99)                         |
| Other Controls      | Latitude<br>and Legal<br>origin | Latitude<br>and Lega<br>origin |
| Observations        | 3,434                           | 3,434                           | 2,798                           | 3,434                           | 3,434                           | 2,798                          |
| R-squared           | 0.37                            | -                               | -                               | 0.37                            | -                               | -                              |
| Number of countries | 138                             | 138                             | 134                             | 138                             | 138                             | 134                            |

Table 1 Effect of Trade and DA Index on IP Index

Standard errors in parentheses; \*\*\* p<0.01, \*\* p<0.05, \* p<0.1; § Instrumental variable:  $Natopen_{ii}$ ; §§ Instrumental variable:  $Natopen_{ii} \times DA$  index<sub>it</sub>

However, the relation between trade and the quality of economic institutions may depend on the quality of political institutions as hypothesis 2 indicates. The model that equation (1) gives captures this effect by including an interaction term between trade and political institutions. The results of estimating several specifications of equation (1) while using the DA index as a measure of political institutions are presented in Columns 4-6. The coefficient of the interaction term is negative and significant for all specifications, whereas the coefficient of trade is positive and significant. The results are consistent with the second hypothesis, which suggests that in the presence of extractive political institutions. These results appear to confirm that authoritarian regimes effectively restrict the broader participation of the public in trade. In addition, they reveal that the regimes may resort to the use of regulations, quotas, and licenses to manipulate the market structure and impede market access, reducing the effect of trade on economic institutions.

For the OLS model in Column 4, the coefficients for trade and the interaction term are 1.36 and -0.09, respectively, and they are statistically significant at a 1 percent level of significance. This indicates that the effect of trade decreases under more extractive political regimes. To visualize the size of the moderating effect, Figure 1 depicts the marginal effect of trade at different values of the DA index for the OLS model in the graph. The graph indicates that the marginal effect of trade decreases as political institutions become more extractive; however, it remains positive and significant, even for the most extractive regime.



Figure 1 Marginal effect of log of trade on IP index for different values of DA index

The overall conclusions remain unchanged when natural openness as an IV is used in estimation. A comparison of Columns 4 and 6 reveals that the coefficient of trade and the interaction term increased in magnitude in Column 6. Figure 2 provides a graph of the total marginal effect of trade at different values of the DA index when natural openness is used as an IV. The graph suggests that at lower values of the DA index, the marginal effect of trade is higher and that as the DA index increases, the marginal effect tends to decrease.



#### Figure 2

Marginal effect of log of trade on IP index for different values of DA index using natural openness as an IV

Moreover, for extremely high values of the DA index, the marginal effect becomes statistically insignificant. Thus, compared to Vietnam, which has extractive political institutions (average DA score of 4.13), Taiwan, with its inclusive political institutions (average DA score of 1.56), will experience faster improvement in its quality of economic institutions as its cross-border trade increases. Moreover, countries such as Myanmar or Somalia, with an average DA score of 5.4 and 5.02, respectively, may not experience any significant improvements in their quality of economic institutions in response to increases in trade.

The effect of the DA index on the IP index is negative and statistically significant for reasonable values of trade share as Figure 3 shows. For instance, the marginal effect of the DA index at the mean value of the trade is -0.49.<sup>7</sup> This suggests that, for a country with an average trade to GDP ratio, deterioration in political institutions will result in the deterioration of economic institutions.

<sup>&</sup>lt;sup>7</sup> The mean value of the trade to GDP ratio in the sample is 78.41, and the corresponding value in log is 4.36.



Figure 3 Marginal effect of DA index on IP index for different values of log of trade

The results using the IH method also suggest that there is a positive causal relation between trade and the quality of economic institutions, and this effect becomes attenuated in the presence of extractive political institutions as shown in Table 2.

| Variables              | D      | evelopment Sp | olit   | H      | Euro Colony Spl | lit    |
|------------------------|--------|---------------|--------|--------|-----------------|--------|
| Log(Trade)             | 1.02   | 2.66          | 2.73   | 1.08   | 4.81            | 5.85   |
|                        | (3.26) | (7.21)        | (6.25) | (6.63) | (6.18)          | (5.44) |
| Log(Trade) × DA index  |        | -0.39         | -0.36  |        | -0.78           | -0.99  |
|                        |        | (4.89)        | (3.88) |        | (4.53)          | (4.25) |
| Control Variables      |        |               |        |        |                 |        |
| DA index               |        | Yes           | Yes    |        | Yes             | Yes    |
| Average GDP per capita |        | Yes           | Yes    |        | Yes             | Yes    |
| GDP per capita growth  |        | Yes           | Yes    |        | Yes             | Yes    |
| Log(Population)        |        | Yes           | Yes    |        | Yes             | Yes    |
| Latitude               |        | Yes           | Yes    |        | Yes             | Yes    |
| Legal Origin           |        |               | Yes    |        |                 | Yes    |
| Constant               | Yes    | Yes           | Yes    | Yes    | Yes             | Yes    |
| Observations           | 3434   | 3434          | 3434   | 3434   | 3434            | 3434   |
| Number of Countries    | 138    | 138           | 138    | 138    | 138             | 138    |

 Table 2

 Identification through Heteroscedasticity: Effect of Trade and DA Index on IP Index

T-statistics in the parentheses. Based on 500 bootstrap samples.

Moreover, the results are similar for both splits of the data. For instance, Column 3 presents the estimates of the full model using the development split, and the result suggests that a 10 percent increase in the trade to GDP ratio is related to an increment of 0.27 units in the IP index for a country with a DA index of 0. For a country with a DA index of 6, this effect decreases to 0.06, which is nearly 4.5 times less than the effect in a country with a DA index of 0. To put this into perspective, consider two countries, A and B, both with similar characteristics and an IP index of 6 in the first year, but with different DA scores. Let country A be an autocracy with a DA score of 6, and let country B be a democracy with a DA score of 0. If both countries experience a 10 percent increase in trade per year for 10 years with all other variables being constant, country B's IP index will increase from 6 in year 1 to 8.7 in year 10, whereas country A will experience a marginal increase of 0.6 units in its IP index as shown in Figure 4.



#### Figure 4

Development of IP index over a 10 year period for a 10% increase in trade to GDP ratio per year

#### **Robustness checks**

Several robustness checks are performed to scrutinize the results. First, internal instruments are used to further assess the sensitivity of the results. Second, an alternative measure of political institutions is used to counter any bias resulting from the construction of a particular index. Last, the sample is split by region and estimation is performed on these sub-samples using the IP index and the DA index as measures of economic and political institutions, respectively, to see if the results hold.

In order to robustly establish the link between trade and economic institutions, the model is estimated using internal instruments (see, for example, Dollar & Kraay, 2003, 2004; Yanikkaya, 2003). The tenth, fifteenth, and twentieth lags of trade are used as IVs. The identifying assumption here is that although institutions in period t affect trade openness in period t, they cannot affect trade in previous periods. The reason for choosing a long lag length as an IV is to minimize the correlation that today's institutions may have with past values of trade openness

because of the often slowly changing nature of economic institutional quality. Table 3 shows the estimation results using lag of trade as an IV, which confirms the main result presented in Table 1. The coefficient estimates of trade and the interaction term have the expected signs and are significant at the 5 percent level. However, in comparison to the results of the IV estimation using natural openness, the coefficients are relatively small.

| Variables              | IV                     | IV                     | IV                      | IV                     | IV                     | IV                      |
|------------------------|------------------------|------------------------|-------------------------|------------------------|------------------------|-------------------------|
|                        | Estimates <sup>‡</sup> | Estimates <sup>§</sup> | Estimates <sup>§§</sup> | Estimates <sup>‡</sup> | Estimates <sup>§</sup> | Estimates <sup>§§</sup> |
| log (trade)            | 0.95***                | 1.00***                | 1.38***                 | 1.26***                | 1.25***                | 1.71***                 |
|                        | (0.09)                 | (0.10)                 | (0.12)                  | (0.15)                 | (0.17)                 | (0.18)                  |
| log (trade) × DA       |                        |                        |                         | -0.10***               | -0.09**                | -0.13**                 |
| index                  |                        |                        |                         | (0.04)                 | (0.05)                 | (0.05)                  |
| DA index               | -0.44***               | -0.42***               | -0.42***                | 0.00                   | -0.04                  | 0.13                    |
|                        | (0.03)                 | (0.03)                 | (0.03)                  | (0.17)                 | (0.19)                 | (0.22)                  |
| log (Avg. GDP per      | 0.64***                | 0.66***                | 0.55***                 | 0.67***                | 0.69***                | 0.61***                 |
| capita)                | (0.04)                 | (0.04)                 | (0.05)                  | (0.04)                 | (0.05)                 | (0.05)                  |
| GDP per capita         | 3.01***                | 2.96***                | 2.59***                 | 3.02***                | 2.96***                | 2.65***                 |
| growth                 | (0.47)                 | (0.49)                 | (0.52)                  | (0.47)                 | (0.49)                 | (0.52)                  |
| log (Population)       | 0.17***                | 0.19***                | 0.24***                 | 0.20***                | 0.21***                | 0.27***                 |
|                        | (0.03)                 | (0.03)                 | (0.03)                  | (0.03)                 | (0.03)                 | (0.04)                  |
| latitude               | 0.52*                  | 0.54*                  | 0.77**                  | 0.36                   | 0.40                   | 0.53                    |
|                        | (0.31)                 | (0.31)                 | (0.32)                  | (0.32)                 | (0.32)                 | (0.33)                  |
| Constant               | -2.83***               | -3.36***               | -4.60***                | -4.61***               | -4.85***               | -6.58***                |
|                        | (0.64)                 | (0.68)                 | (0.76)                  | (0.95)                 | (1.03)                 | (1.10)                  |
| Other Controls         | Legal origin           | Legal origin           | Legal origin            | Legal origin           | Legal origin           | Legal origin            |
| Observations           | 3,326                  | 3,219                  | 3,043                   | 3,326                  | 3,219                  | 3,043                   |
| Number of<br>Countries | 138                    | 138                    | 128                     | 138                    | 138                    | 128                     |

#### Table 3

Effect of Trade and DA index on IP index using lagged values of trade to GDP ratio as IVs and Tobit model

Standard errors in parentheses. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1. <sup>‡</sup> Instrumental variable: *lnopen it-10* and *lnopen it-10* × *DA index it.* <sup>§</sup> Instrumental variable: *lnopen it-15* and *lnopen it-15* × *DA index it.* <sup>§§</sup> Instrumental variable: *lnopen it-20* and *lnopen it-20* × *DA index it.* 

Table 4 presents the results using Freedom House's Political Rights Index (PR Index) as an alternative measure (of political institutions) to the DA index.<sup>8</sup> The overall conclusion drawn from the results is similar to the conclusions obtained using the DA index. Trade has a positive and significant effect on the quality of

<sup>&</sup>lt;sup>8</sup> The PR Index "measures the degree of freedom in the electoral process, political pluralism and participation, and functioning of government. Numerically, Freedom House rates political rights on a scale of 1 to 7, with 1 representing the most free and 7 representing the least free" (Freedom in the Worla, 2010: Survey Methodology). A score of 1 represents a country having free and fair elections, political competition, and autonomy for all citizens, whereas a PR Index value of 7 represents a country in which political rights are essentially missing because of extremely oppressive regimes, civil war, extreme violence, or warlord rule.

economic institutions. Moreover, increased political freedom significantly increases the effect of trade on the quality of economic institutions. These results suggest that the interaction term robustly captures the effect of trade on economic institutions under different political regimes as measured by two different methods. The result of the IV estimation in Column 6 suggests that a 10 percent increase in trade in a country with a PR index of 1 is related to an increment of 0.28 units in the IP index. However, for a country with a PR index of 7, a 10 percent increase in trade merely increases the IP index by 0.014 units, approximately one-half of the effect of a country with a PR index of 1.

#### Table 4

|                      |          | To                | bit                   |          | То                | bit                            |
|----------------------|----------|-------------------|-----------------------|----------|-------------------|--------------------------------|
| Variables            | OLS      | Random<br>Effects | $\mathrm{IV}^\dagger$ | OLS      | Random<br>Effects | $\mathrm{IV}^{\dagger\dagger}$ |
| log(trade)           | 1.19***  | 1.88***           | 2.59***               | 1.80***  | 3.62***           | 2.99***                        |
|                      | (0.07)   | (0.11)            | (0.30)                | (0.14)   | (0.25)            | (0.39)                         |
| log(trade)× PR index |          |                   |                       | -0.14*** | -0.46***          | -0.23***                       |
|                      |          |                   |                       | (0.03)   | (0.04)            | (0.09)                         |
| PR index             | -0.28*** | -0.32***          | -0.24***              | 0.29**   | 1.62***           | 0.71**                         |
|                      | (0.02)   | (0.03)            | (0.02)                | (0.12)   | (0.18)            | (0.36)                         |
| log (Avg. GDP per    | 0.54***  | 1.73***           | 0.34***               | 0.60***  | 3.17***           | 0.50***                        |
| capita)              | (0.04)   | (0.20)            | (0.05)                | (0.04)   | (0.51)            | (0.07)                         |
| GDP per capita       | 2.94***  | 1.76***           | 2.63***               | 2.93***  | 1.64***           | 3.05***                        |
| growth               | (0.47)   | (0.42)            | (0.60)                | (0.47)   | (0.40)            | (0.57)                         |
| log (Population)     | 0.24***  | 2.26***           | 0.48***               | 0.30***  | 4.20***           | 0.47***                        |
|                      | (0.03)   | (0.09)            | (0.06)                | (0.03)   | (0.31)            | (0.06)                         |
| Latitude             | 0.64**   | -5.55***          | 0.76**                | 0.37     | -11.64***         | 0.30                           |
|                      | (0.31)   | (1.80)            | (0.36)                | (0.31)   | (3.94)            | (0.38)                         |
| Constant             | -3.55*** | -28.89***         | -9.87***              | -6.95*** | -60.98***         | -12.61***                      |
|                      | (0.57)   | (1.99)            | (1.51)                | (0.90)   | (5.68)            | (2.14)                         |
| Other Controls       | Legal    | Legal             | Legal                 | Legal    | Legal             | Legal                          |
|                      | Origin   | Origin            | Origin                | Origin   | Origin            | Origin                         |
| Observations         | 3,210    | 3,210             | 2,754                 | 3,210    | 3,210             | 2,754                          |
| R-squared            | 0.35     | -                 | -                     | 0.36     | -                 | -                              |
| Number of Countries  | 135      | 135               | 135                   | 135      | 135               | 135                            |

Effect of Trade and PR Index on IP Index

Standard errors in parentheses. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1. <sup>†</sup> Instrumental variable: *Natopen*<sub>it</sub>.  $^{\dagger\dagger}$  Instrumental variable: *Natopen*<sub>it</sub> and *Natopen*<sub>it</sub> × *PR index*<sub>it</sub>.

Lastly, the model is estimated using different sub-samples of the data because the heterogeneity of the countries may bias the results. The sample is divided by regions and the model is re-estimated using linear regression instead of Tobit because there is little or no censoring in the data. The results are presented in Table 5. The unconditional effect of trade on economic institutional quality is positive and statistically significant for all the regions. After the addition of the

interaction term in the model, the overall conclusions remain unchanged for all regions except for African countries. For these countries, the coefficient of trade is negative, whereas the coefficient of the interaction term is positive. The counterintuitive results for Africa are driven by the fact that the majority of African countries mainly export natural resources and only 10 percent of the sub-sample is categorized as democratic. These results may also suggest that a certain level of development is a precondition for the improvement of institutional quality.

| Variables                 | Africa       | Asia         | Americas     | Europe       | Africa       | Asia         | Americas     | Europe       |
|---------------------------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|--------------|
| log(trade)                | 0.95***      | 1.75***      | 0.94***      | 3.27***      | 0.15         | 2.77***      | 2.21***      | 4.02***      |
|                           | (0.14)       | (0.19)       | (0.25)       | (0.30)       | (0.35)       | (0.32)       | (0.39)       | (0.35)       |
| log(trade)× DA index      |              |              |              |              | 0.19**       | -0.3***      | -0.49***     | -0.55***     |
| 1                         |              |              |              |              | (0.08)       | (60.0)       | (0.11)       | (0.15)       |
| DA index                  | -0.42***     | -0.35***     | -0.91***     | -1.13***     | -1.19***     | 1.09***      | 1.15**       | $1.26^{*}$   |
|                           | (0.06)       | (0.06)       | (0.07)       | (60.0)       | (0.32)       | (0.38)       | (0.48)       | (0.65)       |
| log (Avg. GDP per capita) | 0.94***      | 1.64***      | 0.06         | 0.65**       | 0.95***      | 1.79***      | 0.19         | 0.61**       |
|                           | (0.21)       | (0.19)       | (0.40)       | (0.31)       | (0.21)       | (0.20)       | (0.41)       | (0.31)       |
| GDP per capita growth     | 2.08***      | 1.67***      | 1.23         | 4.35***      | 2.12***      | 1.71***      | 0.80         | 4.90***      |
|                           | (0.60)       | (0.63)       | (1.14)       | (1.18)       | (0.60)       | (0.62)       | (1.13)       | (1.18)       |
| log (Population)          | 1.02***      | 1.01***      | 0.50***      | 0.65***      | 0.97***      | 1.11***      | 0.62***      | 0.72***      |
|                           | (0.13)       | (0.12)       | (0.15)       | (0.10)       | (0.13)       | (0.13)       | (0.16)       | (0.11)       |
| Latitude                  | -0.97        | -0.02        | 0.72         | -4.57*       | -0.97        | -0.02        | 0.89         | -5.30*       |
|                           | (1.82)       | (1.46)       | (1.88)       | (2.66)       | (1.78)       | (1.47)       | (1.89)       | (2.73)       |
| Constant                  | -11.6***     | -22.4***     |              | -13.4***     |              | -28.7***     |              | -16.4***     |
|                           | (2.11)       | (2.97)       |              | (2.80)       |              | (3.34)       |              | (2.94)       |
| Other Controls            | Legal origin |
| Observations              | 677          | 931          | 725          | 784          | 977          | 931          | 725          | 784          |
| Number of Countries       | 37           | 37           | 27           | 36           | 37           | 37           | 27           | 36           |

r Regression: Effect of Trade and DA Index on IP Index in Various Splits of the Data

T**able 5** Linear Reg

# Conclusions

This study explores the relation between trade and economic institutions under different political regimes. There is ample theoretical grounding for the belief that trade has a positive effect on the quality of economic institutions. However, the effect of trade on economic institutions is likely to vary for different political regimes. The political elite are likely to establish extractive economic institutions if political power is not subject to checks and balances. In an authoritarian regime, the political elite can effectively restrict the broader participation of the public in trade and may also resort to the use of regulations, quotas, and licenses to manipulate the market structure and impede market access. Higher trade flow is, therefore, not necessarily associated with better economic institutions. Thus, this study hypothesize that higher trade under extractive political regimes will lead to smaller improvements in institutional quality than that in democratic societies.

The results can be summarized as follows: The study confirms a positive and significant effect of trade on institutional quality. Further, under extractive political regimes, trade has a significantly reduced effect on economic institutional quality. The results hold true for both pooled and panel data specifications and under various model specifications. The findings suggest that in political regimes where there is some democratic accountability, increased trade will result in improvement in economic institutional quality, whereas in the absence of democratic accountability, higher trade volumes will have substantially smaller effects on economic institutions. This shows that increased trade flow alone cannot improve the quality of economic institutions in a country. Rather, political institutions also play a key role in this relationship.

The results also invite several policy implications for donor agencies and international financial institutions (IFIs). Countries that transition from autocracies to more pluralistic regimes will benefit from increased trade in terms of improvements in their economic institutions. Thus, trade liberalization policies to lower trade barriers can be an effective tool for improving institutional quality in newly democratized countries. Moreover, preferential trade agreements that advanced economies initiate with newly democratized countries can lead to institutional change. The results also stress the importance of context-specific policy design, taking into account the domestic social and political context instead of simply adopting a blueprint approach.

The discourse on trade and institutional quality is far from complete and further study is necessary to explore the link between different types of institutions and 44

different individual channels through which trade affects the quality of institutions. It would also be of interest to see what type of trade (primary goods, industrial goods, services) leads to greater changes in institutional quality.

## References

- Acemoglu, D., Johnson, S., & Robinson, J. (2001). The Colonial Origins of Comparative Development: An Empirical Investigation. *American Economic Review*, 91(5), 1369-1401.
- Acemoglu, D., Johnson, S., & Robinson, J. (2005a). Institutions as a Fundamental Cause of Long-Run Growth. In A. Philippe & N. D. Steven (Eds.), *Handbook of economic growth* (Vol. Volume 1, Part A, pp. 385-472): Elsevier.
- Acemoglu, D., Johnson, S., & Robinson, J. (2005b). The Rise of Europe: Atlantic Trade, Institutional Change, and Economic Growth. *American Economic Review*, 95(3), 546-579.
- Ades, A., & Di Tella, R. (1999). Rents, competition, and corruption. *American Economic Review*, 89(4), 982-993.
- Aidt, T. S., & Gassebner, M. (2010). Do Autocratic States Trade Less? The World Bank Economic Review, 24(1), 38-76.
- Al-Marhubi, F. A. (2005). Openness and governance: Evidence across countries. Oxford Development Studies, 33(3-4), 453-471.
- Alonso, J. A., & Garcimartín, C. (2013). The determinants of institutional quality. More on the debate. *Journal of International Development, 25*(2), 206-226.
- Andrews, M. (2013). *The limits of institutional reform in development: Changing rules for realistic solutions*: Cambridge University Press.
- Banlaoi, R. (2004). Globalization and nation-building in the Philippines: State predicaments in managing society in the midst of diversity. Iin Y. Sato, (Ed.) *Growth and Governance in Asia*, Honolulu: Asia-Pacific Center for Security Studies.
- Bardhan, P. (1997). Corruption and development: a review of issues. *Journal of economic literature*, *35*(3), 1320-1346.
- Bergh, A., Mirkina, I., & Nilsson, T. (2014). Globalization and Institutional Quality— A Panel Data Analysis. *Oxford Development Studies*, *42*(3), 365-394.
- Bhattacharyya, S. (2012). Trade liberalization and institutional development. *Journal of Policy Modeling*, 34(2), 253-269.
- Bonaglia, F., Braga de Macedo, J., & Bussolo, M. (2001). How Globalization Improves Governance.
- Broadman, H. G., & Recanatini, F. (2001). Seeds of corruption–Do market institutions matter? MOST: Economic Policy in Transitional Economies, 11(4), 359-392.
- Coe, D. T., & Helpman, E. (1995). International R&D Spillovers. *European Economic Review*, *39*(5), 859-887.

- Do, Q.-T., & Levchenko, A. A. (2006). Trade, inequality, and the political economy of institutions.
- Dollar, D., & Kraay, A. (2003). Institutions, trade, and growth. *Journal of monetary* economics, 50(1), 133-162.
- Dollar, D., & Kraay, A. (2004). Trade, Growth, and Poverty\*. *The Economic Journal*, *114*(493), F22-F49.
- Easterly, W., & Levine, R. (2003). Tropics, germs, and crops: how endowments influence economic development. *Journal of monetary economics*, 50(1), 3-39.
- Frankel, J. A., & Romer, D. (1999). Does trade cause growth? *American Economic Review*, 379-399.
- Freedom House. (2010). *Freedom in the World 2010*: The Annual Survey of Political Rights and Civil Liberties, Freedom House, New York. Data retrieved from: <u>www.freedomhouse.org</u>.
- Gatti, R. (2004). Explaining corruption: are open countries less corrupt? *Journal of International Development*, *16*(6), 851-861.
- Grossman, G. M., & Helpman, E. (1991). Trade, knowledge spillovers, and growth. *European Economic Review*, 35(2), 517-526.
- Grossman, G. M., & Helpman, E. (1995). Technology and trade. In G. M. Grossman & K. Rogoff (Eds) *Handbook of international economics*, Amsterdam: North-Holland.
- Harms, P., & An de Meulen, P. (2013). Demographic structure and the security of property rights: The role of development and democracy. *European Journal of Political Economy*, 29, 73-89.
- Head, K., Mayer, T., & Ries, J. (2010). The erosion of colonial trade linkages after independence. *Journal of International Economics*, *81*(1), 1-14.
- Heston, A., Summers, R., & Aten, B. (2011). Penn World Table Version 7.0. Center for International Comparisons of Production, Income and Prices at the University of Pennsylvania.
- Hutchcroft, P. D. (1997). The politics of privilege: assessing the impact of rents, corruption, and clientelism on Third World development. *Political Studies*, 45(3), 639-658.
- ICRG. (2012). International Country Risk Guide Researchers Dataset, Table 3B. The Political Risk Services Group, third edition.
- Islam, R., & Montenegro, C. E. (2002). What determines the quality of institutions? World Bank Policy Research Working Paper(2764).
- Jansen, M., & Nordås, H. K. (2004). Institutions, trade policy and trade flows. *Working paper Centre for Economic Policy Research*.

- Knack, S., & Keefer, P. (1995). Institutions and economic performance: cross-country tests using alternative institutional measures. *Economics & Politics*, 7(3), 207-227.
- Knutsen, C. H. (2015). Why Democracies Outgrow Autocracies in the Long Run: Civil Liberties, Information Flows and Technological Change. *Kyklos*, 68(3), 357-384.
- La Porta, R., Lopez-de-Silanes, F., Shleifer, A., & Vishny, R. (1999). The quality of government. *Journal of Law, Economics, and organization, 15*(1), 222-279.
- Levchenko, A. A. (2007). Institutional quality and international trade. *The Review of Economic Studies*, 74(3), 791-819.
- Levchenko, A. A. (2012). International Trade and Institutional Change. *Journal of Law, Economics, and organization*.
- Linders, G.-J. M., de Groot, H. L., & Rietveld, P. (2005). *Institutional Determinants of Bilateral Trade: An Analysis according to Product Type*.
- Litwin, C. (1998). Trade and income distribution in developing countries.
- Mansfield, E. D., Milner, H. V., & Rosendorff, B. P. (2002). Why Democracies Cooperate More: Electoral Control and International Trade Agreements. *International Organization*, 56(03), 477-513.
- Mayer, T., & Zignago, S. (2011). Notes on CEPII's distances measures: The GeoDist database.
- Neeman, Z., Paserman, M., & Simhon, A. (2004). Corruption and openness. *Centre for Rationality and Interactive Decision Theory, Hebrew University.*
- Nicolini, M., & Paccagnini, A. (2011). Does Trade Foster Institutions? *Review of Economics and Institutions*, 2(2).
- North, D. C. (1990). *Institutions, institutional change and economic performance:* Cambridge university press.
- North, D. C., & Weingast, B. R. (1989). Constitutions and commitment: the evolution of institutions governing public choice in seventeenth-century England. *The Journal of Economic History*, 49(04), 803-832.
- Nsouli, S. M., Atoian, R., & Mourmouras, A. (2004). Institutions, program implementation, and macroeconomic performance. *International Monetary Fund*.
- Rajan, R., & Subramanian, A. (2007). Does aid affect Governance? *The american economic review*, *97*(2), 322-327.
- Rigobon, R. (2003). Identification through heteroskedasticity. *Review of Economics and Statistics*, 85(4), 777-792.
- Rigobon, R., & Rodrik, D. (2005). Rule of law, democracy, openness, and income. *Economics of Transition*, *13*(3), 533-564.

- Rodrik, D., Subramanian, A., & Trebbi, F. (2004). Institutions rule: the primacy of institutions over geography and integration in economic development. *Journal of economic growth*, 9(2), 131-165.
- Roland, G. (2004). Understanding institutional change: fast-moving and slow-moving institutions. *Studies in Comparative International Development*, *38*(4), 109-131.
- Scully, G. W. (1988). The institutional framework and economic development. *The Journal of Political Economy*, 652-662.
- Spilimbergo, A., Londoño, J. L., & Székely, M. (1999). Income distribution, factor endowments, and trade openness. *Journal of development Economics*, 59(1), 77-101.
- Treisman, D. (2000). The causes of corruption: A cross-national study. *Journal of Public Economics*, 76(3), 399-457.
- Wei, S. J. (2000). Natural openness and good government. *National bureau of economic research*.
- World Trade Organization. (2011). Trade growth to ease in 2011 but despite 2010 record surge, crisis hangover persists, [Press release]. Retrieved from <u>http://www.wto.org/english/news\_e/pres11\_e/pr628\_e.htm</u>
- Yanikkaya, H. (2003). Trade Openness and Economic Growth: A Cross-Country Empirical Investigation. *Journal of development Economics*, 72(1), 57-89

Appendix

|                              | Full 5 | Full Sample  | 1 <sup>st</sup> Q | 1st Quartile | $2^{nd}$ Q | 2 <sup>nd</sup> Quartile | $3^{rd}$ Q | 3 <sup>rd</sup> Quartile | $4^{\rm th}{ m Q}$ | 4 <sup>th</sup> Quartile |
|------------------------------|--------|--------------|-------------------|--------------|------------|--------------------------|------------|--------------------------|--------------------|--------------------------|
| Variable                     | Mean   | Std.<br>Dev. | Mean              | Std.<br>Dev. | Mean       | Std.<br>Dev.             | Mean       | Std.<br>Dev.             | Mean               | Std.<br>Dev.             |
| IP index                     | 7.26   | 2.53         | 6.01              | 2.30         | 7.00       | 2.40                     | 7.75       | 2.37                     | 8.27               | 2.49                     |
| Trade                        | 78.41  | 51.85        | 32.92             | 10.45        | 56.67      | 5.79                     | 79.78      | 8.66                     | 144.26             | 60.47                    |
| DA index                     | 2.25   | 1.65         | 2.70              | 1.58         | 2.11       | 1.57                     | 2.10       | 1.67                     | 2.09               | 1.70                     |
| PR index                     | 3.49   | 2.17         | 4.11              | 2.15         | 3.38       | 2.09                     | 3.31       | 2.13                     | 3.10               | 2.16                     |
| GDP per capita               | 10388  | 12927        | 5821              | 8790         | 7953       | 8928                     | 12443      | 14091                    | 15333              | 16063                    |
| Latitude                     | 0.29   | 0.19         | 0.23              | 0.15         | 0.29       | 0.19                     | 0.31       | 0.20                     | 0.31               | 0.20                     |
| British Legal Origin         | 0.32   | 0.47         | 0.35              | 0.48         | 0.27       | 0.44                     | 0.27       | 0.44                     | 0.37               | 0.48                     |
| French Legal Origin          | 0.48   | 0.50         | 0.53              | 0.50         | 0.59       | 0.49                     | 0.42       | 0.49                     | 0.38               | 0.49                     |
| Socialist Legal Origin       | 0.12   | 0.32         | 0.08              | 0.28         | 0.07       | 0.25                     | 0.12       | 0.32                     | 0.21               | 0.41                     |
| German Legal Origin          | 0.05   | 0.21         | 0.04              | 0.19         | 0.03       | 0.17                     | 0.08       | 0.27                     | 0.03               | 0.17                     |
| Scandinavian Legal<br>Origin | 0.04   | 0.19         | 0.00              | 0.03         | 0.04       | 0.19                     | 0.11       | 0.32                     | 0.01               | 0.09                     |
| Population                   | 42352  | 139959       | 102351            | 229185       | 39634      | 118492                   | 17716      | 79404                    | 9705               | 16297                    |
| GDP Growth (%)               | 4.25   | 7.85         | 3.2               | 8.31         | 3.95       | 7.12                     | 4.55       | 8.57                     | 5.23               | 9.07                     |

| Table A2. | <b>Correlation Matrix</b> |  |
|-----------|---------------------------|--|

|                       | IP Index | Trade | DA Index | PR Index | GDP per capita | Population | Latitude | GDP per<br>capita growth |
|-----------------------|----------|-------|----------|----------|----------------|------------|----------|--------------------------|
| IP Index              | -        |       |          |          |                |            |          |                          |
| Trade                 | 0.36     | Ц     |          |          |                |            |          |                          |
| DA Index              | -0.46    | -0.14 | 1        |          |                |            |          |                          |
| PR Index              | -0.43    | -0.17 | 0.79     | П        |                |            |          |                          |
| GDP per capita        | 0.6      | 0.36  | -0.49    | -0.52    | 1              |            |          |                          |
| Population            | -0.09    | -0.49 | -0.03    | 0.16     | -0.15          | П          |          |                          |
| Latitude              | 0.32     | 0.08  | -0.46    | -0.31    | 0.51           | 0.2        | 1        |                          |
| GDP per capita growth | 0.15     | 0.09  | -0.06    | -0.05    | 0.10           | 0.05       | 0.08     | 1                        |

# Chapter 3 Catch-up in Institutional Quality: An Empirical Assessment

## Introduction

Since the 1980s, a key objective of reforms aimed at promoting economic growth and development has been to improve institutional quality. According to the World Bank (2000), dysfunctional and ineffective institutions and weak governance are "increasingly seen to be at the heart of the economic development challenge" and "building effective and accountable institutions is arguably the core challenge for sustainable poverty reduction" (p. 1). In response to this challenge, governments and multilateral agencies have shifted focus from getting prices right to getting institutions right in developing countries, often by emulating the institutions of developed countries (Rodrik, 2008). At the same time, rapid globalization has led to increased efforts to harmonize institutions across countries and to bind countries formally to common rules (Dolowitz & Marsh, 2000). This focus on a common set of institutions is exemplified by the creation of the World Trade Organization (WTO) in 1995 and the creation of trade unions such as the EU, NAFTA, and ASEAN.

Despite the resources and efforts devoted to institution building and institutional harmonization, there is little consensus on the effectiveness of these reforms (Andrews, 2013). Proponents argue that developing countries must adopt institutions from industrialized countries in order to bypass the stages that the latter already have been through (Mkandawire, 2009). Others argue that this "blueprint" approach may produce ineffective and poorly enforced rules and laws (Andrews, 2013). Moreover, the blueprint approach assumes that institutions from developed countries are optimal developmental instruments, but in fact they do not consider the unique socioeconomic conditions, local knowledge, and culture of the host country (Evans, 2004). The unsuccessful implementation of

governance-related conditions imposed by international financial institutions corroborates this view (Kapur & Webb, 2000).

Considering the emphasis on institutions in the development process and the widely reported challenges in reforming institutions, it is fitting to ask whether a catch-up in institutional quality has occurred. Scant empirical literature has explored development trends in institutional quality across countries (e.g., see Elert & Halvarsson, 2012; Nieswiadomy & Strazicich, 2004; Savoia & Sen, 2013). Nieswiadomy and Strazicich (2004) report sigma convergence using Freedom House's political rights and civil liberties indices. Savoia and Sen (2013) provide evidence of beta convergence based on analyzing the index for corruption and bureaucratic quality from the International Country Risk Guide and the index of legal system quality from the Fraser Institute's Economic Freedom of the World project. Elert and Halvarsson (2012) also report evidence of cross-country beta convergence in the Economic Freedom of the World index.

These studies use standard testing methods that assume catch-up is a continuous process or has already occurred. Thus, they provide an incomplete account of catch-up in institutional quality. These tests do not account for catch-up as a process that exists in certain time periods and that affects countries differently as countries differ in terms of their economic and social structure, culture, legal system, and capacity to enforce laws. This implies that not all countries follow a similar institutional change trajectory over time, rather it is likely that countries form groups with distinct institutional change trajectories. Moreover, previous research has not distinguished between economic and political institutions, which is important from policy and theoretical standpoints because the dynamics of change can be very different for economic institutions vis-à-vis political institutions.

This study contributes to the institutional change literature from two perspectives. First, in contrast to previous studies, this paper tests for catch-up in institutional quality and allows for heterogeneity in institutional change trajectories by grouping countries into clubs. Compared to previous studies, this approach offers a more nuanced view of institutional change by testing for the number and composition of clubs and then analysing their institutional change trajectories to determine whether catch-up has occurred. Clubs with weaker institutions should follow a faster institutional change trajectory, or catch-up, than that of clubs with good institutions. Building on previous studies, this paper uses a static factor model to test for catch-up in institutional quality by identifying clubs (e.g., see Andersson, Edgerton, & Opper, 2013; Henning, Enflo, & Andersson, 2011). Using this method identifies turning points in the institutional change trajectories of different clubs by providing information on the exact timing of improvements 54

or deterioration in institutions. Moreover, this method provides a way to investigate whether the catch-up in institutional quality is restricted to certain time periods or if it prevails throughout the entire observation period. This approach identifies the beginning and end of the catch-up process, as well as the speed of institutional catch-up across countries.

Second, this study departs from previous studies by differentiating between economic and political institutions and accounting for the differences in the dynamics of change observed in these institutions. This distinction is important, because theoretical literature on institutional change stresses that economic and political institutions affect political elites differently, which in turn affects how these institutions change. To capture these differences, this study uses three different measures of institutional quality using data from the KOF Index of Globalization, the Economic Freedom of the World index, and the International Country Risk Guide. The three measures focus on: (1) restrictions on trade and capital flows; (2) the quality of legal structure, security of property rights, freedom of exchange with foreigners, and regulation of credit, labor, and business; and (3) the quality of political institutions and rule of law. To enable comparisons across these three institutional indices, the factor model for each index is estimated using a joint sample of 81 countries from 1985 to 2010.

The results suggest different institutional change patterns for economic and political institutions. In the case of economic institutions, most countries experience institutional change in similar directions, and countries with weak institutions have a higher rate of change, indicating a catch-up process across countries. On the contrary, countries form two clubs for political institutions, with most low-income countries in one club and most high- and middle-income countries in the other. The trends for these two clubs indicate a pattern of catch-up at the start of the period, but the converging trend does not persist.

The rest of the paper is organized as follows. Section 2 provides a theoretical discussion on whether institutional quality may catch-up across countries and develops the empirical hypotheses related to institutional catch-up across countries. Section 3 outlines the estimation strategy. Section 4 introduces the data set, followed by results and conclusions in Sections 5 and 6.

# Theoretical background

The theoretical and empirical literature provides an ambiguous answer regarding catch-up in institutional quality across countries. It offers three different perspectives about institutional change and the possibility of institutional catch-up: the catch-up perspective, the lagging-behind perspective, and the mixed account.

### Catch-up Perspective

The catch-up perspective suggests several potential channels for institutional catch-up across countries. For example, international organizations like the International Monetary Fund (IMF) and World Bank impose policies and reforms that aim to improve the quality of institutions and governance (Dolowitz & Marsh, 2000; Kapur & Webb, 2000). These reforms often transplant previously accepted institutional blueprints for development (Evans, 2004; Rodrik, 2008). For example, the Washington Consensus in the 1990s aimed to improve the quality of economic institutions in the developing world (Rodrik, 2008). These policies include improved credit market policies, reduced taxes (tariffs) on international trade, and relaxed restrictions on the movement of capital (Gwartney, Lawson, & Block, 1996; Rodrik, 1999). Moreover, governments of many developed countries attach conditions to their bilateral aid, trade, and investment agreements (Chang, 2011) that require developing countries to adopt Western market institutions (Kapur & Webb, 2000; Stiglitz, 1999).

Countries tend to emulate the institutions of other countries. Emulation offers readily available blueprints and avoids the uncertainty of experimenting with new institutional arrangements. Developing countries can adopt institutions from developed countries and thereby use "better" institutions without paying the same "prices" (Chang, 2007). In theory, this lowers innovation costs and hastens diffusion, which enables institutional catch-up across countries (Mamadouh, De Jong, & Lalenis, 2002).

International economic integration and globalization can facilitate the exchange of ideas, including ideas about laws and regulations. Therefore, it may encourage the transfer and implementation of legal knowledge (La Porta, Lopez-de-Silanes, & Shleifer, 2008; Sachs, & Warner, 1995). Globalization alzz fosters competition among countries for foreign direct investment in capital and in business in general, exerting pressure on countries to adopt good legal rules and regulations and provide trade- and investment-friendly institutions, including privatization, deregulation, balanced budgets, low inflation, and strong property rights (Kelejian, Murrell, & Shepotylo, 2013; Marsh & Sharman, 2009). Importantly, domestic investors have an exit option (and foreign investors have the choice not to enter) and may thereby force governments to improve domestic institutions (Drezner, 2001; Holzinger & Knill, 2005). Thus, globalization may incentivize countries to improve their respective institutional environments in a global competition for capital and skilled labor. In sum, the above arguments suggest the following hypothesis:

Hypothesis 1: Countries with weak institutions will experience a greater change in their institutional quality and catch-up, compared to countries with strong institutions.

## Lagging-behind Perspective

Two main strands of literature question the effectiveness of these types of institutional reforms. The structural school criticizes their poor design and implementation. The political-economy school revolves around the willingness of elites to initiate and sustain them.

According to the structural view, emulation of a developed country's institutions may not fit a developing country's context and can conflict with the prevailing social and cultural context (Berkowitz, Pistor, & Richard, 2003a; Rodrik, 2008; Roland, 2004). When governments implement institutional reforms, either based on the blueprints provided by Western economists or through trial and error, the behavior of economic actors often disagrees with the formal rules. Instead, informal norms based on networks and culture shape the behavior of economic agents (Nee, 1998; Page & Bednar, 2006). Thus, reactions to new institutional environments may differ because of differences in cultures and norms. When reactions differ, performance may vary.

Legal institutions imported from the West are likely to be ineffective in developing countries unless they are adapted to the local context or some familiarity with those institutions already exists (Berkowitz et al., 2003a; Berkowitz, Pistor, & Richard, 2003b). Such rules, which were developed in a foreign socioeconomic order, may not apply to local circumstances (Berkowitz et al., 2003a). Thus, interpretation of the rule differs more within the borrowing country than it does within the country of origin, making the transplanted rules largely ineffective (Berkowitz et al., 2003b).

Governments in developing countries may not have the resources and capabilities to enforce these adopted institutions (Khan, 2012). Judges and police may not be trained and may be unfamiliar with the new institutional arrangements, and courts may be politicized and unpredictable, making the legal system expensive and inefficient (Hay, Shleifer, & Vishny, 1996). Thus, Berkowitz et al. (2003a) and Hay et al. (1996) argue that these blueprint reform efforts often fail because the existing institutional environment and laws conflict with the new institutions.

The political-economy stance suggests that institutional reforms in developing countries may be hindered by elites who benefit from existing economic and political institutions (Acemoglu, Johnson, & Robinson, 2005). These elites oppose institutional reforms that threaten their political power. In this situation, low-quality economic and political institutions can persist (Acemoglu & Robinson, 2012). Similarly, Sonin (2003) argues that the elite establish corrupt relationships with state authorities to manipulate redistributive processes. They consequently oppose measures to prevent corruption and protect the public. Resource inequality also enables the rich to subvert the political, regulatory, and legal institutions for their own benefit and leads them to favor established institutions over new efficient ones (Glaeser, Scheinkman, & Shleifer, 2003).

As North, Wallis, Webb, and Weingast (2009) assert, most of today's developing world is characterized by limited access, whereby only the elite enjoy access to and control of valuable resources. These limitations create rents for the elites, and the risk of losing the rents in a violent movement enourages the elite to cooperate rather than fight with the coalition in power (North et al., 2009). Hence, when reforms aim to transplant elements that are associated with the open-access order (e.g., competition, markets, democracy) directly into limited access orders, they fail. These arguments suggest the following hypothesis:

Hypothesis 2: Countries with weak institutions will not experience any significant improvement in their institutional quality and will lag behind.

## **Mixed Account**

A third strand of literature argues that the dynamics of change may differ depending on the nature of the institution being reformed. Multiple arguments suggest that economic institutions are more susceptible to change than are political institutions. First, the elites are more likely to undertake reforms that could change existing economic institutions, because commercialization and market liberalization provides new opportunities for their enrichment, as economic success provides increasing rents (Nye, 2011). The shift towards market institutions also increases opportunities among agents of the state, as it offers new economic niches (Nee & Lian, 1994). The introduction of market institutions removes political constraints on the accumulation of personal wealth and thus enables the elite to create new market value from access to or trading in public property (Walder, 2003). Moreover, the ruling class, especially an authoritarian ruler, benefits from providing secure property and contract rights, which expand the tax base, increase rents and market income accruing from asset ownership (McGuire & Olson, 1996; Olson, 1993). Concerns about the security of asset ownership in the event of a loss of political power provide an incentive for the elite to maintain secure property rights when institutional restrictions on expropriation are weak or absent (Polishchuk & Syunyaev, 2015). A notable example is China, where political actors allowed economic reforms instituted by economic actors, as the reforms also benefited the Chinese government through tax income, increased employment, and structural change (Nee & Opper, 2012; Taube, 2009).

The political elite can extend their tenure and secure continuing societal support by instituting economic reforms. For instance, political elites support economic reforms to deflect attention from political reforms and preserve their political power (Winiecki, 1996). Additionally, politicians and the political elite view economic reforms (especially those based on "best practices") as tool to garner short-term support (Andrews, 2012). This support typically comes from donors, but it also originates from local contexts. Implementing the "best practice" reform means more money in the national budget, better performance on global indicators of management and governance, and so on, all of which builds political support at home (Andrews, 2012). The political elites who hold power have an incentive to maintain the political institutions that give them political power (Acemoglu & Robinson, 2006, 2008). Therefore, political institutions persist when the political stakes are high or when alternative institutional arrangements are costly for those who currently hold political power and can use force to maintain existing political institutions (Acemoglu, 2006). This leads to the third hypothesis.

# Hypothesis 3: Compared to political institutions, economic institutions will exhibit a different pattern of change and will be more likely to catch-up.

Ultimately, whether economies with poor-quality institutions catch-up to economies with high-quality institutions and whether it happens in a political or an economic sphere is a matter of empirical debate. The theoretical literature does not provide a clear answer to this question. Therefore, this paper investigates whether a catch-up process has occurred in the institutional quality for countries with weak institutions.

# Methodology

To determine whether catch-up (in economic variables such as institutions, productivity, or income) across countries has occurred, researchers usually apply standard cross-sectional or time series tests. These tests rely on several restrictive assumptions. For instance, the cross-sectional approach assumes that all countries follow a universal model that governs the changes in institutional quality and imposes strict homogeneity restrictions (Bos, Economidou, Koetter, & Kolari, 2010). This approach does not allow countries to follow different trajectories in institutional change, even though some countries lack legal and judiciary systems and proper law enforcement. Given the differences in economic and social structures, it is unlikely that all countries follow an identical institutional change path and catch-up at the same speed. Alternatively, the time series approach tests if convergence has already happened, rather than if catch-up is occurring (Bernard & Durlauf, 1996; Carvalho & Harvey, 2005; Harvey & Bates, 2003).

Another possibility is that countries belong to clubs with different institutional change trajectories. Thus, instead of testing whether convergence has already occurred at the beginning of the period, a test should be performed for the presence of clubs that may only begin to catch-up at later stages of the observation period. Such a test can be constructed by using a factor model (Andersson et al., 2013; Andersson & Ljungberg, 2015; Henning et al., 2011). Factor analysis models covariance relationships among many variables in terms of a few underlying unobservable factors (Johnson & Wichern, 2007). Using a factor model allows grouping of countries based on similarities in institutional change path and helps identify if countries follow a common or idiosyncratic pattern of change.

A factor model alone is insufficient to test for catch-up in institutional quality, however. Distinguishing trends from short-term noise is also essential. Measurement error can occur in the institutional indices, because the institutional quality indices are based on subjective assessments by experts. Thus, the ratings may be influenced by knowledge of recent economic performance (Chang, 2011). Year-to-year changes in the data can mask the long-run catch-up pattern. Moreover, the literature on institutional change has established that changes in institutions are not instantaneous but rather take time. It is therefore highly likely that institutions across countries catch-up in the long term rather than the short term.

To separate the short run from the long run, institutional measures for each country are transformed using the Maximal Overlap Discrete Wavelet Transform

(MODWT).<sup>9</sup> The MODWT is a band pass filter that can be used to analyze the variation of a time series at different frequencies, where each frequency represents a separate time horizon, such as short run and long run (Andersson et al., 2013). Although it is possible to decompose data using several different methods, such as the Hodrick-Prescott filter or Fourier transformation, wavelet transformation offers considerable advantages. Wavelet decomposition combines time and frequency domains. It is localized both in time and in frequency, which preserves the time domain and frequency domain information of the original series (Maslova, Onder, & Sanghi, 2013). Thus, it does not introduce phase shifts that change the location of events in time<sup>10</sup> and allows for the observation of structural breaks, outliers, and nonlinearities in the data series (Percival & Walden, 2006; Ramsey, 1999).<sup>11</sup> Unlike the Hodrick-Prescott filter, in which data are decomposed into short and long runs, the filtered time horizons are known. This data-driven technique for separating short and long runs ensures that short-term fluctuations arising from measurement errors in the data do not affect the results.

The test for institutional catch-up is constructed as follows: let  $Y_{it}$  be the quality of institutions or a measure of policy for country *i* at time *t*. The change in institutional quality is given by  $\Delta y_{it} = \log(Y_{it}) - \log(Y_{it-1})$ , where the first difference is taken to avoid spurious results caused by non-stationarity in the data and where the log accounts for incremental institutional change as a country reaches its steady state. Taking the first difference does not affect our ability to analyze the trend in the data (e.g., see Brockwell & Davis, 1998; Percival & Walden, 2006).

The first step is to decompose the change in institutional quality  $\Delta y_{it}$  into a shortrun  $\Delta y_{it}^{SR}$  and a long-run  $\Delta y_{it}^{LR}$ , where  $\Delta y_{it} = \Delta y_{it}^{SR} + \Delta y_{it}^{LR, 12}$  The short-run period lasts up to eight years. The trend represents relatively persistent changes, with a period lasting for more than eight years. The distinction between the short and long runs is based on the results of the catch-up test, which provides evidence that change dynamics are different for the changes lasting more than eight years, compared to the changes lasting fewer than eight years. Because institutional change is slow, it makes sense to define the long run as a period longer than eight

<sup>&</sup>lt;sup>9</sup> MODWT transformation is performed in Matlab using WMTSA Wavelet Toolkit.

<sup>&</sup>lt;sup>10</sup> When using wavelet decomposition, one-off events such as crises do not affect the decomposition at other points in time. In contrast, with traditional smoothing techniques, such as the moving average, the impact of a one-off event spreads over several periods.

<sup>&</sup>lt;sup>11</sup> For a more detailed technical account of the MODWT, see Crowley (2007) and Andersson (2008).

<sup>&</sup>lt;sup>12</sup> The cyclical components represent cycles of 2–4 years and 4–8 years. By construction, MODWT decomposes the data such that each *jth* cycle is 2<sup>*j*</sup> long. The trend accounts for fluctuations higher than 2<sup>*j*</sup>.

years, as it reflects permanent change in institutional quality and provides smoother data with reduced noise in the institutional indices. Moreover, given the sample length, it is not possible to decompose the data into longer time horizons. Thus, these persistent movements form the  $\Delta y_{it}^{LR}$  part of  $\Delta y_{it}$ .

In the second step, the long-term change in institutional quality  $\Delta y_{it}^{LR}$  is modelled as a set of common factors and a country-specific idiosyncratic factor,

$$\Delta y_{it}^{LR} = \alpha_i + \sum_{r=1}^{q} \beta_{ir} f_{rt}^{LR} + \varepsilon_{it}^{LR} \qquad (1)$$

where  $\alpha_i$  is the country-specific constant, q is the number of common factors or trends,  $f_{rt}^{LR}$  denotes the q common factors or trends,  $\beta_{ir}$  represents factor loadings for the common factors, and  $\varepsilon_{it}^{LR}$  is the idiosyncratic component for each country.<sup>13</sup> The number of common factors indicates the number of clubs, and the factor loadings show which country belongs to which club (a country can belong to more than one club) and provide the contribution of each country to a common factor. To estimate this factor model, principal component analysis is used.

Three possible scenarios can be identified using the model given in equation (1). First, one major common factor could explain most of the variation in the change in institutional quality and the fact that loadings are non-zero and have a similar sign. In this case, institutions in all countries move together in a similar direction over time. However, this scenario does not imply an equal average change in institutional quality across all countries, because the constant for each country is different. Thus, some countries experience greater change than others, but the trajectory of change is the same. A second scenario includes one common factor in which the factor loadings have different signs for different countries. In this scenario, a divergence in institutional quality occurs. A third scenario is that q >1. For a subset of countries, the factor loadings are non-zero for a particular common factor. In this scenario, countries form independent clubs, with certain clubs experiencing greater institutional change than others. Institutional catch-up occurs if countries with poor-quality institutions experience a higher rate of change in their institutional quality than the countries with high-quality institutions.

To identify significant loadings and thus club composition, the analysis relies on the common convention of using 20% of explained variation as an appropriate

<sup>&</sup>lt;sup>13</sup> The factor model can be applied to short-run changes; however, short-run changes might contain noise.

cut-off (e.g., see Fidell & Tabachnick, 2006; Stevens, 2012).<sup>14</sup> Countries belong to a particular factor if that factor explains more than 20% of the variation for that country. Because the selection of a cut-off point is ultimately a normative decision, alternative cut-offs of 15% and 25% are also used to scrutinize the findings. However, the choice of these different cut-off points does not have a strong effect on the results, as shown in Tables A5 and A6 in the appendix.

## Data

To measure institutional quality and distinguish between the political and economic institutions, this study employs three measures of institutional quality using data from the Economic Freedom of the World index, the KOF Index of Globalization, and the Political Risk Index from International Country Risk Guide. A total of 81 countries are included from 1985 to 2010. Table A1 in the appendix summarizes the composition of the main indices and the indices used in this study.

The Economic Freedom of the World index reflects the quality of a country's economic institutional and policy environment (Gwartney, Holcombe, & Lawson, 2004). Several studies have used it to measure institutional quality (De Haan & Sturm, 2000; Gwartney et al., 2004; Hall, Sobel, & Crowley, 2010). However, De Haan, Lundström, and Sturm (2006) and De Haan and Sturm (2000) question its inclusion of government spending and monetary policy, as these variables do not necessarily restrict citizens' economic freedom and they reflect policy outcomes rather than rules. Following their critique, this study uses an index formulated by taking the average of the following sub-indices: legal structure and security of property rights; freedom to exchange with foreigners; and regulation of credit, labour, and business.<sup>15</sup> These three components more closely capture the restrictions on an individual's economic freedom and economic transactions, thus providing insight about the quality of economic institutions of a country. These three sub-indices are further divided into several sub-components, as explained in Table A2 in the appendix. This index will be referred to as the Economic Freedom index. The index is a continuous variable, ranging between zero and ten, with a higher score corresponding to a higher

<sup>&</sup>lt;sup>14</sup> According to Stevens (2012), a variable that shares at least 15% of the variation with the factor should be used for interpretation purposes

<sup>&</sup>lt;sup>15</sup> Following the construction of the Economic Freedom of the World's main index, an average is taken.

quality of institutional and policy environment. Data for this index have been recorded every five years from 1985 to 2000 and annually since 2001. A linear interpolation helps fill in data for missing years. This interpolation does not affect the results, as the interest lies in studying long-run changes (e.g., see Andersson & Ljungberg, 2015; Andersson & Karpestam, 2013).

The second measure is a sub-index of the economic globalization index, which is a part of the KOF globalization index (Dreher, 2006). The sub-index assesses restrictions on long-distance flows of goods, capital, and services and closely represents the "rules of the game." It is used to measure economic institutions, because it reflects the quality of institutions related to market liberalization and competition. Tariffs and import barriers divert resources to the government, and tariffs, quotas, and other trade barriers create lucrative opportunities for private diversion (Hall & Jones, 1999). Trade and capital flow restrictions are measured using data on hidden import barriers, mean tariff rates, taxes on international trade (as a share of current revenue), and an index of capital controls (Dreher, 2006). It takes a value between 0 and 100, where a higher value represents a lower degree of restrictions. This measure will be referred to as the Trade Restrictions index. Table A3 describes its sub-components.

The third index is based on different components of the political risk index from the International Country Risk Guide. The variables in the political risk index are the most commonly used measures of institutional quality in the empirical literature on institutions and growth (e.g., Knack & Keefer, 1995; Hall & Jones, 1999; Acemoglu, Johnson, & Robinson, 2001). They allow assessment of the political stability and government quality in a country (International Country Risk Guide, 2012). However, in addition to capturing institutional quality, the political risk index includes variables that capture economic performance (measured by socioeconomic conditions) and political violence and conflict (measured by external and internal conflict and religious and ethnic tension) that are not suitable for accessing the quality of institutions. Therefore, a sub-index is created by aggregating the scores of each country on the following components: investment profile, corruption, democratic accountability, law and order, and bureaucracy quality.<sup>16</sup> The resultant sub-index ranges from 0 to 34, with higher values indicating better quality of institutions. This sub-index allows assessment of the government's role as a protector against private diversion and as a diverter, as well as assessment of the restrictions on rulers to use power for personal benefit. Therefore, it is classified as measuring the quality of political institutions in a

<sup>&</sup>lt;sup>16</sup> It is created by following the same methodology as the complete index

country and referred to as the Government Quality index. Table A4 describes its components.

Because the aim of the study is to differentiate between economic and political institutions and to test for catch-up in them separately, it is important to investigate whether the indices actually measure different dimensions of institutions. The correlations indicate that this is the case. The average correlations of the Government Quality index, with the Trade Restrictions and Economic Freedom indices respectively, are 0.32 and 0.40. These are relatively low compared to the correlation of 0.75 between the Trade Restrictions and Economic Freedom indices. The correlations between these three indices indicate that the Trade Restrictions and Economic Freedom indices, whereas the Government Quality index captures a different dimension of institutional quality.

### Descriptive statistics

Table 1 shows a summary statistics of all three indices. Countries are divided into low-, middle-, and high-income to illustrate differences across level of economic development. Table 1 implies two stylized facts. First, the gap in institutional quality between high-income countries and low-income countries is relatively wide. The average score of high-income countries on the Government Quality index is approximately 1.75 times higher than that of low-income countries. Similarly, the average scores of high-income countries are 2.3 times and 1.55 times greater than those of low-income countries for the Trade Restrictions and Economic Freedom indices, respectively. Additionally, all three indices have increased over time since 1985, except for the Trade Restrictions index in advanced countries, which deteriorated marginally. On average, all three indices increased over the period.

Second, the cross-sectional dispersion over the whole sample (as expressed by the standard deviation) decreases from the beginning to the end of the period for every index. This finding indicates some catch-up in institutional quality over time but does not confirm that the difference decreased among all countries. For the Government Quality index, the decrease is monotonic until 1995 and then the dispersion picks up again or stabilizes, suggesting that a catch-up effect in institutional quality has stopped or decelerated. For the Economic Freedom index, the decrease in dispersion is relatively monotonic, whereas for the Trade Restrictions index, dispersion increases until 1995 and then decreases. According to the Government Quality and Economic Freedom indices, high-income

countries remain a more homogeneous group than low- and middle-income countries, which show greater variability in institutional quality over time. However, for the Trade Restrictions index, high-income countries have more variation than low- and middle-income countries. The apparent gap in institutional quality between advanced economies and the rest of the world implies that convergence has not yet occurred. The fact that the standard deviation in different groups does not decrease monotonically supports this finding. Thus, there is a need to test the catch-up hypothesis using a formal testing strategy.

|                                     |         | Full Sample | 1985 | 1990 | 1995 | 2000 | 2005 | 2010 |
|-------------------------------------|---------|-------------|------|------|------|------|------|------|
| Full Sample (N=2106)                | Mean    | 21.6        | 19.7 | 19.4 | 21.1 | 22.3 | 23.3 | 23.0 |
|                                     | St. Dev | 6.36        | 7.05 | 6.59 | 4.85 | 6.05 | 6.61 | 6.40 |
| High-Income Countries (N=676)       | Mean    | 28.4        | 27.9 | 26.7 | 26.4 | 28.2 | 30.8 | 30.2 |
|                                     | St. Dev | 3.55        | 3.92 | 4.21 | 2.85 | 3.19 | 2.71 | 2.58 |
| Middle-Income Countries (N=728)     | Mean    | 20.4        | 17.4 | 17.5 | 20.3 | 22.0 | 22.3 | 21.7 |
|                                     | St. Dev | 4.24        | 3.95 | 4.06 | 3.04 | 3.95 | 3.71 | 4.05 |
| Low-Income Countries (N=702)        | Mean    | 16.3        | 14.1 | 14.2 | 16.8 | 16.8 | 17.2 | 17.3 |
|                                     | St. Dev | 4.04        | 4.09 | 3.82 | 2.71 | 4.47 | 4.21 | 3.88 |
| Panel (b): Trade Restrictions index |         |             |      |      |      |      |      |      |
| Full Sample (N=2106)                | Mean    | 56.2        | 45.1 | 47.0 | 54.0 | 61.4 | 64.0 | 63.4 |
|                                     | St. Dev | 24.6        | 25.2 | 25.8 | 25.9 | 25.3 | 20.2 | 18.2 |
| High-Income Countries (N=676)       | Mean    | 7.67        | 72.6 | 75.2 | 79.2 | 87.1 | 82.6 | 78.4 |
|                                     | St. Dev | 17.5        | 17.6 | 17.9 | 18.4 | 18.2 | 15.6 | 15.7 |
| Middle-Income Countries (N=728)     | Mean    | 55.4        | 42.2 | 44.2 | 55.6 | 61.1 | 63.5 | 63.5 |
|                                     | St. Dev | 16.6        | 14.3 | 14.6 | 15.6 | 14.8 | 15.1 | 15.3 |
| Low-Income Countries (N=702)        | Mean    | 34.3        | 21.7 | 22.8 | 28.0 | 36.9 | 46.6 | 49.0 |
|                                     | St. Dev | 14.8        | 10.6 | 10.5 | 12.1 | 11.8 | 1.11 | 9.84 |
| Panel (c): Economic Freedom index   |         |             |      |      |      |      |      |      |
| Full Sample (N=2106)                | Mean    | 6.28        | 5.29 | 5.71 | 6.44 | 6.74 | 69.9 | 6.71 |
|                                     | St. Dev | 1.50        | 1.70 | 1.66 | 1.39 | 1.36 | 1.27 | 1.20 |
| High-Income Countries (N=676)       | Mean    | 7.80        | 7.17 | 7.53 | 7.88 | 8.25 | 8.03 | 7.89 |
|                                     | St. Dev | 0.80        | 0.89 | 0.89 | 0.87 | 0.67 | 0.59 | 0.64 |
| Middle-Income Countries (N=728)     | Mean    | 6.09        | 4.84 | 5.44 | 6.40 | 6.65 | 6.53 | 6.47 |
|                                     | St. Dev | 1.03        | 1.19 | 1.07 | 0.77 | 0.62 | 0.75 | 0.94 |
| Low-Income Countries (N=702)        | Mean    | 5.01        | 3.95 | 4.23 | 5.11 | 5.38 | 5.58 | 5.82 |
|                                     | St. Dev | 1.06        | 1.00 | 0.93 | 0.85 | 0.82 | 0.98 | 0.91 |

Table 1. Summary statistics of indices used over 1985-2010

## Results and discussion

This section presents results from the catch-up test. It discusses the extent to which each of the factors explains institutional change through variance decomposition and what characteristics explain a country's sensitivity to the factors. It also identifies patterns in the trajectory of institutional change for economic and political institutions.

#### Economic institutions

The first two factors account for most of the variation in the economic institutional indices, explaining 71% and 81% of all the variations in the trend for the Trade Restrictions and Economic Freedom indices, respectively (see Table 2). The high percentage of explained common variation in the trend shows a strong common world factor that drives institutional change for economic institutions for most countries in the sample. Consequently, in the long run, institutional changes are more likely to be affected by outside or global factors than by country-specific factors. The presence of a world factor is further supported by the fact that, for the Trade Restrictions and Economic Freedom indices, 71% and 76% respectively of the countries in the sample are significantly affected by the first factor with the majority of the loadings having a similar sign (see Table 3).<sup>17</sup>

Table 2

|                          | Factor 1 | Factor 2 | Factor 3 |
|--------------------------|----------|----------|----------|
| Trade Restrictions index | 42%      | 29%      | 14%      |
| Economic Freedom Index   | 54%      | 27%      | 10%      |
| Government Quality Index | 42%      | 19%      | 18%      |

Variation Explained by first three Factors

<sup>&</sup>lt;sup>17</sup> A country is significantly affected if 20% or more of the variation is explained by that particular factor.

#### Table 3

Percentage of high, middle and low-income countries loading significantly on the first factor for the economic institutions

|                         |                          | Economic Fr         | reedom index          |                     |
|-------------------------|--------------------------|---------------------|-----------------------|---------------------|
|                         | Fac                      | tor 1               | Fact                  | tor 2               |
| -                       | Positive<br>loading      | Negative<br>loading | Positive<br>loading   | Negative<br>loading |
| High-Income Countries   | 80%                      | 0%                  | 20%                   | 8%                  |
| Middle-Income Countries | 66.6%                    | 8.33%               | 33.3%                 | 25%                 |
| Low-Income Countries    | 70.4%                    | 6.81%               | 13.6%                 | 36.3%               |
|                         | Trade Restri<br>Factor 1 |                     | ictions index<br>Fact | tor 2               |
| -                       | Positive<br>loading      | Negative<br>loading | Positive<br>loading   | Negative<br>loading |
| High-Income Countries   | 80%                      | 0%                  | 0%                    | 24%                 |
| Middle-Income Countries | 58.3%                    | 8.33%               | 25%                   | 50%                 |
|                         |                          |                     |                       |                     |

Note: Within group percentages are reported; Income classification based on the year 1990

To identify which types of countries are significantly affected by the world factor, the countries are classified into groups based on their income level in 1990. As the result shows, countries in the high-, middle-, and low-income groups are equally affected by the first factor for both economic institution indices. The existence of a world factor is further reinforced by the average variation explained by the first factor, which is similar for all regions of the world, as shown in Table 4.<sup>18</sup> For the Trade Restrictions index, the average variation explained by the world factor ranges between 47% and 68%, whereas for the Economic Freedom index, it ranges between 39% and 56%. The results indicate that a common world factor drives changes in economic institutions in most countries and that most countries experience changes in similar directions.

<sup>&</sup>lt;sup>18</sup> Regional grouping is based on continents
|         | Full                           | Significant Countries     |                                |                           |
|---------|--------------------------------|---------------------------|--------------------------------|---------------------------|
| Region  | Trade<br>Restrictions<br>Index | Economic<br>Freedom Index | Trade<br>Restrictions<br>Index | Economic<br>Freedom Index |
| Africa  | 0.31                           | 0.38                      | 0.46                           | 0.50                      |
| America | 0.40                           | 0.58                      | 0.47                           | 0.64                      |
| Asia    | 0.31                           | 0.30                      | 0.53                           | 0.47                      |
| Europe  | 0.49                           | 0.48                      | 0.56                           | 0.56                      |
| Pacific | 0.29                           | 0.45                      | 0.39                           | 0.68                      |
| Total   | 0.38                           | 0.44                      | 0.50                           | 0.56                      |

 Table 4

 Region-wise explained variance by the first factor for the Trade Restrictions and Economic Freedom index

The presence of a common world factor underscores the fact that both developing countries with weak institutions and developed countries with strong institutions experience institutional change. Rapid globalization in the last few decades may explain this world factor. Globalization fosters competition among countries and results in a race to the top in institutional quality (Kelejian et al., 2013; Marsh & Sharman, 2009). The harmonization of national policies through international or supranational law, whereby countries must agree and comply with international rules through multilateral negotiations, also could have contributed to this process of co-movement in the institutional change across countries (Knill, 2005).

Most countries are significantly affected by the world factor. Others, such as those in the middle- and low-income groups (e.g., Malaysia, India, and Kenya) have institutional change trajectories that deviate from the very strong first common factor. To analyze what characteristics explain a country's sensitivity to global influences on institutional change, the proportion of variance explained by the world factor is regressed on several potentially important characteristics: (1) real GDP per capita in 2006; (2) average IMF loan participation from 1985 to 1995; (3) initial quality of the institutional index; (4) democratic accountability; (5) dummy for the EU; and (6) dummy for the North American countries. GDP per capita is used to explore whether the differences in variation explained by the world factor differs on the basis of economic development. Average IMF loan participation is added to test whether countries with high IMF loans are more likely to follow the world trend. Initial quality of economic institutions tests whether countries with poor economic institutions initially are more likely to be influenced by the world factor. Democratic accountability is added to check whether democratic countries are likely to be more affected by the world factor because of their high level of economic integration.

Table 5 presents the cross-sectional regression results for both economic institutional indices. For the Economic Freedom index, the initial level of the index is negative and significant, suggesting that countries with poor initial economic institutions are more affected by the world factor. GDP per capita and IMF loan participation are positively associated with the explained variance and are significant at 5% and 10% significance levels, respectively. Democratic accountability and regional dummy variables are insignificant. The result suggests that countries with higher IMF loan participation and higher GDP per capita tend to have higher sensitivity to the world factor. Pressure from the IMF to adopt institutions and policies based on the blueprint approach thus may be an effective way to bring about institutional change in developing countries.

#### Table 5

| Cross-sectional | regression   | results o | of variance and | constant | for Economic institutions |
|-----------------|--------------|-----------|-----------------|----------|---------------------------|
| Oroso sectional | - regression | reound    | or variance and | comotune |                           |

| Variables                      | Economic Freedom<br>Index | Trade Restrictions<br>Index |
|--------------------------------|---------------------------|-----------------------------|
| Economic Freedom 1985          | -0.11***                  |                             |
|                                | (0.03)                    |                             |
| IMF Loan Participation         | 0.23*                     | -0.12                       |
|                                | (0.12)                    | (0.11)                      |
| Log of GDP per capita          | 0.10***                   | -0.01                       |
|                                | (0.03)                    | (0.03)                      |
| Democratic Accountability 1985 | 0.04                      | 0.06**                      |
|                                | (0.03)                    | (0.02)                      |
| EU                             | 0.05                      | 0.17**                      |
|                                | (0.09)                    | (0.08)                      |
| North America                  | -0.09                     | 0.18                        |
|                                | (0.17)                    | (0.15)                      |
| Trade Restrictions1985         |                           | -0.003                      |
|                                |                           | (0.002)                     |
| Constant                       | 0.05                      | 0.32                        |
|                                | (0.21)                    | (0.20)                      |
| Observations                   | 81                        | 81                          |
| R-squared                      | 0.21                      | 0.18                        |

Standard errors in parentheses. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1

Table 6 also presents a similar picture. The bottom 15% of developing countries with minimal variance, as explained by the first factor for the Economic Freedom index, had relatively low IMF loan participation rates from 1985 to 1995. This includes China, Indonesia, India, Malaysia, and Thailand, which followed more "home-grown" initiatives in reforming their economic institutions (Easterly, 2006). By contrast, the top 15% of developing countries with the highest variance had an average IMF loan participation rate of 49%. Most of these countries belong to Latin America, such as Argentina and Peru, which underwent economic

reforms in the 1980s and 1990s as part of the Washington Consensus (Rodrik, 2006).

### Table 6

List of countries with highest and lowest variation explained for Economic Freedom index

|   | Variance     | IMF           |                     | Variance                | IMF            |
|---|--------------|---------------|---------------------|-------------------------|----------------|
|   | Explained by | participation |                     | Explained by            | participation  |
| Country   | first factor | rate          | Country             | first factor            | rate           |
| Top 12 countries with highest explained variation |              |               | Bottom 12 coun      | tries with lowest expla | ined variation |
| Peru  | 0.96         | 0.44          | Ivory Coast         | 0.00                    | 0.45           |
| Argentina   | 0.95         | 0.81          | Thailand            | 0.00                    | 0.28           |
| Tunisia   | 0.94         | 0.37          | Nigeria             | 0.00                    | 0.23           |
| Panama  | 0.94         | 0.5           | Papua New<br>Guinea | 0.00                    | 0.32           |
| El Salvador                                       | 0.94         | 0.49          | Malaysia            | 0.00                    | 0              |
| Philippines                                       | 0.93         | 0.91          | Bahrain             | 0.00                    | 0              |
| Jamaica   | 0.91         | 0.73          | Oman                | 0.00                    | 0              |
| Hungary   | 0.91         | 0.53          | Botswana            | 0.02                    | 0              |
| Trinidad and<br>Tobago                            | 0.89         | 0.15          | Indonesia           | 0.05                    | 0.15           |
| Nicaragua   | 0.84         | 0.1           | Uganda              | 0.08                    | 0              |
| Poland  | 0.78         | 0.37          | Kenya               | 0.10                    | 0.17           |
| Costa Rica  | 0.77         | 0.47          | India               | 0.12                    | 0.12           |

For the Trade Restrictions index, the results show that democratic countries respond more to the world factor influence, compared to other countries. Democratization reduces the ability of governments to use trade barriers as a strategy for building political support. Political leaders in labor rich countries may prefer lower trade barriers as democracy increases and may thus end up following global trends. Moreover, the dummy for EU is also significant, suggesting that EU countries have significantly higher explained variance compared to other countries. This finding again corresponds with the expectation, as EU countries are highly integrated in the world economy and are characterized by a single external tariff applied by all member states to imports from third countries. Apart from these two variables, no other variable is significant for the Trade Restrictions index.

The institutional change patterns of the countries affected by the world factor for the Trade Restrictions and Economic Freedom indices are shown in Figure 1. Low-income countries have a higher rate of change vis-a-vis high- and middleincome countries for both economic institutional indices. Countries with weak institutions experience a higher rate of change in their economic institutions.



Panel A: EconomicFreedom Index

#### Figure 1

Average change in Trade Restrictions and Economic Freedom index for the significant countries for World factor. (Note: Group 1 consists of low-income countries that load significantly and positively on the first factor and group 2 consists of high and middle-income countries).

If this pattern of change continues, countries with weak institutions will eventually catch-up with the countries with strong institutions. However, the catch-up process will be slow, as countries with strong institutions continue to improve their institutional quality. Moreover, catch-up in the rate of change does not imply a catch-up in levels. Figure 2 compares the scores of an average country from each group on both indices over time. As shown, a considerable gap exists between the scores of the two groups for both indices, showing that the convergence in levels has not yet happened, although the gap between the two groups is diminishing.





Long run evolution of economic institutions

The temporal evolution of the world factor for economic institutions also reveals some of the main global cyclical episodes of the past few decades that might have had an effect on the economic institutional quality of the countries. The average changes in the Trade Restrictions and Economic Freedom indices can also be used as a guide to study the direction of change for countries loading significantly on the world factor. For both indices, there are two noticeable periods of change. First, there is an upward trend from 1985 to 1993 for the Economic Freedom index and from 1985 to 1996 for the Trade Restrictions index. The upward trend supports a general move towards market-supporting institutions, low trade barriers, and fewer capital flow restrictions in the late 1980s and early 1990s. This is also reflected in the reduction in the average applied tariff rate in developing countries from 32% in 1984 to 16.6% in 1995<sup>19</sup> (see Figure 3) and by the relatively high rate of change in the Trade Restrictions index for developing countries during the same period. Moreover, the accelerated change in economic institutions in developing countries during the 1980s and 1990s coincides with the adoption of policies under the Washington Consensus. The upward trend also coincides with the introduction of the EU, NAFTA, and WTO, which may explain the changes in institutional quality in developed countries. These findings show that international organizations can influence domestic institutional quality and may help explain the co-movement in institutional change.



Figure 3 Average tariffs over the years

<sup>&</sup>lt;sup>19</sup> Source: World Bank trade databases: http://siteresources.worldbank.org/INTRES/Resources/tar2005.xls

The second noticeable change occurs in the late 1990s and early 2000s as changes in both indices started to decline and even became negative. The increased hidden trade barriers in the form of strict quality standards may explain this change (Beghin, 2006). The declining rate of change in these two indices could also be associated with the recurring economic and financial crises in both the developing and developed world. The counter-cyclical relationship between trade protectionism and the business cycle (Bown & Crowley, 2013; Knetter & Prusa, 2003) suggests that the financial crisis during this period might have forced countries to impose capital flow restrictions and resort to protectionist trade policies. Moreover, the decline in the change in the Trade Restrictions index after 1996 indicates that, although the introduction of the WTO had a positive effect in the short run, this effect may have diminished over the years.

## **Political Institutions**

Similar to economic institutions, the first two factors explain 61% of the variation in changes in political institutions (see Table 2). However, the long-term trend for political institutions exhibits different clustering patterns than those observed for economic institutions. The first and second factors have a significant impact on 55% and 40% of the countries, respectively, reflecting a lack of a common world factor. Rather, two main factors drive the changes in political institutions, and different groups of countries respond to different common factors, forming two clusters. The result suggests that political institutions follow a different pattern of change than economic institutions and confirms the third hypothesis. It also suggests that political elites might respond differently to different types of institutional reform because of their vested interests and the fear of redistribution of their political power.

The first cluster comprises 33 countries that load positively and significantly on the first factor and are primarily low- or lower-middle-income countries (see Table 7). Additionally, the first group includes four middle-income countries and two high-income countries (Greece, Hungary, Korea, Portugal, Cyprus, and Israel). These countries were similar to the developing countries in terms of their institutional quality in the 1990s, with high levels of corruption, poor bureaucracy, and low levels of democratic accountability. However, over the years they have experienced significant improvements in these components.

#### Table 7

|                          | Fac                 | tor 1               | Fac                 | tor 2               |
|--------------------------|---------------------|---------------------|---------------------|---------------------|
|                          | Positive<br>loading | Negative<br>loading | Positive<br>loading | Negative<br>loading |
| Income Classification    |                     |                     |                     |                     |
| High-Income Countries    | 8%                  | 32%                 | 60%                 | 0%                  |
| Middle-Income Countries  | 33.3%               | 8.33%               | 41.6%               | 0%                  |
| Low-Income Countries     | 61.3%               | 6.81%               | 13.6%               | 15.9%               |
| Democracy Classification |                     |                     |                     |                     |
| Free Democracies         | 27.5%               | 27.5%               | 47.5%               | 2.50%               |
| Partly Free Democracies  | 69.5%               | 4.34%               | 2.17%               | 4.34%               |
| Not-Free Democracies     | 33.3%               | 0%                  | 11.1%               | 27.8%               |

Percentage of different types of countries loading significantly on the first factor two factors for the Government Quality index

Note: Within group percentages are reported; Income classification based on the year 1990; Democracy classification based on Freedom House category of democracy in 1985

Among high-income countries, 60% load positively and significantly on the second factor, together with 41% of middle-income countries and 13% of low-income countries. The fact that countries follow common trends in terms of changes in their political institutions underscores the fact that most countries experience common changes in institutional quality despite the two distinct underlying trends. The division of countries into two clubs highlights a close linkage between the initial quality of political institutions and subsequent club formation. Countries with weak political institutions are significantly affected by the first factor, whereas countries with strong institutions form the second group. This implies the existence of path dependence in political institutions.

The classification of countries into these two groups is further supported by the regression results of the proportion of variance explained by the first and second factors on the income classification, as well as on the initial level of the Government Quality index. As shown in Table 8, low- and middle-income countries have significantly higher explained variance compared to high-income countries. In contrast, for factor 2 high-income countries have significantly higher explained variance compared to Government Quality renders the coefficient of dummy variables insignificant. However, the coefficient on Government Quality is negative for factor 1 and positive for factor 2, indicating that the first factor belongs to countries with poor initial political

institutions and that the second factor belongs to countries with good initial political institutions.

### Table 8

Cross-sectional regression results of variance and constant for Government Quality index

| Variables     | Variance<br>Factor 1 | Variance<br>Factor 2 | Variance<br>Factor 1 | Variance<br>Factor 2 | Constant | Constant |
|---------------|----------------------|----------------------|----------------------|----------------------|----------|----------|
| Middle Income | 0.14*                | 0.02                 | 0.001                | -0.04                | 0.70**   | -1.47*** |
| Countries     | (0.07)               | (0.06)               | (0.10)               | (0.06)               | (0.29)   | (0.28)   |
| Low Income    | 0.14*                |                      | -0.046               |                      | 0.77**   | -2.09*** |
| Countries     | (0.07)               |                      | (0.12)               |                      | (0.29)   | (0.33)   |
| High Income   |                      | 0.27***              |                      | 0.02                 |          |          |
| Countries     |                      | (0.06)               |                      | (0.10)               |          |          |
| ICRG 1985     |                      |                      | -0.01*               | 0.01***              |          | -0.21*** |
|               |                      |                      | (0.006)              | (0.005)              |          | (0.02)   |
| Constant      | 0.25***              | 0.17***              | 0.651***             | -0.12                | 0.280    | 6.07***  |
|               | (0.054)              | (0.05)               | (0.219)              | (0.11)               | (0.210)  | (0.55)   |
| Observations  | 81                   | 81                   | 81                   | 81                   | 81       | 81       |
| R-squared     | 0.06                 | 0.21                 | 0.10                 | 0.30                 | 0.10     | 0.64     |

Standard errors in parentheses. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1

Countries with weak institutions form a single cluster, because it is highly likely that they follow a trajectory determined by their past institutions. The ruling elites may not replace them with better quality institutions if they have a vested interest in maintaining the status quo (Acemoglu & Robinson, 2012). This result contrasts with the results for economic institutions, for which a common world factor dominates. It also confirms a difference in the patterns of change for economic and political institutions, as postulated by the third hypothesis.

Figure 4 presents the average long-run change in the quality of political institutions for both the developed and developing country groups. The institutional change path for the developing country group indicates that these countries experienced improvements in their political institutions in the late 1980s and most of the 1990s. This initial change coincides with the third wave of democracy<sup>20</sup>, which restricted the arbitrary actions of rulers and bureaucrats (Diamond, 1996; Huntington, 1993). Many developing countries in Latin America and Asia underwent a process of democratization in the 1980s and 1990s that in turn might have helped to reduce corruption and secure property rights (Knutsen, 2011; Rock, 2009). Moreover, the reforms bacad on aid assistance,

<sup>&</sup>lt;sup>20</sup> For a detailed discussion on the waves of democracy, see Huntington (1993).

which were targeted at improving the quality of governance in developing countries during the 1980s and 1990s, may have contributed to this initial positive change observed for the developing country group (Naim, 2000; Andrews, 2012).



#### Figure 4

Average change in Government Quality index for developing and developed countries

The developed country group experienced a positive change in the early 1990s, coinciding with the introduction of the EU and NAFTA. Joining a regional economic cooperation arrangement requires member countries to comply with a range of policy requirements, which in turn improves policy credibility (World Bank, 2004). The improved credibility often arises from improved institutional quality and has a direct impact on the investment climate of a country. Accordingly, the average long-run change in the investment profile component of the Government Quality index for the developed country group has a correlation of 0.86 with the second factor, which shows that changes in the investment profile drove changes in the political institutions index.

The developing country group experienced a higher average change in the quality of political institutions than the developed country group, thereby gradually

reducing the gap in institutional quality (in absolute terms) between the two groups. This result is confirmed by regressing the constant on the initial level of the political institution as well as on the income group dummies (see Table 8). Results confirm that countries with a poor initial level of the political institution experienced higher institutional change. The result, however, does not indicate that catch-up occurred between these two groups, as the acceleration in institutional change did not last long and seems to have slowed down or even disappeared in the new millennium. This weakening of the catch-up process in recent periods instead seems compatible with the view that the adoption of Western-style institutions in the global South may not have been as successful as expected, because of developing-economy-specific constraints (Berkowitz et al., 2003a, b; Roland, 2004; Rodrik, 2008; Khan, 2012).

Another plausible explanation for the slowdown of catch-up could be that political elites, influential minorities, or interest groups in the developing countries oppose these changes, especially in the political context (Acemoglu & Robinson, 2006, 2008). Therefore, any external or internal influences to change de jure institutions may leave the sources of de facto power intact, and groups that have lost their de jure power may use their de facto power to re-create a similar system to the previous one (Acemoglu & Robinson 2006, 2008). This process eventually leads to the reversal of policies and any associated reforms.

In sum, the results for all three indices indicate that most countries experience comovement in institutional change, irrespective of their level of development or other characteristics. This finding confirms that policymakers and governments often look outwards, imitating policy elements from other countries and emulating institutions from elsewhere, instead of designing policies and institutions to meet domestic goals and interests and fit the domestic culture. This finding holds true even for the developed countries in the sample, as they also respond significantly to the common trend. Another important implication of the results is that the changes in economic institutions follow a global trend. However, the changes in economic institutions are higher for developing countries than for developed countries, suggesting that political elites in many developing countries may want to change these institutions so as to increase their rents and maintain their legitimacy. To protect the status quo, these changes happen in the economic sphere only and not in the political arena. This result partially confirms the first hypothesis that a catch-up in institutional quality occurs across countries, at least in the economic sphere. Moreover, the higher rate of change for economic institutions is in line with the third hypothesis that states that economic institutions are more susceptible to change and more likely to converge than are political institutions.

## Conclusion

Rapid globalization in recent years and a shift in policy focus towards getting institutions right, have raised the following questions: (1) To what extent does a catch-up in institutional quality occur between countries? (2) Are contemporary differences in institutional quality between countries transitory or permanent? The theoretical and empirical literature does not answer these questions. This study uses a static factor model combined with MODWT and three different institutional measures to test for long-term catch-up in institutional quality.

First, the analysis shows that the pattern of change for economic institutions is different from that for political institutions. Changes in the economic institutions are driven by a common world factor, and most countries with weak institutions experience greater institutional change vis-à-vis countries with strong institutions. Hence, differences in institutional quality between countries may be transitory, but they persist for a long time because countries with strong institutions also experience institutional change. Two main trends occur for political institutions: one trend for countries with weak political institutions and another for countries with strong institutions. The evolution of the trends shows that catch-up occurs at the start of the period. However, the acceleration in catch-up is short-lived and seems to slow quickly and even disappear in the new millennium. Second, the trajectory of institutional change experienced by many countries coincides with some of the major trends and events in the last three decades. These include rapid globalization, the introduction of the EU and NAFTA, the emphasis on policies based on the Washington Consensus, the third wave of democracy, and the financial crisis of 2008.

These findings are consistent with the theoretical assertions that economic institutions are more susceptible to change and more likely to converge than are political institutions. This underscores the theoretical argument that political elites are likely to undertake reforms in economic institutions, as such changes directly benefit the ruling elites by increasing their rents and tax base and elongating their tenure (Nye, 2011; Nee & Lian, 1994; Andrews, 2012; McGuire & Olson, 1996). As for political institutions, enhancing efficiency and improving institutions may not supplant weak institutions in developing countries. These institutions are likely to be path dependent, as it is not in the interest of the political elites to change the existing extractive political institutions (Acemoglu & Robinson, 2012). Based on these findings, future research on institutions and institutional change should categorically differentiate between economic and political institutions and use different measures for them.

The findings also have implications for national and global reform policy in developing countries. Contrary to the general understanding that country-specific factors affect institutional change, evidence indicates that common global factors affect institutional change locally in the long run. Thus, global efforts to harmonize institutions across countries and force countries to adopt certain policies and institutions may lead to institutional catch-up in the long run, at least when such policies aid at improving market-oriented economic institutions. However, the institutional reforms promoted by international financial institutions based on the blueprint approach might not succeed in the long run for political institutions. In this scenario, instead of imposing a single blueprint model based on the experiences of developed countries, policy makers should seek ways to introduce more context-specific reforms that account for the domestic policy environment, the social and economic context, and the prevailing social norms.

A few limitations of the analysis should be noted. First, the study sample contains only 81 countries. Expanding the sample might enable future researchers to better understand the reasons behind common variations and better discern patterns of institutional change. A second data limitation is that the data set spans only 26 years. This range limits decomposition of the long-run institutional change, which lasted for more than 16 years. Future studies should use a more current data set to analyze the catch-up pattern in long-run institutional change. Last, the analysis dates back to 1985 and thus does not identify the impact of prior significant economic and political events, such as the ascendance of communism, Latin America's debt crisis, the 1970s oil crises, and the popularity of import substitution policies on the development and evolution of both political and economic institutions in the global South.

## References

- Acemoglu, D. (2006). Modeling inefficient institutions. *National Bureau of Economic Research*.
- Acemoglu, D., Johnson, S., & Robinson, J. A. (2001). The Colonial Origins of Comparative Development: An Empirical Investigation. *The American economic review*, 91(5), 1369-1401.
- Acemoglu, D., Johnson, S., & Robinson, J. (2005a). Institutions as a Fundamental Cause of Long-Run Growth. In A. Philippe & N. D. Steven (Eds.), *Handbook of economic growth* (Vol. Volume 1, Part A, pp. 385-472): Elsevier.
- Acemoglu, D., & Robinson, J. A. (2006). De facto political power and institutional persistence. *The American economic review*, 325-330.
- Acemoglu, D., & Robinson, J. A. (2008). Persistence of Power, Elites and Institutions. *American Economic Review*, 98(1).
- Acemoglu, D., & Robinson, J. A. (2012). Why Nations Fail: The Origins of Power, Prosperity, and Poverty. New York NY: Crown Business
- Andersson, F.N.G. (2008), *Wavelet Analysis of Economic Time Series*, (diss.), Lund Economic Studies, 149.
- Andersson, F. N. G., Edgerton, D. L., & Opper, S. (2013). A matter of time: revisiting growth convergence in China. *World Development*, 45, 239-251.
- Andersson, F. N. G., & Karpestam, P. (2013). CO<sub>2</sub> emissions and economic activity: Short-and long-run economic determinants of scale, energy intensity and carbon intensity. *Energy Policy*, 61, 1285-1294.
- Andersson, F. N.G., & Ljungberg, J. (2015). Grain market integration in the Baltic Sea Region in the 19th century. *The Journal of Economic History*.
- Andrews, M. (2013). The limits of institutional reform in development: changing rules for realistic solutions: Cambridge University Press.
- Barro, R. J., & Lee, J. W. (2005). IMF programs: Who is chosen and what are the effects?. *Journal of Monetary Economics*, *52*(7), 1245-1269.
- Beghin, J. C. (2006). Nontariff barriers. *Center for Agricultural and Rural Development, Iowa State University*.
- Berkowitz, D., Pistor, K., & Richard, J.-F. (2003a). Economic development, legality, and the transplant effect. *European Economic Review*, 47(1), 165-195.
- Berkowitz, D., Pistor, K., & Richard, J.-F. (2003b). The transplant effect. *The American Journal of Comparative Law*, 163-203.

- Bernard, A. B., & Durlauf, S. N. (1996). Interpreting tests of the convergence hypothesis. *Journal of econometrics*, 71(1), 161-173.
- Bos, J. W., Economidou, C., Koetter, M., & Kolari, J. W. (2010). Do all countries grow alike? *Journal of Development Economics*, 91(1), 113-127.
- Bown, C. P., & Crowley, M. A. (2013). Import protection, business cycles, and exchange rates: evidence from the Great Recession. *Journal of International Economics*, 90(1), 50-64.
- Brockwell, P. J., & Davis, R. A. (1998). *Time series: theory and methods*. Springer Series in Statistics, New York.
- Carvalho, V. M., & Harvey, A. C. (2005). Growth, cycles and convergence in US regional time series. *International Journal of Forecasting*, 21(4), 667-686.
- Chang, H.-J. (2011). Institutions and economic development: theory, policy and history. *Journal of Institutional Economics*, 7(04), 473-498.
- Chang, H-J. (2007). Understanding the relationship between institutions and economic development, in Chang, Ha-Joon (ed.), *Institutional Change and Economic Development*, New York: United Nations University Press: 17-34.
- Crowley, P. (2007). A guide to wavelets for economists. *Journal of Economic Surveys*, 21, 207–267.
- De Haan, J., Lundström, S., & Sturm, J.-E. (2006). Market-oriented institutions and policies and economic growth: A critical survey. *Journal of Economic Surveys*, 20(2), 157-191.
- De Haan, J., & Sturm, J.-E. (2000). On the relationship between economic freedom and economic growth. *European Journal of Political Economy*, *16*(2), 215-241.
- Diamond, L. J. (1996). Is the third wave over? Journal of democracy, 7(3), 20-37.
- Dolowitz, D. P., & Marsh, D. (2000). Learning from abroad: The role of policy transfer in contemporary policy-making. *Governance*, 13(1), 5-23.
- Dreher, A. (2006). Does globalization affect growth? Evidence from a new index of globalization. *Applied Economics*, *38*(10), 1091-1110.
- Drezner, D. W. (2001). Globalization and policy convergence. *International studies review*, *3*(1), 53-78.
- Easterly, W.R. (2006). *The white man's burden: why the West's efforts toaid the rest have done so much ill and so little good.* New York:Penguin Press.
- Elert, N., & Halvarsson, D. (2012). Economic freedom and institutional convergence. *Ratio working paper*(196).
- Evans, P. (2004). Development as institutional change: the pitfalis of monocropping and the potentials of deliberation. *Studies in comparative international development*, *38*(4), 30-52.

- Fidell, L. S., & Tabachnick, B. G. (2006). Using multivariate statistics. New York: Harper and Row.
- Glaeser, E., Scheinkman, J., & Shleifer, A. (2003). The injustice of inequality. *Journal of Monetary Economics*, 50(1), 199-222.
- Gwartney, J., Lawson, R., & Block, W. (1996). Economic freedom of the world. Canada: Fraser Institute.
- Gwartney, J., Lawson, R., & Hall, J. (2012). Economic Freedom of the World: 2012 Annual Report.
- Gwartney, J. D., Holcombe, R. G., & Lawson, R. A. (2004). Economic freedom, institutional quality, and cross-country differences in income and growth. *Cato J.*, 24, 205.
- Hall, R. E., & Jones, C. I. (1999). Why Do Some Countries Produce So Much More Output Per Worker Than Others?. *The Quarterly Journal of Economics*, 114(1), 83-116.
- Hall, J. C., Sobel, R. S., & Crowley, G. R. (2010). Institutions, capital, and growth. *Southern Economic Journal*, 77(2), 385-405.
- Harvey, A., & Bates, D. (2003). *Multivariate unit root tests and testing for convergence*: University of Cambridge, Department of Applied Economics.
- Hay, J. R., Shleifer, A., & Vishny, R. W. (1996). Toward a theory of legal reform. *European Economic Review*, 40(3), 559-567.
- Henning, M., Enflo, K., & Andersson, F. N.G. (2011). Trends and cycles in regional economic growth: How spatial differences shaped the Swedish growth experience from 1860–2009. *Explorations in Economic History*, 48(4), 538-555.
- Holzinger, K., & Knill, C. (2005). Causes and conditions of cross-national policy convergence. *Journal of European public policy*, *12*(5), 775-796.
- Huntington, S. P. (1993). The third wave: Democratization in the late twentieth century (Vol. 4): University of Oklahoma press.
- ICRG (2012). International Country Risk Guide Researchers Dataset, Table 3B. The Political Risk Services Group, third edition.
- Johnson, R. A., & Wichern, D. W. (2007). *Applied multivariate statistical analysis* (6th ed.). New York: Prentice Hall.
- Kapur, D., & Webb, R. (2000). Governance-Related Conditionalities Of The International Financial Institutions.
- Kelejian, H. H., Murrell, P., & Shepotylo, O. (2013). Spatial spillovers in the development of institutions. *Journal of Development Economics*, 101, 297-315.

- Khan, M. (2012). Governance and Growth: History, Ideology, and Methods of Proof. In A. Noman, K. Botchwey, H. Stein, & J.E. Stiglitz (Eds), *Good Growth and Governance in Africa: Rethinking Development Strategies*, 51-79.
- Knack, S., & Keefer, P. (1995). Institutions and economic performance: cross- country tests using alternative institutional measures. *Economics and Politics*, 7(3): 207– 228.
- Knetter, M. M., & Prusa, T. J. (2003). Macroeconomic factors and antidumping filings: evidence from four countries. *Journal of International Economics*, 61(1), 1-17.
- Knill, C. (2005). Introduction: Cross-national policy convergence: concepts, approaches and explanatory factors. *Journal of European public policy*, *12*(5), 764-774.
- La Porta, R., Lopez-de-Silanes, F., & Shleifer, A. (2008). The economic consequences of legal origins. *Journal of Economic Literature*, 46(2), 285-332.
- Knutsen, C. H. (2011). Democracy, dictatorship and protection of property rights. *The Journal of Development Studies*, 47(1), 164-182.
- Mamadouh, V., De Jong, M., & Lalenis, K. (2002). An introduction to institutional transplantation *The theory and practice of institutional transplantation* (pp. 1-16): Springer.
- Marsh, D., & Sharman, J. C. (2009). Policy diffusion and policy transfer. *Policy studies*, 30(3), 269-288.
- Maslova, I., Onder, H., & Sanghi, A. (2013). Growth and volatility analysis using wavelets. *World Bank Policy Research Working Paper*(6578).
- McGuire, M. C., & Olson, M. (1996). The economics of autocracy and majority rule: the invisible hand and the use of force. *Journal of Economic Literature*, 72-96.
- Mkandawire, T. (2009). *Institutional monocropping and monotasking in Africa*: Oxford University Press.
- Naim, M. (2000). Fads and fashion in economic reforms: Washington Consensus or Washington Confusion? *Third World Quarterly*, 21(3), 505-528.
- Nee, V. (1998). Norms and networks in economic and organizational performance. *American Economic Review*, 85-89.
- Nee, V., & Lian, P. (1994). Sleeping with the enemy: A dynamic model of declining political commitment in state socialism. *Theory and Society*, 23(2), 253-296.
- Nee, V., & Opper, S. (2012). *Capitalism from below: Markets and institutional change in China*: Harvard University Press.
- Nieswiadomy, M. L., & Strazicich, M. C. (2004). Are Political Freedoms Converging? *Economic Inquiry*, 42(2), 323-340.

- North, D. C., Wallis, J. J., Webb, S. B., & Weingast, B. R. (2009). Limited Access Orders: Rethinking the Problems of Development and Violence. *Unpublished working paper*.
- Nye, J. V. (2011). Why Do Elites Permit Reform? *The Annual Proceedings of the Wealth and Well-Being of Nations*, 53.
- Olson, M. (1993). Dictatorship, Democracy, and Development. *American political science review*, *87*(03), 567-576.
- Page, S. E., & Bednar, J. (2006). Culture, Institutional Performance, and Path Dependence. Working Paper No. WP2006–6. Institute of Governmental Studies, University of California, Berkeley, CA.
- Percival, D. B., & Walden, A. T. (2006). *Wavelet methods for time series analysis* (Vol. 4): Cambridge University Press.
- Polishchuk, L., & Syunyaev, G. (2015). Ruling elites' rotation and asset ownership: implications for property rights. *Public Choice*, *162*(1), 159-182.
- Ramsey, J. B. (1999). The contribution of wavelets to the analysis of economic and financial data. *Philosophical Transactions of the Royal Society of London A: Mathematical, Physical and Engineering Sciences, 357*(1760), 2593-2606.
- Rock, M. T. (2009). Corruption and democracy. *Journal of Development Studies*, 45(1), 55-75.
- Rodrik, D. (2006). Goodbye Washington Consensus, Hello Washington Confusion? A Review of the World Bank's "Economic Growth in the 1990s: Learning from a Decade of Reform". *Journal of Economic Literature*, 44(4), 973-987.
- Rodrik, D. (1999). Governing the Global Economy: Does One Architectural Style Fit All? *Brookings Trade Forum* (pp. 105-139): Brookings Institution Press.
- Rodrik, D. (2008). Second-Best Institutions. *American Economic Review*, 98(2), 100-104.
- Roland, G. (2004). Understanding institutional change: fast-moving and slow-moving institutions. *Studies in comparative international development*, *38*(4), 109-131.
- Sachs, J. D., & Warner, A. (1995). Economic reform and the process of global integration. *Brookings papers on economic activity*, 1-118.
- Savoia, A., & Sen, K. (2013). Do We See Convergence in Institutions: A Cross-Country Analysis. *Development Economics and Public Policy Working Paper Series*(33).
- Sonin, K. (2003). Why the rich may favor poor protection of property rights. *Journal of comparative economics*, *31*(4), 715-731.
- Stevens, J. P. (2012). Applied multivariate statistics for the social sciences: Routledge.
- Stiglitz, J. E. (1999). More instruments and broader goals: moving toward the Post-Washington consensus. *Revista de Economia Política, 19*(1), 94-120.

- Taube, M. (2009). Principles of property rights evolution in China's rural industry. In:
   T. Heberer & G. Schubert (ed.) (2009), *Regime Legitimacy in Contemporary China. Institutional Change and Stability*, London-New York: Routledge.
- Walder, A. G. (2003). Elite opportunity in transitional economies. *American Sociological Review*, 899-916.
- Winiecki, J. (1996). Why economic reforms fail in the Soviet system: a property rightsbased approach. In L.J. Alston, T Eggertsson &D.C. North, *Empirical Studies in Institutional Change*. Cambridge University Press, Cambridge.
- World Bank. (2004). World Development Report 2005: A Better Investment Climate for Everyone. Washington, DC: World Bank.
- World Bank. (2000). Reforming Public Institutions and Strengthening Governance. A World Bank Strategy. Washington, DC: World Bank.

## Appendix

| Source                              | Composition  |  |  |  |
|-------------------------------------|--|--|--|--|
| Economic Freedom of the World Index | Economic Globalization                                   |  |  |  |
|                                     | Index of Acctual Economic Flows                          |  |  |  |
|                                     | Index of Economic Restrictions *                         |  |  |  |
|                                     | Social Globalization                                     |  |  |  |
|                                     | Political Glibalization                                  |  |  |  |
| KOF Index of Globalization          | Size of Government: Expenditures, Taxes, and Enterprises |  |  |  |
|                                     | Legal Structure and Security of Property Rights *        |  |  |  |
|                                     | Access to Sound Money                                    |  |  |  |
|                                     | Freedom to Trade Internationally *                       |  |  |  |
|                                     | Regulation of Credit, Labor, and Business *              |  |  |  |
| Political Risk Index                | Government Stability                                     |  |  |  |
|                                     | Socioeconomic Conditions                                 |  |  |  |
|                                     | Investment Profile *                                     |  |  |  |
|                                     | Internal conflict  |  |  |  |
|                                     | External conflict  |  |  |  |
|                                     | Corruption *   |  |  |  |
|                                     | Military in politics                                     |  |  |  |
|                                     | Religious Tensions                                       |  |  |  |
|                                     | Law and Order *  |  |  |  |
|                                     | Ethinic Tensions   |  |  |  |
|                                     | Democratic Accountablity *                               |  |  |  |
|                                     | Bureaucracy Quality *                                    |  |  |  |

#### Table A1: Composition of the Indices

Note: Components with \* are the ones used in this study

| Variable   | Description  |
|--|--|
| Data on Restrictions   |  |
| Hidden Import Barriers                                       | The index is based on the Global Competitiveness Report's<br>survey question: "In your country, tariff and non-tariff barriers<br>significantly reduce the ability of imported goods to compete in<br>the domestic market." The question's wording has varied slightly<br>over the years.  |
| Mean Tariff Rate   | As the mean tariff rate increases, countries are assigned lower<br>ratings. The rating declines toward zero as the mean tariff rate<br>approaches 50%.   |
| Taxes on International Trade<br>(percent of current revenue) | Taxes on international trade include import duties, export duties,<br>profits of export or import monopolies, exchange profits, and<br>exchange taxes. Current revenue includes all revenue from taxes<br>and non-repayable receipts (other than grants) from the sale of<br>land, intangible assets, government stocks, or fixed capital assets,<br>or from capital transfers from nongovernmental sources. It also<br>includes fines, fees, recoveries, inheritance taxes, and non-<br>recurrent levies on capital. Data are for central government and ir<br>percent of all current revenue.                                  |
| Capital Account Restrictions                                 | Index based on two components: (i) Beginning with the year<br>2002, this sub-component is based on the question: "Foreign<br>ownership of companies in your country is (1) rare, limited to<br>minority stakes, and often prohibited in key sectors or (2)<br>prevalent and encouraged". For earlier years, this sub-component<br>was based on two questions about "Access of citizens to foreign<br>capital markets and foreign access to domestic capital markets".<br>(ii) Index based on the IMF's Annual Report on Exchange<br>Arrangements and Exchange Restrictions, including 13 different<br>types of capital controls. |

Table A2: Variables comprising the Trade Restrictions index

Source: Dreher (2006), Dreher et al. (2008)

#### 11 4.2 \* \* . .

| Area II: Legal st                   | ructure and security of property rights  |
|-------------------------------------|--|
| A. Judicial inde<br>disputes.       | pendence: the judiciary is independent and not subject to interference by the government or parties in   |
|                                     | rts: a trusted legal framework exists for private businesses to challenge the legality of government actions   |
|                                     | intellectual property.   |
|                                     | rference in rule of law and the political process.   |
|                                     |  |
| E. Integrity of t                   |  |
|                                     | ment of contracts  |
|                                     | estrictions on the sale of real property   |
| H. Reliability o                    | *  |
| I. Business costs                   |  |
|                                     | m to exchange with foreigners  |
| A. Taxes on inte                    |  |
|                                     | taxes on international trade as a percentage of exports plus imports.  |
| ii. Mean tariff r                   |  |
|                                     | viation of tariff rates.   |
| B. Regulatory to                    |  |
| -                                   | rt barriers: No barriers other than published tariffs and quotas.  |
|                                     | orting: the combined effect of import tariffs, license fees, bank fees, and the time required for  |
|                                     | ed-tape raises costs of importing equipment by (10=10% or less; 0=more than 50%).  |
|                                     | etween official exchange rate and black market rate.   |
|                                     | he movement of capital and people  |
| i. Access of citiz                  | ens to foreign capital markets and foreign access to domestic capital markets.   |
| ii. Restrictions of<br>among 13 IMF | in the freedom of citizens to engage in capital market exchange with foreigners—index of capital control:<br>categories.   |
| iii. Freedom of                     | oreigners to visit   |
| Area V: Regulat                     | ion of credit, labor, and business   |
| A. Credit Mark                      | et Regulations   |
| i. Ownership of                     | banks: percentage of deposits held in privately owned banks.   |
| ii. Extension of                    | credit: percentage of credit extended to the private sector.   |
| iii. Interest rate                  | controls: interest rate controls on bank deposits and/or loans are freely determined by the market.  |
| B. Labor Marke                      | t Regulations  |
| i. Hiring regula                    | ions and minimum wage: the difficulty of hiring index measures (a) whether fixed-term contracts are  |
| prohibited for p                    | ermanent tasks; (b) the maximum cumulative duration of fixed-term contracts; and (c) the ratio of the  |
| minimum wage                        | for a trainee or first-time employee to the average value added per worker.  |
| ii. Hiring and fi                   | ring practices: hiring and firing practices of companies are determined by private contract.   |
| iii. Share of labo                  | r force whose wages are set by centralized collective bargaining.  |
| iv. Hours regula                    | tions. Countries with less rigid work rules receive better scores in this component.   |
| v. Mandated co                      | st of worker dismissal.  |
| vi. Use of consc                    | ripts to obtain military personnel.  |
| C. Business Reg                     | ulations   |
| i. Administrativ                    | e requirements: Complying with administrative requirements (permits, regulations, reporting) issued by   |
| the government                      | in your country is (1 = burdensome, 7 = not burdensome).   |
|                                     | costs: Standards on product/service quality, energy and other regulations (outside environmental<br>our country are: (1 = Lax or non-existent, 7 = world's most stringent).                      |
| iii. Starting a ne                  | w business: starting a new business is generally easy.   |
|                                     | nts/bribes/favoritism: irregular, additional payments connected with import and export permits, business<br>ze controls, tax assessments, police protection, or loan applications are very rare. |
| т                                   |  |

v. Licensing restrictions: the time in days and monetary costs required to obtain a license ... ... nstruct a standard warehouse.

vi. Cost of tax compliance: the time required per year for a business to prepare, file, and pay taxes on corporate income, value added or sales taxes, and taxes on labor.

Source: Gwartney et al. (2012)

| Variable                  | Description  |
|---------------------------|--|
| Investment Profile        | Factors affecting the risk to investment not covered by other political,<br>economic and financial risk components Sub-components: contract<br>viability/expropriation, profits repatriation, payment delays.                    |
| Corruption                | Corruption within the political system<br>Financial corruption and corruption in the form of excessive<br>patronage, nepotism, job reservation, favors for favors and<br>suspiciously close ties between politics and business.  |
| Law and Order             | Law: The strength and impartiality of the legal system.<br>Order: popular observance of the law (people following the law).  |
| Democratic Accountability | How responsive government is to its people on the basis that the less<br>responsive it is, the more likely is it that the government will fall<br>(peacefully or violently).<br>Ranging from Alternating democracies to Autarchy |
| Bureaucracy Quality       | The strength and expertise to govern without drastic changes in policy<br>or interruptions in government services.<br>Ability to absorb shocks to minimize revision of policy when<br>governments change.                        |

Table A4: Variables comprising the Government Quality index

Source: ICRG (2012)

## Table A5: Percentage of high, middle and low-income countries loading significantly on the first factor for the economic institutions

|                         | Economic Freedom Index<br>Factor 1 |                     | Trade Restr         | ictions Index       |
|-------------------------|------------------------------------|---------------------|---------------------|---------------------|
|                         |                                    |                     | Factor 1            |                     |
|                         | Positive<br>loading                | Negative<br>loading | Positive<br>loading | Negative<br>loading |
| 15% cutoff              |                                    |                     |                     |                     |
| High-Income Countries   | 84%                                | 0%                  | 84%                 | 0%                  |
| Middle-Income Countries | 66.6%                              | 0%                  | 58.3%               | 8.33%               |
| Low-Income Countries    | 70.4%                              | 6.81%               | 68.1%               | 6.81%               |
| 25% cutoff              |                                    |                     |                     |                     |
| High-Income Countries   | 72%                                | 0%                  | 76%                 | 0%                  |
| Middle-Income Countries | 58.3%                              | 0%                  | 58.3%               | 0%                  |
| Low-Income Countries    | 61.3%                              | 6.81%               | 54.5%               | 6.81%               |

Note: Within group percentages are reported; Income classification based on the year 1990

|                         | Factor 1            |                     | Factor 2            |                     |
|-------------------------|---------------------|---------------------|---------------------|---------------------|
|                         | Positive<br>loading | Negative<br>loading | Positive<br>loading | Negative<br>loading |
| 15% cutoff              |                     |                     |                     |                     |
| High-Income Countries   | 8%                  | 32%                 | 60%                 | 0%                  |
| Middle-Income Countries | 33.3%               | 8.33%               | 41.6%               | 0%                  |
| Low-Income Countries    | 61.3%               | 6.81%               | 13.6%               | 15.9%               |
| 25% cutoff              |                     |                     |                     |                     |
| High-Income Countries   | 8%                  | 28%                 | 60%                 | 0%                  |
| Middle-Income Countries | 33.3%               | 16.6%               | 41.6%               | 8.33%               |
| Low-Income Countries    | 45.4%               | 4.54%               | 11.3%               | 11.3%               |

Table A6: Percentage of high, middle and low-income countries loading significantly on the first two factors for the Government Quality index

Note: Within group percentages are reported; Income classification based on the year 1990.

# Chapter 4 Firm Ownership and Provincial CO<sub>2</sub> emissions in China

With Fredrik N.G. Andersson (Lund University) and Sonja Opper (Lund University)

## Introduction

Within only three decades, China has emerged from one of the world's poorest agricultural economies to a major manufacturing economy taking up the largest share of global carbon dioxide (CO<sub>2</sub>) emissions. Projections of future emissions point toward a continuous upward sloping trajectory until 2035-2040 (Andersson & Karpestam, 2013; Yuan, Xu, Zhang, Hu, & Xu, 2014). Clearly, China's capitalist transformation and high-growth strategy poses a threat to any global attempts to slow down global warming. Yet, China's turn toward a market economy and distinct growth strategy are two factors that need to be disentangled. There is a wide consensus that China's leadership prioritized economic growth over environmental concerns throughout most of the reform period, causing severe air, water, and land pollution (He, Lu, Mol, & Beckers, 2012). The 11th five-year plan (2006-2011) was in fact the first to place greater emphasis on environmentally sustainable development (He et al., 2012). Distinct from China's growth strategy, however, the country also underwent a gradual capitalist transformation from a fully state-owned economy to a hybrid economy largely relying on private production and mixed ownership forms (Nee & Opper, 2012). This raises the question of whether, and to what extent, the country's opening up to capitalist forms of production either reinforced or possibly mitigated accumulating environmental costs.

A focus on firm ownership as a potential factor explaining pollution levels is well aligned with the "structural policy approach" to environmental policies, which aims at improving the economy's overall economic efficiency both in terms of total factor productivity (Penn World Tables)<sup>21</sup> and domestic material consumption (Global Material Flow database).<sup>22</sup> Given broad evidence confirming cross-ownership differences in terms of economic efficiency, specifically indicating higher energy consumption of state-owned enterprises (see e.g., Anderson, 1995; Talukdar & Meisner, 2001; Wang & Jin, 2007), ownership (and ownership reforms) should be a natural component of structural policy considerations. Yet, the effect of firm ownership on emissions has attracted relatively little political as well as scholarly attention. Initial firm-level analyses have produced an inconclusive account, with some studies confirming that stateowned enterprises pollute more than private enterprises (Fisher-Vanden, Jefferson, Liu, & Tao, 2004; Jiang, Lin, & Lin, 2014; Wang & Wheeler, 2003), while others diagnose no significant difference between each ownership type (Wang & Jin, 2007; Wang & Wheeler, 2005).

Building on these micro-level accounts studying random samples of firms, this paper shifts attention to the question regarding whether regional differences in ownership structure have a large enough impact to affect regional pollution levels. Our macroeconomic approach focuses on provincial variation to test whether regional differences in ownership structure have a measurable effect on overall provincial CO<sub>2</sub> emissions. Moving to a macroeconomic approach that exploits provincial differences does not only help to separate general growth effects from the role of China's capitalist transformation, but also suits China's decentralized quasi-federalist structure that allocates authority to provincial government in terms of policy implementation and enforcement (Montinola, Qian, & Weingast, 1995), which also includes shared responsibilities between the national government and local authorities in terms of environmental policies (Wang & Wheeler, 2003; Mol & Carter, 2006). The emerging inter-provincial differences in terms of production structure and pollution levels hold great analytical promise. Empirical results not only help to model future trajectories of emissions, but, from a policy perspective, are essential for estimating the overall effect of different policy reforms prioritizing or limiting distinct ownership forms.

We model the effect of firm ownership using provincial data covering the period from 1992 to 2012. Our approach is to decompose the short-run and long-run cause of emissions into scale, energy intensity, and carbon intensity components.

<sup>&</sup>lt;sup>21</sup> https://pwt.sas.upenn.edu/.

<sup>&</sup>lt;sup>22</sup> http://www.materialflows.net/home/.

We find that capital growth is the main driver of emissions growth and that private firm capital is more energy-efficient than capital employed in non-private firms. Over the long-term, emissions would have grown by between 3 and 4 percentage points more per year had private firms been as inefficient as nonprivate firms. This result underscores that continuing market-oriented reforms that increase general firm efficiency will dampen future CO<sub>2</sub> emissions growth. The greater competitive pressure on private firms compared with non-private firms is likely to be a main driver of the negative long-run correlation between private enterprises and emissions. In addition, we find that continued structural changes from an agricultural economy to a modern industrialized economy will stifle future emissions growth. Coupled with an active environmental policy, that is, regulation of carbon price, our results show that overall emissions growth could be reduced to close to zero despite the economy continuing to grow in excess of 5% per year. These insights offer valuable lessons for many developing countries, which continue to rely on large state-owned sectors. Economic development and environmental concerns can be coupled into a win-win situation through reforms that enhance productivity and competition, and indirectly, less environmental degradation.

The remainder of the paper is organized as follows: Section 2 discusses causal channels linking firm ownership and  $CO_2$  emissions, both from a general theoretical perspective and a country-specific perspective, taking context-specific institutional factors into account. Section 3 presents the data and empirical analysis and Section 4 concludes the paper.

## Firm Ownership and the Carbon Footprint

## Firm Ownership and the Carbon Footprint: A Theoretical Background

Behavioral differences in the management of private and state-owned enterprises have been subject to a long-standing debate on the relative advantages of both ownership forms (Shleifer, 1998; Shleifer & Vishny, 1994). While the comparative study of performance effects has played an important role both in the study of established market economies (Boardman & Vining, 1989; D'souza & Megginson, 1999; Dewenter & Malatesta, 2001; Megginson, & Netter, 2001) and transition economies (Djankov & Murrell, 2002; Estrin, Hanousek, Kočenda, & Svejnar, 2009; Frydman, Gray, Hessel, & Rapaczynski, 1999), second-order effects on environmental outcomes have only gained attention recently. We highlight two causal channels through which ownership can affect pollution levels. First, private and state-owned firms have different objective functions *(functional perspective)*. Second, owners and managers of private and state-owned firms face a different institutional and resource environment, which invites distinct behavioral responses *(institutional or resource-based perspective)*.

The functional perspective typically interprets private enterprises as following a single objective—profit maximization—whereas state-owned enterprises are assumed to pursue multiple objectives reflecting policy-makers' complex priorities in terms of social, economic, political, or even environmental preferences. The environmental net effect of both objective functions is ambiguous. Pure profit maximization of private firms can rely on waste avoidance or minimization, energy efficient technologies, and highly effective resource use. Profit maximization also provides incentive to the private firms to engage in technology transfer from foreign invested enterprises (Taube, 2003) leading to efficient technology. All of these strategies would also reduce pollution levels as a second-order effect (Schmid & Robin, 1995; Shirley, Kikeri, & Nellis, 1992). However, the opposite effect could occur if profit maximizers pursue their objective function by not internalizing production externalities or by undercutting technology and environmental standards (Eiser, Reicher, & Podpadec, 1996). The overall environmental impact naturally depends on the relative strength of both effects.

For state-owned enterprises, the net environmental effects are equally difficult to predict. On the one hand, the multifaceted objectives of state-owned firms can include the task to advance environmental protection in order to help government bodies comply with aggregate environmental targets (Liu & Wang, 2011; Wang & Jin, 2007; Grout & Stevens, 2003). Yet, weaker profit orientation may also contribute to less efficient resource use and higher pollution levels (Andrews & Dowling, 1998; Boycko, Shleifer, & Vishny, 1996; Djankov & Murrell, 2002).

Differences in institutional and resource environments add to the complexity of the ownership-environment link. Politically well-connected state-owned firms often enjoy 'soft budget constraints' (Kornai, 1980) as a result of the generous provision of tax benefits, subsidies, and government guaranteed low-interest loans (Adhikari, Derashid, & Zhang, 2006; Bunkanwanicha & Wiwattanakantang, 2009; Ding, Jia, Wu, & Zhang, 2014; Faccio, 2006). Whether these softer financial constraints and the resulting lower competitive pressure translate into strengthened ecological responsibility or into inefficient resource use awaits empirical testing. Similarly, managers of state-owned enterprises may either employ their closer political ties to lobby for lighter environmental regulations and to avoid environmental penalties in the case of non-compliance (Wang & Jin, 2007), or they may even experience closer scrutiny by monitoring agencies that 98 aim at upholding fair standards. The exact outcome will depend on the insulation of government bodies and the importance of political capital in the regulatory economy (Evans, 1995; Nee, Opper, & Wong, 2007).

## Firm Ownership and the Carbon Footprint: The Case of China

Institutional incentives and resource constraints are necessarily context specific. In China's reform economy, functional differences between state-owned enterprises and private firms may have been even more pronounced than generally assumed. The constitutional equality of state-owned production and private firms was not granted before 2004, reflecting the government's continuing priority treatment of public ownership throughout most of the reform period. Stateowned enterprises continue to dominate key sectors regarded as essential for overall economic development (Fan, Kanbur, Wei, & Zhang, 2014). Broad empirical evidence confirms that state-owned firms are expected to serve multiple goals beyond profit-making. These include a mix of employment, income, structural change, global integration and local growth targets, which are activated depending on the overall economic situation (Bergsager & Korppoo, 2013; Chang & Wong, 2009; Der Heide & Taube, 2013; Hu, Opper, & Sonia, 2006). The influence of multiple objectives pursued by state-owned firms-often exercised through direct political interference (Chang & Wong, 2004; Nee et al., 2007; Taube, 2005)—is reflected by lower investment efficiency and profitability compared with non-public firms (Chen, Sun, Tang, & Wu, 2011; Lardy, 2014; Su & He, 2012). There is, however, no indication that the pursuance of environmental goals played a part in this. Until the beginning of the 11th fiveyear plan in 2006, the government did not emphasize environmental goals; neither did performance assessments of political leaders incorporate 'soft' goals such as environmental standards. Instead, performance assessments relied on the provision of employment and aggregate economic growth (Landry, 2003; Bo, 2002; Li, 1998).

By contrast, the emergence of a private firm economy was not part of governmentsponsored policy reforms, but rather reflected the bottom-up rise of a new entrepreneurial class that sought profit opportunities in industrial niches insufficiently served by state-owned enterprises (Nee & Opper, 2012). From the beginning of China's economic reforms, private entrepreneurs operated outside the state allocation system and relied on rapidly expanding free market exchange, which left entrepreneurs with no alternative but to focus on profit-making to secure the survival of their companies. With the majority of companies focusing on industries with low market entry barriers, competitive pressure tends to be intense and forces companies to continuously upgrade production processes and minimize resource input. A longitudinal study of private firms in China's Yangzi Delta region reports a stable rate of process innovation between 60% and 64% for the period between 2002 and 2011 (Nee & Opper, 2015). This observation is in line with national firm survey data collected by the World Bank in 2012, which suggests that close to 64% of the interviewed private firms introduced new technology and equipment over the preceding three years. By contrast, only 35% of the companies involving state ownership invested in new technology and equipment.

Functional differences between both ownership forms are further reinforced through institutional differences, which increase competitive pressure on private companies. With factor market liberalization in China generally lagging behind product market liberalization (NERI), private companies are hard-pressed to secure investment capital, land, and skilled labor. The state-dominated banking system channels the majority of capital into state-owned firms and government sponsored projects (Andersson, Burzynska, & Opper, 2014), which leaves private companies with few alternatives but to rely on retained earnings and, to some extent, informal loans from friends and business partners (Nee & Opper, 2012; Tsai, 2002). Productivity maximization thus became a natural response to compensate for the lack of capital and to generate profits from which to further grow the company (Lardy, 2014).

Differences in social and political capital add to the disadvantaged position of private firms. State-owned enterprises typically maintain closer ties with the government and local bureaucracy. In particular, managers of large state-owned enterprises can wield a large influence on local policy makers, possibly helping them to shape local regulation in their interest (Wang and Jin, 2007). Beyond these forms of state-capture commonly found in transition economies (Hellman, Jones, & Kaufmann, 2003), political ties also shield companies from pollution payments and punishments in the case of non-compliance with regulatory standards (Wang & Jin, 2007; Wang, Mamingi, Laplante, & Dasgupta, 2003; Wang & Wheeler, 2003). While private companies also invest strategically in government ties (Ma & Parish, 2006; Nee & Opper, 2010; Taube, 2013; Xin & Pearce, 1996), there is no indication that they enjoy advantages comparable with their publicly owned counterparts. The same World Bank survey conducted in 2012 indicates that private companies are much more likely to be subject to routine inspections than publicly owned companies. While 68% of the interviewed private company managers reported company inspections over the preceding years, the corresponding share of publicly owned and partially stateowned companies was less than 34% (World Bank, 2012).

To summarize, we assert that functional as well as institutional differences between state-owned companies and private firms in China suggest that private firms are likely to leave a smaller carbon footprint than their publicly owned competitors.

## Measuring the Size of the Private Sector

The ownership structure of China's economy is complex and includes, alongside the pure forms of private enterprises and state-owned enterprises, a range of hybrid ownership forms, such as rural and urban collectively owned enterprises, cooperative enterprises, joint-ownership enterprises, limited liability enterprises, and share-holding enterprises with mixed ownership forms (Bruton, Peng, Ahlstrom, Stan, & Xu, 2015). Finally, the portfolio of distinct ownership types also includes wholly foreign-owned enterprises as well as small-scale units of selfemployment. For simplicity, we include in the category of private firms only those that are registered as private companies. These companies have at least eight employees and were either established by a natural person or a majority of the company is owned by a natural person (Lardy, 2014). We exclude from our measure companies registered as limited liability enterprises and shareholding enterprises because both registration categories combine purely private firms with organizational hybrids, which combines private and state-owned shares to varying degrees. A large share of these companies are partially privatized, formerly fully state-owned companies still containing some-and in the case of stock listed companies—decisive state-ownership shares (Lee, 2009).

Without detailed information on each individual company, it is impossible to determine the exact ownership of these companies. In our study, we therefore compare the environmental effect of private enterprises against all other ownership types, although we acknowledge that some of these may include a sizable private component. Our estimation results are therefore likely to underestimate the effect of private ownership. Any support for our hypothesis therefore provides strong evidence that private enterprises pollute less that state-owned enterprises.

We use the share of private employment in total employment as a proxy for the relative size of the private sector. Use of total employment numbers rather than number of firms helps to control for regional differences in the firm scale. An alternative would be to rely on output statistics, which are, however, not available at the provincial level before the late 1990s, thus this would reduce our sample to a relatively short observation period. For the available years, however, high

correlation coefficients (0.8) between employment share and output share confirm the employment share as a good indicator of regional private firm activities.

## **Empirical Analysis**

## **Econometric Model**

We model the provincial growth rate in carbon emissions.<sup>23</sup> The econometric model is derived from the emissions decomposition model in which economic activity affects emissions through three separate channels: scale, energy intensity, and carbon intensity (Andersson & Karpestam, 2013; Lise, 2006; Tol, 2007; Tol, Pacala, & Socolow, 2009). *Scale* is the level of energy demand caused by economic activity. *Energy intensity* is defined as the average amount of energy required for each unit of economic activity and thus reflects energy efficiency. *Carbon intensity* is the average amount of CO<sub>2</sub> emitted when producing one unit of energy. Analytically, however, carbon intensity is of minor interest because China's overall energy mix has remained relatively stable since the early 1990s, with only modest increases in non-fossil fuels, such as renewable energies and nuclear power (China Energy Databook, 2013). Our focus is therefore on modeling changes in scale and energy intensity.

To capture changes in scale (i.e., the level of economic activity) we use capital growth and employment growth.<sup>24</sup> We expect capital growth to be the main driver of energy demand. Energy intensity is modeled using our main variable of interest-private firms-and a vector of control variables. To isolate the impact of the private sector on energy intensity, our regression model includes the growth rate in the log of the share of private employment,  $\Delta ln$  (private employment<sub>it</sub>/total employment<sub>it</sub>). This specification provides two important properties: First, using the log of the share in private sector

<sup>&</sup>lt;sup>23</sup> We calculate the growth rate in emissions as the log growth rate,  $\Delta ln((CO_2)_{jt})$ , where *j* denotes province and *t* denotes time.

<sup>&</sup>lt;sup>24</sup> Scale is sometimes modeled using GDP growth, but a reliance on GDP growth would produce ambiguous results because changes in GDP growth are caused by both extensive factors (affecting scale) and intensive factors (affecting the energy intensity). We therefore decompose GDP growth into its three main sources: capital growth, employment growth, and total factor productivity (TFP) and use the first two sources to model scale and the latter source to model energy intensity. A description of how we decompose GDP growth is available in Appendix A.

employment, we take into account that private sector employment growth has outpaced growth in the rest of the economy.<sup>25</sup> Second, using the change in the share rather than the change in the log share, which we use, could cause spurious results (Gerdes, 2010).<sup>26</sup>

Regarding our vector of control variables, first, we include total factor productivity growth to capture productivity improvements in the economy. Higher productivity is expected to reduce energy intensity because the same level of production is achieved as before but with less capital and employment. Growth in the share of manufacturing and service production is included to account for differences in energy intensity across different industries, with the manufacturing industries assumed to be more energy intensive than the service and agricultural industries (Andersson & Karpestam, 2013).

Additionally, growth in private transports (person-km), growth in freight transports (ton-km), log-change in the share of urban population, and real energy prices (coal and oil prices) are included as control variables. We expect a larger share of the population living in urban areas to increase the energy intensity (Jones, 1991). The effect of transport growth is ambiguous. On the one hand, increasing geographical specialization of modern production increases transport volumes of goods and raw material (Rodrigue, 2006), which is expected to increase the energy intensity. On the other hand, greater specialization of production typically co-evolves with efficiency increases of the economy and lower energy intensity of available transport facilities (Rodrigue, Comtois, & Slack, 2013). The expected net effect of freight growth is therefore uncertain. Finally, real coal and oil prices<sup>27</sup> are expected to have a disciplining effect on energy use by creating incentives to reduce energy waste and to invest in energy-saving new technologies.

All explanatory variables refer to the same time period as the carbon emissions. The only exception is energy prices, which enter the estimation model with a fiveyear time lag. In this case, we assume that actors need a certain adjustment period to respond to price changes because both behavioral changes in the production process as well as the introduction of new energy-efficient technologies (and new skills) require a certain time. We have also experimented with shorter lag lengths

<sup>&</sup>lt;sup>25</sup> The share of private sector employment in wholly privately owned firms has increased from less than 1% in 1992 to 21% in 2009 (China Statistical Yearbook, 2011).

<sup>&</sup>lt;sup>26</sup> We have also estimated the models using the change in the absolute share, but our statistical tests clearly indicate that using the log of the share improves the models' statistical properties.

<sup>&</sup>lt;sup>27</sup> Real energy prices are calculated using the GDP deflator.

of less than five years, with the only effect being that coefficient estimates for real energy prices become insignificant. Estimation results for the other parameters, including those for private ownership, remain unaffected.

Most of the variables included in the model are supply-side variables that affect the economic structure. Such changes take time, and we expect these to explain the long-term movement in emissions but not, to any great extent, the short-term volatility in emissions (Andersson & Karpestam, 2013). In our models, we therefore distinguish between short-term and long-term effects of the explanatory variables. The parameters in the models are estimated using the band spectrum regression estimator (see e.g., Engle, 1974; Andersson, 2011). In simple terms, the band spectrum regression estimator is a two-step estimator: First, all variables are decomposed into short-run and long-run components using a band-pass filter. Second, the regression model is estimated using the decomposed data whereby short-run and long-run parameter estimates are obtained. We use the maximal overall discrete wavelet transform (MODWT) as the band-pass filter. Although other filters can be used, we employ the MODWT because it is suitable for time series that possibly contain structural breaks, outliers, and other non-recurring events that may otherwise negatively affect decomposition of the data (Percival & Walden, 2006).<sup>28</sup> Compared to other estimators that distinguish between short and long-term effects, the band spectrum estimator has good small sample properties (Andersson, 2008, 2011).

Following previous studies (e.g., Ramsey and Lampart, 1998; Assenmacher-Wesche and Gerlach, 2008), we define the short-run as business cycle fluctuations in the economy lasting up to eight years. Accordingly, the long-run is defined as persistent changes in the economy lasting more than eight years. We have also tested other distinctions between the short-run and long-run. Those tests show that fluctuations up to eight years have different dynamics from more persistent changes in the economy. There is, however, no significant difference in the results if we define the long-run as 16 years and beyond compared with our definition of the long-run (eight years and beyond). Specifically, our regression models are given by:

$$\Delta ln((CO_2)_{jt}) = \beta_{j1} + \beta_2 \Delta X_{SRjt} + \beta_3 \Delta X_{LRjt} + \beta_4 \Delta Z_{SRjt} + \beta_5 \Delta Z_{LRjt} + \varepsilon_{jt}$$
(1)

where X is a vector containing the variables explaining changes in scale and Z are the variables explaining changes in the energy intensity. By construction, the

<sup>&</sup>lt;sup>28</sup> For more information about the MODWT, see Percival and Walden (2006), Crowley (2007) and Anderson (2008).

decomposition of the data into short-run and long-run variations is linear whereby,  $\Delta X_{jt} = \Delta X_{SRjt} + \Delta X_{LRjt}$  and  $\Delta Z_{jt} = \Delta Z_{SRjt} + \Delta Z_{LRjt}$ . We model the first difference to make the data stationary and avoid spurious regressions without affecting the ability to identify respective time horizons. In fact, in small samples, the use of first differenced data and band spectrum regression may yield more reliable parameter estimates than cointegration techniques (Andersson, 2011).

We estimate two main model specifications: In Model 1, all variables are included independently. In Model 2 we also consider interaction effects between capital growth and growth in the share of private employment, capital growth with growth in the share of manufacturing capital, and capital growth with growth in the share of service sector production. Given that we expect capital growth to be the main driver of scale, we include these interaction variables to test if any change in ownership or industrial composition has an effect on the energy intensity or if emission is only affected by these variables if they are linked to changes in the capital composition.

Provincial level data for energy prices is not available for the entire observation period, which forces us to rely on national level energy prices. This implies that we cannot include fixed time effects and price levels in the same model specification. We therefore estimate two specifications of Models 1 and 2. In the first set of models (Models 1a and 2a), we include all explanatory variables but only introduce fixed provincial effects. In the second set of models (Models 1b and 2b), we exclude energy prices, which allows us to include both fixed provincial effects and fixed time effects. In these models the effect of changes in national energy prices are captured by the fixed time effects.

## Data Sources and Descriptive Statistics

Data is collected from five sources: China Data Online<sup>29</sup>, China Energy Databook (2013)<sup>30</sup>, China Statistical Yearbook, Zhao and Burnett (2014), and Wu (2009). A detailed description of the data and the respective data sources is available in Appendix B.

Emissions of  $CO_2$  have grown on average, by 6.1%, but yearly growth is relatively volatile, with fluctuations of +/-18% p.a. Most of these fluctuations are due to short-term causes (see Table 1). The short-term standard deviation in emissions

<sup>&</sup>lt;sup>29</sup> http://chinadataonline.org/.

<sup>&</sup>lt;sup>30</sup> https://china.lbl.gov/.
is 12.1 compared with the long-run standard deviation of 4.4.<sup>31</sup> Growth in the share of private employment is 18.4% per year on average, with almost equal short-term and long-term volatility of 11.6 and 11.8, respectively. Capital stock and freight transport have, on average, grown in excess of 10% per year, while employment growth, growth in the share of manufacturing, and growth in the share of service production has been less than 1.5% per year. Both real coal prices and real oil prices increased during the sample period.

|                    | Average | Short-run Std.Dev. | Long-run Std.Dev. |
|--------------------|---------|--------------------|-------------------|
| CO <sub>2</sub>    | 6.1     | 12.1               | 4.4               |
| Labor              | 1.4     | 3.5                | 1.4               |
| Capital            | 11.2    | 0.4                | 7.9               |
| TFP                | 4.6     | 1.1                | 1.5               |
| Manufacturing      | 0.8     | 1.9                | 1.8               |
| Service            | 0.8     | 1.9                | 1.8               |
| Private Employment | 18.4    | 11.6               | 11.8              |
| Urbanization       | 0.4     | 4.3                | 2.3               |
| РКМ                | 7.4     | 5.6                | 2.7               |
| ТКМ                | 10.8    | 15.6               | 7.9               |
| Coal               | 4.2     | 2.8                | 1.7               |
| Oil                | 8.3     | 6.8                | 2.2               |

Table 1: Descriptive statistics: Average percentage change 1992–2010

Note. The percentage changes are calculated as  $x_t = ln(X_t) - ln(X_{t-1})$ .

At the provincial level,  $CO_2$  emissions per unit of capital ( $CO_2$ /capital intensity) are lowest along the east coast and higher in the northern and western provinces (see Panel A in Figure 1). The five provinces with the highest emissions emit 3.75 times as much  $CO_2$  per unit of capital as the five least polluting provinces. The intensity has, on average, been reduced by 52% during the sample period and the reduction is generally higher in the Eastern provinces compared with the Western and Northern provinces (see Panel B). The private employment share follows a similar, but inverse pattern, as the  $CO_2$ /capital intensity. The private employment share is higher along the east coast, and this part of the country has also had among the largest increases in the share of private employment (see Figure 2).

<sup>&</sup>lt;sup>31</sup> The total volatility is obtained by summing the short-run and the long-run volatility.



Panel A: Average CO<sub>2</sub>/capital intensity



Panel B: Percentage change in CO2/capital intensity

#### Figure 1.CO<sub>2</sub>/capital intensity 1992-2012

Note: A darker (lighter) shade illustrates higher (lower) CO<sub>2</sub> emissions per GDP unit in panel A and a greater (smaller) reduction in CO<sub>2</sub> emissions per GDP unit in panel B.



Panel A: Average private employment share



Panel B: Average percentage change in the employment share.

Figure 2. Private employment share 1992–2012

The link between the  $CO_2$ /capital intensity and the share of private employment is illustrated in Figure 3, which compares the relationship between the levels of intensity and employment (Panel A), as well as the change in the two variables (Panel B). Evidently, a higher share of private sector employment is associated with a lower  $CO_2$ /capital intensity. This linear relationship is even more pronounced when changes in the share of private enterprise employment are plotted against the change in the  $CO_2$ /capital intensity. Rapid growth of private sector employment is clearly correlated with a rapid decline in the  $CO_2$ /capital intensity.

One potential explanation for the negative correlation between private firms and  $CO_2$  emissions is that the private sector is more involved in light industries with lower emissions compared with the state-owned sector. This is not the case, however, as Table 2 reveals. This Table illustrates emissions from the manufacturing industries as well as the relative distribution of private and state-owned net fixed investments in each sector.<sup>32</sup> The two most polluting industries are the chemical (row 7) and metal industries (row 9), both in terms of total emissions (column 1) and emissions per unit of net fixed assets (column 3). Both private and state-owned firms have invested a relatively large capital share in these two sectors. Of the total private mining and manufacturing capital, 24% is invested in metal industries and 10% is invested in chemical industries. The corresponding investment shares for state-owned firms are 27% and 11%, respectively. There are some differences in the investment shares for the remaining industries, but none of these industries are major polluting industries and therefore not consequential for our results.

To summarize, Figure 3 and Table 2 offer some initial support for our claim that private firms are less polluting than state-owned firms and that these differences are not due to structural differences in firm activities.

<sup>&</sup>lt;sup>32</sup> Similar data is not available for the service sector. Despite this, the manufacturing sectors emit 4.5 times more than the service sector, thus any differences between private and state-owned investments in the service sector are unlikely to have any major impacts on the results.



Panel A: CO2 emissions per capital unit and private employment share



Panel B: Percentage change in CO2/capital intensity and private employment share

#### Figure 3. CO<sub>2</sub>/capital intensity and private employment share.

Note: The average *CO<sub>2</sub>/capital* intensity and the average private employment share in Panel A have been normalized to 1.

|  | CO <sub>2</sub> emissions<br>(kilotones) | Net fixed<br>assets<br>(100<br>million<br>yuan) | CO <sub>2</sub> /<br>net<br>fixed<br>assets | Distribution of<br>private net fixed<br>assets | Distribution of<br>SOE net fixed<br>assets |
|--|--|---|---|--|--|
|  | (1)                                      | (2)   | (3)   | (4)  | (5)  |
| (1) Mining and quarrying   | 195 472                                  | 19 513  | 10.0  | 6.2%   | 30.6%                                      |
| (2) Food,<br>beverages, and<br>tobacco                                       | 70 816                                   | 5 907   | 12.0  | 10.5%  | 3.6%                                       |
| (3)Textile, textile<br>products, leather,<br>and footwear                    | 53 266                                   | 4 310   | 12.4  | 10.8%  | 0.6%                                       |
| (4) Wood and<br>products of wood<br>and cork                                 | 12 007                                   | 886   | 13.6  | 2.2%   | 0.1%                                       |
| (5) Pulp, paper,<br>printing, and<br>publishing                              | 52 004                                   | 2 350   | 22.1  | 3.9%   | 1.6%                                       |
| (6) Coke, refined<br>petroleum, and<br>nuclear fuel                          | 100 868                                  | 5 181   | 19.5  | 2.3%   | 7.7%                                       |
| (7) Chemicals<br>and chemical<br>products                                    | 269 228                                  | 9 761   | 27.6  | 10.4%  | 10.6%                                      |
| (8) Rubber and plastics  | 23 432                                   | 1 883   | 12.4  | 4.2%   | 0.6%                                       |
| (9) Other non-<br>metallic mineral,<br>basic metals, and<br>fabricated metal | 1 340 739                                | 24 451  | 54.6  | 24.1%  | 27.9%                                      |
| (10) Machinery,<br>electrical, and<br>optical<br>equipment                   | 58 147                                   | 11 947  | 4.9   | 20.4%  | 7.9%                                       |
| (11) Transport<br>equipment  | 25 382                                   | 6 747   | 3.8   | 4.8%   | 8.8%                                       |
| Total  | 2 201 360                                | 93 025  | 23.7  | 100%   | 100%                                       |

Table 2:CO2 emissions and net fixed private and state-owned assets in the mining and manufacturing sector in 2009

### **Regression Results and Policy Conclusions**

Estimation results are presented in Table 3.<sup>33</sup> As expected, there is no significant difference between the results including real energy prices but no time fixed effects (Models 1a and 2a) and those results excluding real energy prices and including fixed time effects (Models 1b and 2b). According to the information criterion, Model b has slightly better statistical properties compared with Model a, but the differences in the information criterion are small.

Overall, each respective model's explanatory power, as estimated by the adjusted  $R^2$ , is relatively high for the long-run, between 0.8 and 0.9, but more modest for the short-run, approximately 0.1. The lower explanatory power for the short-run model indicates that fluctuations in short-run emissions are due to other non-economic factors, for example, an unusually cold winter, whereas economic factors play a key role in determining emission levels over the long term. These results indicate that emissions growth can fluctuate for periods of up to a five-year, and periodically depart from expectations purely based on economic growth. However, such short-term variations in emissions are to be expected (see Table 1) and can even lead to relatively large temporary deviations from the trajectory expected by economic trends.

<sup>&</sup>lt;sup>33</sup> Short-run TFP has been excluded from the model due to multicollinearity.

#### Table 3 Regression results

|   | Moc               | lel 1a             | Mod                | lel 1b             | Мос                | lel 2a             | Moc                | lel 2b             |
|---|-------------------|--------------------|--------------------|--------------------|--------------------|--------------------|--------------------|--------------------|
|   | Short-run         | Long-run           | Short-run          | Long-run           | Short-run          | Long-run           | Short-run          | Long-run           |
| Labor <sub>jt</sub>   | 0.03<br>(0.20)    | 0.34<br>(0.34)     | 0.13<br>(0.16)     | 0.18<br>(0.36)     | 0.04<br>(0.20)     | 0.96<br>(0.61)     | 0.14<br>(0.16)     | 0.78<br>(0.65)     |
| Capital <sub>jt</sub>   | 1.40*<br>(0.75)   | 0.65***<br>(0.21)  | 0.83<br>(1.23)     | 0.62***<br>(0.21)  | 0.39<br>(0.93)     | 1.09***<br>(0.35)  | 0.13<br>(1.26)     | 1.00***<br>(0.37)  |
| $\mathrm{TFP}_{jt}$   |                   | 0.48<br>(0.48)     |                    | 0.46<br>(0.45)     |                    | 0.96*<br>(0.46)    |                    | 0.84*<br>(0.43)    |
| $Manuf_{jt} \\$   | -1.4**<br>(0.64)  | -0.66<br>(0.43)    | -1.36*<br>(0.75)   | -0.67<br>(0.45)    | -0.34**<br>(0.67)  | -0.95<br>(1.28)    | -1.36*<br>(0.77)   | -1.12<br>(1.28)    |
| Service <sub>jt</sub>   | -1.4**<br>(0.056) | -1.18***<br>(0.41) | -1.63***<br>(0.56) | -1.12***<br>(0.42) | -1.45***<br>(0.51) | -0.97***<br>(0.21) | -1.69***<br>(0.50) | -0.67***<br>(0.20) |
| $PKM_{jt}$  | -0.19*<br>(0.12)  | -0.40**<br>(0.20)  | -0.07<br>(0.13)    | -0.39*<br>(0.20)   | -0.17<br>(0.11)    | -0.31*<br>(0.16)   | -0.06<br>(0.11)    | -0.33*<br>(0.17)   |
| $TKM_{jt} \\$   | -0.03<br>(0.04)   | 0.07<br>(0.07)     | -0.03<br>(0.05)    | 0.06<br>(0.07)     | -0.03<br>(0.04)    | 0.04<br>(0.05)     | -0.03<br>(0.05)    | 0.04<br>(0.05)     |
| Urbanization <sub>jt</sub>  | 0.41*<br>(0.22)   | -0.01<br>(0.25)    | 0.23<br>(0.20)     | -0.11<br>(0.26)    | 0.42*<br>(0.21)    | 0.16<br>(0.28)     | 0.22<br>(0.19)     | 0.09<br>(0.31)     |
| Private <sub>jt</sub>   | 0.04<br>(0.06)    | -0.22***<br>(0.04) | -0.02<br>(0.05)    | -0.17**<br>(0.08)  | -0.04<br>(0.06)    | 0.15<br>(0.28)     | -0.01<br>(0.05)    | 0.17<br>(0.25)     |
| Private <sub>jt</sub> ×Capital <sub>jt</sub>  |                   |                    |                    |                    | 0.18***<br>(0.07)  | -0.03***<br>(0.01) | 0.18**<br>(0.06)   | -0.03***<br>(0.01) |
| $Manuf_{jt} \!$ |                   |                    |                    |                    | 0.19<br>(0.78)     | 0.09<br>(0.13)     | 0.54<br>(0.89)     | 0.10<br>(0.13)     |
| $Service_{jt} {\times} Capital_{jt}$  |                   |                    |                    |                    | 0.28<br>(0.58)     | 0.50<br>(0.63)     | 0.37<br>(0.84)     | 0.48<br>(0.35)     |
| Coal <sub>jt-5</sub>  | 0.17<br>(0.26)    | -1.35***<br>(0.33) |                    |                    | 0.31<br>(0.26)     | -1.60***<br>(0.31) |                    |                    |
| Oil <sub>jt-5</sub>   | 0.04<br>(0.12)    | -0.58*<br>(0.32)   |                    |                    | -0.02<br>(0.12)    | -0.30<br>(0.32)    |                    |                    |
| Fixed province  | Yes               |                    | Yes                |                    | Yes                |                    | Yes                |                    |
| Fixed time  | No                |                    | Yes                |                    | No                 |                    | Yes                |                    |
| Adjusted R <sup>2</sup>   | 0.04              | 0.827              | 0.090              | 0.789              | 0.05               | 0.812              | 0.089              | 0.903              |
| BIC   | 5.767             |                    | 5.663              |                    | 5.721              |                    | 5.806              |                    |

Note: (a) Standard errors are estimated using Arrelano's (1987) robust standard errors. (b) \*\*\*, \*\*, \* denote statistically significant at the 1%, 5%, and 10% level, respectively. (c) TFP is excluded from the short-run model due to multicolinearity.

Considering Model 1, we find no significant short-run effect of ownership on emissions. But, we do find a significant and negative effect for the long-run: the estimated long-run elasticity is -0.17 for Model 1a and -0.22 for Model 1b. On average, the share of private sector employment has grown by 18.4% per year, which indicates that private sector growth has reduced the annual growth rate in emissions in the range of 3 and 4 percentage points.

The main driver of emissions growth (scale effect) in the long-run is capital accumulation, whereas employment growth remains insignificant. The long-run  $CO_2$ /capital elasticity is 0.6, which translates into an average 7% increase of emissions annually due to capital growth. Total Factor Productivity (TFP) growth has no significant effect on the energy intensity, a result that is in line with China's growth strategy emphasizing capital accumulation (Andersson, Edgerton, & Opper, 2013): capital growth is close to 11%. Notwithstanding, with the size of the capital stock/GDP ratio reaching high levels even in comparison with international standards (Penn World Tables, 2015) capital growth is likely to slow down in the future. A lower capital growth rate would also imply a lower rate of emissions growth.

For the remaining control variables, we confirm that a larger service sector and more private transport are associated with lower emissions in the long-run. The combined effect, however, is insufficient to stop further emissions growth. Real energy prices also contribute to slower emissions growth. Especially the coal price, with an estimated elasticity of -1.35, incentivizes producers to reduce energy consumption. However, producers do not respond to short-term fluctuations but accommodate them only as a response to stable price increases over longer periods of time.

Model 2 traces the causal channel through which ownership affects emission outcomes. For the short-run, we find a positive relationship between emissions and the interaction variable between capital growth and growth in private sector employment. For the long-run we find a significant negative effect. From these results we can draw two main conclusions. First, the general ownership effect identified in Model 1 disappears. Instead, private ownership affects emissions growth through ownership-specific differences in capital stock investments. Second, these results also show that the private sector is more receptive to changes in the market than non-private firms. In the short-run, private firm production responds to temporary changes in demand in line with changing market conditions, which in turn causes a positive correlation with emissions. In the longrun, competitive pressure forces private firms to make efficient use of their capital and invest more in energy-efficient technologies than non-private firms, which in turn causes a negative long-run correlation with emissions. To summarize, competitive pressure forces private firms to limit their carbon footprint.

In Model 2, the estimated CO<sub>2</sub>/capital elasticity increases from approximately 0.65 to 1.09 compared with Model 1. Given that we control for private sector capital using the interaction variable, the estimated elasticity applies to nonprivate enterprises. A one-unit increase in non-private owned capital increases emissions by 1% ceteris paribus. To illustrate the importance of the private sector we calculate the average CO<sub>2</sub>/capital elasticity for each province based on the regional share of private and non-private firms. Figure 4 illustrates our results, with an average elasticity of 0.71 and provincial averages ranging from a minimum of 0.48 (Jiangsu and Zhejiang) to a maximum of 0.94 (Hainan). Along the east coast, the center of private firm production, the estimates are usually below 0.65, while corresponding elasticities of inland provinces typically exceed 0.75. While these differences in the elasticity may appear small, the cumulative effects over time become substantial due to the country's rapidly expanding capital stock. As a thought experiment, we compare actual CO2 emissions growth with a hypothetical growth rate that would have been accomplished based on stateowned production only. For the average province, this suggests annual savings in CO<sub>2</sub> emissions growth of 3.4 percentage points. Over a period of 20 years, realized emissions are approximately 35% to 40% lower than they would have been in an economy absent private firm production.



Figure 4. Estimated CO<sub>2</sub>/capital elasticity for the respective provinces.

Our results offer the basis for some policy implications. Most importantly, our findings suggest that China's continuing economic growth does not necessarily imply an ever-growing carbon footprint. On the contrary, reforms that emphasize increasing marketization and privatization are likely to also limit emissions growth. This may involve further hardening of budget constraints of state-owned enterprises to increase market orientation and efficient resource use, but may also imply a continuing reduction of state-owned shares in public firms. In parallel, bureaucratic reforms need to ensure equal treatment of firms independent of ownership. Indication of greater leeway of law enforcement and scarce inspection efforts of state-owned firms compared with private firms are likely to have a decisive impact on the management's approach to energy efficiency. Our results show that emissions growth would be substantially smaller if the efficiency gap was closed between private and non-private firms. For example, if the CO<sub>2</sub>/capital elasticity throughout China was as low as in Zhejiang province, the country's leader in terms of private firm production, the average annual growth rate in emissions due to capital accumulation would be 2.5 percentage points lower every year.

Although continued economic reforms facilitating closure of the efficiency gap between private and non-private enterprises would have a substantial effect on the future trajectory of emissions, it is not enough to stop emissions from growing. Our results, however, also point toward additional policies that will reduce emissions. For example, producers respond fairly sensitively to energy prices. Higher carbon prices, for instance, would lead to greater energy efficiency throughout the Chinese economy. A long-term increase in the real coal price of 1% would decrease energy intensity by 1.35%. Given the currently low level of partly subsidized—energy prices (Lin & Jiang, 2011), market liberalization of the energy market would substantially reduce the country's environmental costs associated with  $CO_2$  emissions.

In addition, China's continuing structural change is likely to contain future emissions growth. Two transmission channels are important: First, given the relatively lower energy intensity, the expansion of a modern service economy will also bring about a slowdown in emissions growth. Second, continued restructuring of the economy and technological updating of production clusters, in our models captured by private transport, will also reduce the future growth rate in  $CO_2$  emissions by reducing the energy intensity.

A back-of-the envelope calculation highlights our point. Assume capital continues to grow by 7.5% per year (i.e., by the same rate as the official GDP growth target), and assume further that the efficiency gap between private and non-private production is reduced such that all provinces reach the same  $CO_2$ /capital intensity 116

as Zhejiang province, then annual emissions growth will reach 3.6% due to the scale effect. Further assume that policymakers introduce a real long-term increase of the coal price by 2% above inflation, then emissions growth would be reduced to 0.9% per year. Finally, with continued structural transformation, the growth rate could be further reduced and potentially even turn negative.

It is important to note that these results are based on our long-run estimations because short-term variations in emissions are only, to a small degree, caused by economic activity. Policies attempting to reduce emissions should therefore consistently focus on the long-run and abstract from short-term volatility in emissions. Moreover, to avoid disrupting the economic process, any system aimed at reducing emission, such as an Emissions Trading System (ETS) currently planned to be introduced in China (Lo, 2012), must be designed to accommodate for these short-term fluctuations (with a standard deviation of 12.1) to avoid both negative economic consequences (see e.g., Andersson & Karpestam, 2011, 2013) and a disturbance of the price signal.

### Conclusion

China has emerged as the world's top polluter of greenhouse gases. While China's political leadership is well aware of the costs associated with accumulating environmental damage and already facing societal discontent (Knight, 2014), there is still great skepticism regarding how the commitment and enforcement of tighter environmental laws would affect the country's growth trajectory. Our study shows that China's ongoing reforms and continuing capitalist transformation need not be at odds with environmental goals. To the contrary, there is substantial unexploited opportunity to limit  $CO_2$  emissions without sacrificing ambitious growth targets, but it will require a change of policies.

Our focus was on the impact of ongoing structural ownership changes of the economy and associated regional pollution levels. Our results show that private firms are more efficient than non-private firms, which translates into measurable effects at the provincial level. The inherently greater market and profit orientation of private firms forces managers to look for efficient production technologies and to operate the available capital stock with minimal resource waste. This is clear evidence of the beneficial second-order effects of free markets, which limit wasteful production through competitive pressure. By contrast, deviation from the market mechanism—either through subsidized state-ownership or through controlled energy prices—reduces competitive pressure and leads to suboptimal production in terms of energy efficiency.

These findings have broader implications because they suggest that only the continuing modernization and transformation of China's economy into a full-fledged market economy will be able to decrease pollution levels to the international standards of industrialized economies. Modernization, not a slowdown in ongoing reforms, needs to be the future path for development of the Chinese economy. Clearly, it is a myth to assume that the environment cannot digest a growing China. Quite to the contrary, the environment cannot digest a stagnating China because this would imply freezing current inefficiencies that stem from unfinished ownership and market reforms.

## References

- Adhikari, A., Derashid, C., & Zhang, H. (2006). Public policy, political connections, and effective tax rates: Longitudinal evidence from Malaysia. *Journal of Accounting and Public Policy*, 25(5), 574-595.
- Anderson, K. (1995). Mining, privatization and the environment. *Minerals and Energy*, *11*(3), 25-28.
- Andersson, F. N. G.(2008). *Wavelet analysis of economic time series,* (diss.). Lund Economic Studies, 149.
- Andersson, F. N. G. (2010). Policy Update: Business cycles and the European Union Emission Trading System. *Carbon Management*, 1(2), 199-201.
- Andersson, F. N. G. (2011). Band spectrum regression using wavelet analysis. Lund University Department of Economics Working Paper, 2011: 22.
- Andersson, F. N. G., Burzynska, K., & Opper, S. (2014). Lending for Growth? A Granger Causality Analysis of China's Finance-Growth Nexus.
- Andersson, F. N. G., Edgerton, D. L., & Opper, S. (2013). A matter of time: Revisiting growth convergence in China. World Development, 45, 239-251.
- Andersson, F. N. G., & Karpestam, P. (2011). A flexible CO<sub>2</sub> targeting regime. *Economics Bulletin*, *31*(1), 297-308.
- Andersson, F. N. G., & Karpestam, P. (2013). CO 2 emissions and economic activity: Short-and long-run economic determinants of scale, energy intensity and carbon intensity. *Energy Policy*, *61*, 1285-1294.
- Andrews, W. A., & Dowling, M. J. (1998). Explaining Performance Changes in Newly Privatized Firms. *Journal of Management Studies*, 35(5), 601-617.
- Arellano, M. (1987). Computing Robust Standard Errors for Within-Group Estimators. Oxford Bulletin of Economics and Statistics, 49, 431-434.
- Assenmacher-Wesche, K., & Gerlach, S. (2008). Money growth, output gaps and inflation at low and high frequency: Spectral estimates for Switzerland. *Journal of Economic Dynamics and Control, 32*(2), 411-435.
- Bergsager, H., & Korppoo, A. (2013). China's State-Owned Enterprises as Climate Policy Actors: The Power and Steel Sectors. *Nordic Council of Ministers, Copenhagen*.
- Bo, Z. (2002). *Chinese provincial leaders: Economic performance and political mobility since 1949*: East Gate Book.
- Boardman, A. E., & Vining, A. R. (1989). Ownership and performance in competitive environments: A comparison of the performance of private, mixed, and stateowned enterprises. *Journal of law and Economics*, 1-33.

- Boycko, M., Shleifer, A., & Vishny, R. W. (1996). A theory of privatisation. *The Economic Journal*, 309-319.
- Bruton, G. D., Peng, M. W., Ahlstrom, D., Stan, C., & Xu, K. (2015). State-owned enterprises around the world as hybrid organizations. *The Academy of Management Perspectives, 29*(1), 92-114.
- Bunkanwanicha, P., & Wiwattanakantang, Y. (2009). Big business owners in politics. *Review of Financial Studies, 22*(6), 2133-2168.
- Chang, E. C., & Wong, S. M. (2004). Political control and performance in China's listed firms. *Journal of Comparative Economics*, 32(4), 617-636.
- Chang, E. C., & Wong, S. M. (2009). Governance with multiple objectives: Evidence from top executive turnover in China. *Journal of Corporate Finance*, *15*(2), 230-244.
- Chen, S., Sun, Z., Tang, S., & Wu, D. (2011). Government intervention and investment efficiency: Evidence from China. *Journal of Corporate Finance*, *17*(2), 259-271.
- China Statistical Yearbook, various years.
- Crowley, P. M. (2007). A guide to wavelets for economists. *Journal of Economic Surveys*, 21(2), 207-267.
- D'souza, J., & Megginson, W. L. (1999). The financial and operating performance of privatized firms during the 1990s. *The Journal of Finance, 54*(4), 1397-1438.
- Der Heide, P. I., & Taube, M. (2013). China's Iron and Steel Industry at the Global Markets Interface: Structural Developments and Industrial Policy Interventions. *The Copenhagen Journal of Asian Studies*, 29(2), 110-142.
- Dewenter, K. L., & Malatesta, P. H. (2001). State-owned and privately owned firms: An empirical analysis of profitability, leverage, and labor intensity. *American Economic Review*, 320-334.
- Ding, S., Jia, C., Wu, Z., & Zhang, X. (2014). Executive political connections and firm performance: Comparative evidence from privately-controlled and state-owned enterprises. *International Review of Financial Analysis*, 36, 153-167.
- Djankov, S., & Murrell, P. (2002). Enterprise restructuring in transition: A quantitative survey. *Journal of Economic Literature*, 739-792.
- Eiser, J. R., Reicher, S. D., & Podpadec, T. J. (1996). Attitudes to privatization of UK public utilities: Anticipating industrial practice and environmental effects. *Journal of Consumer Policy*, 19(2), 193-208.
- Engle, R. F. (1974). Band spectrum regression. International Economic Review, 1-11.
- Estrin, S., Hanousek, J., Kočenda, E., & Svejnar, J. (2009). The effects of privatization and ownership in transition economies. *Journal of Economic Literature*, 699-728.

- Evans, P. B. (1995). *Embedded autonomy: states and industrial transformation* (Vol. 25): Cambridge Univ Press.
- Faccio, M. (2006). Politically connected firms. The American economic review, 369-386.
- Fan, S., Kanbur, R., Wei, S.-J., & Zhang, X. (2014). *The Oxford Companion to the Economics of China*: Oxford University Press.
- Feenstra, R. C., Inklaar, R., & Timmer, M. (2013). The next generation of the Penn World Table. *Forthcoming American Economic Review*
- Fisher-Vanden, K., Jefferson, G. H., Liu, H., & Tao, Q. (2004). What is driving China's decline in energy intensity? *Resource and Energy Economics*, 26(1), 77-97.
- Frydman, R., Gray, C., Hessel, M., & Rapaczynski, A. (1999). When does privatization work? The impact of private ownership on corporate performance in the transition economies. *Quarterly journal of economics*, 1153-1191.
- Gerdes, C. (2010). Using Shares' vs.'Log of Shares' in Fixed-Effect Estimations.
- Grout, P. A., & Stevens, M. (2003). The assessment: financing and managing public services. *Oxford Review of Economic Policy*, 19(2), 215-234.
- Hanmin, L., & Xingxiu, W. (2011). The Nature, Features and Governance of Stateowned Energy Enterprises. *Energy Proceedia*, 5, 713-718.
- He, G., Lu, Y., Mol, A. P., & Beckers, T. (2012). Changes and challenges: China's environmental management in transition. *Environmental Development*, *3*, 25-38.
- Hellman, J. S., Jones, G., & Kaufmann, D. (2003). Seize the state, seize the day: state capture and influence in transition economies. *Journal of Comparative Economics*, 31(4), 751-773.
- Hu, Y., Opper, S., & Sonia, M. (2006). Political economy of labor retrenchment: Evidence based on China's state-owned enterprises. *China Economic Review*, 17(3), 281-299.
- Jiang, L., Lin, C., & Lin, P. (2014). The determinants of pollution levels: Firm-level evidence from Chinese manufacturing. *Journal of Comparative Economics*, 42(1), 118-142.
- Jones, D. W. (1991). How urbanization affects energy-use in developing countries. *Energy Policy*, 19(7), 621-630.
- Knight, J. (2014). Economic Causes and Cures of Social Instability in China. China & World Economy, 22(2), 5-21.
- Kornai, J. (1980). Economics of shortage (Vol. 2): North Holland.
- Landry, P. F. (2005). The political management of mayors in post-Deng China. *The Copenhagen Journal of Asian Studies*, 17, 31-58.
- Lardy, N. R. (2014). *Markets Over Mao: The Rise of Private Business in China*: Peterson Institute for International Economics.

- Lee, J. (2009). State owned enterprises in China: Reviewing the evidence. *OECD* Occasional Paper, 6-7.
- Li, D. D. (1998). Changing incentives of the Chinese bureaucracy. *American Economic Review*, 393-397.
- Lin, B., & Jiang, Z. (2011). Estimates of energy subsidies in China and impact of energy subsidy reform. *Energy Economics*, 33(2), 273-283.
- Lise, W. (2006). Decomposition of CO<sub>2</sub> emissions over 1980–2003 in Turkey. *Energy Policy*, *34*(14), 1841-1852.
- Liu, H., & Wang, X. (2011). The Nature, Features and Governance of State-owned Energy Enterprises. *Energy Procedia*, *5*, 713-718
- Lo, A. Y. (2012). Carbon emissions trading in China. *Nature Climate Change*, 2(11), 765-766.
- Ma, D., & Parish, W. L. (2006). Tocquevillian moments: Charitable contributions by Chinese private entrepreneurs. *Social Forces*, *85*(2), 943-964.
- Megginson, W. L., & Netter, J. M. (2001). From state to market: A survey of empirical studies on privatization. *Journal of Economic Literature*, 321-389.
- Mol, A. P., & Carter, N. T. (2006). China's environmental governance in transition. *Environmental politics*, 15(02), 149-170.
- Montinola, G., Qian, Y., & Weingast, B. R. (1995). Federalism, Chinese style: the political basis for economic success in China. *World Politics*, 48(01), 50-81.
- Nee, V., & Opper, S. (2010). Political capital in a market economy. *Social Forces, 88*(5), 2105-2132.
- Nee, V., & Opper, S. (2012). *Capitalism from below: Markets and institutional change in China:* Harvard University Press.
- Nee, V., & Opper, S. (2015). Network effects, cooperation and entrepreneurial innovation in China. *Asian Business & Management*, 14(4), 283-302.
- Nee, V., Opper, S., & Wong, S. (2007). Developmental state and corporate governance in China. *Management and Organization Review*, *3*(1), 19-53.
- Percival, D. B., & Walden, A. T. (2006). *Wavelet methods for time series analysis* (Vol. 4): Cambridge University Press.
- Ramsey, J. B., & Lampart, C. (1998). Decomposition of economic relationships by timescale using wavelets. *Macroeconomic dynamics*, 2(01), 49-71.
- Rodrigue, J.-P., Comtois, C., & Slack, B. (2013). *The geography of transport systems*: Routledge.
- Rodrigue, J. P. (2006). Transportation and the geographical and functional integration of global production networks. *Growth and Change*, *37*(4), 510-525.

- Schmid, A., & Robin, D. (1995). Focus on Environment: Brazil Emerges as Latin American's New Environmental Hotspot. *ENR*, 234(21), 42-43.
- Shirley, M. M., Kikeri, S., & Nellis, J. (1992). Privatization: the Lessons from experience. *World Bank Publication*, 11104.
- Shleifer, A. (1998). State versus private ownership. *The Journal of Economic Perspectives*, 12(4), 133.
- Shleifer, A., & Vishny, R. W. (1994). Politicians and firms. *The Quarterly Journal of Economics*, 995-1025.
- Su, D., & He, X. (2012). Ownership structure, corporate governance and productive efficiency in China. *Journal of productivity analysis, 38*(3), 303-318.
- Talukdar, D., & Meisner, C. M. (2001). Does the private sector help or hurt the environment? Evidence from carbon dioxide pollution in developing countries. *World Development*, 29(5), 827-840.
- Taube, M. (2003). The Role of Host Country Private Sector Enterprises in the Realization of Positive Externalities from FDI-Inflows. OECDCUTS Roundtable on Foreign Direct Investment in Transition Economies: Challenges, Policies and Good Practices. Istanbul.
- Taube, M. (2005). *The Chinese economy: How much market-how much state?* Paper presented at the CESifo Forum.
- Taube, M. (2013). Relational corruption in the PR China. Institutional foundations and its (Dys)-functionality for economic development and growth. *Zeitschrift für Vergleichende Politikwissenschaft*, 7(1), 89-116.
- Tol, R. S. (2007). Carbon dioxide emission scenarios for the USA. *Energy Policy*, 35(11), 5310-5326.
- Tol, R. S., Pacala, S. W., & Socolow, R. H. (2009). Understanding long-term energy use and carbon dioxide emissions in the USA. *Journal of Policy Modeling*, *31*(3), 425-445.
- Tsai, K. S. (2004). *Back-alley banking: Private entrepreneurs in China*: Cornell University Press.
- Wang, H., & Jin, Y. (2007). Industrial ownership and environmental performance: evidence from China. *Environmental and Resource Economics*, 36(3), 255-273.
- Wang, H., Mamingi, N., Laplante, B., & Dasgupta, S. (2003). Incomplete enforcement of pollution regulation: Bargaining power of Chinese factories. *Environmental and Resource Economics*, 24(3), 245-262.
- Wang, H., & Wheeler, D. (2003). Equilibrium pollution and economic development in China. *Environment and Development Economics*, 8(03), 451-466.

Wang, H., & Wheeler, D. (2005). Financial incentives and endogenous enforcement in China's pollution levy system. *Journal of Environmental Economics and Management*, 49(1), 174-196.

World Bank. (2012). China - Enterprise Survey.

Wu, Y. (2009). China's capital stock series by region and sector: Citeseer.

- Xin, K. K., & Pearce, J. L. (1996). Guanxi: Connections as substitutes for formal institutional support. *Academy of management journal*, *39*(6), 1641-1658.
- Yuan, J., Xu, Y., Zhang, X., Hu, Z., & Xu, M. (2014). China's 2020 clean energy target: Consistency, pathways and policy implications. *Energy Policy, 65*, 692-700.
- Zhao, X., & Burnett, J. (2013). Forecasting province-level CO<sub>2</sub> emissions in China. *Letters in Spatial and Resource Sciences*.

## Appendix

### **Appendix A: Total Factor Productivity Estimation**

Total factor productivity is estimated using a Cobb-Douglas production function assuming constant returns to scale:

$$Y_{it} = A_{it} L_{it}^{1-\beta} K_{it}^{\beta}, \qquad (A1)$$

where  $Y_{it}$  is real GDP for province *i* at time point *t*,  $A_{it}$  is technology, *L* is employment, and *K* is real capital. Dividing by A1 by *K*, taking the log and first difference, and distinguishing between short and long-run effects provides the following regression model:

$$\Delta y_{it} = \alpha_{it} + \beta_{SR} l_{it}^{SR} + \beta_{LR} l_{it}^{LR} + f_t + e_{it}, \quad (A2)$$

where  $y = log\left(\frac{Y}{K}\right) l = log\left(\frac{L}{K}\right)$ , and  $\beta_{SR}$  and  $\beta_{LR}$  are the short-run and longrun labor elasticities, respectively. By estimating A2, we obtain estimates of the labor elasticities from which we can estimate TFP as the Solow residual:

$$\Delta \widehat{TFP}_{it} = \Delta y_{it} - \hat{\alpha}_{it} + \hat{\beta}_{LR} l_{it}^{LR} + \hat{\beta}_{SR} l_{it}^{SR}.$$
 (A3)

Estimates of the parameters are given in Table A2.1. In the short-run, all variations in output linked to capital and labor are due to variations in labor. This result is expected because it takes time for firms to adjust the capital stock, and a temporary variation in demand is more likely to be met by a change in employment rather than a change in the capital stock. Over the long term, however, labor elasticity is 0.55 and capital elasticity is 0.45.

| Variable                           | Parameter estimate |  |
|------------------------------------|--------------------|--|
| l <sub>it</sub> <sup>SR</sup>      | 1.17***            |  |
|                                    | (0.03)             |  |
| 10                                 | 0.48***            |  |
| $l_{it}^{LR}$                      | (0.05)             |  |
| Short –run Adjusted R <sup>2</sup> | 0.12               |  |
| Long run Adjusted R <sup>2</sup>   | 0.80               |  |

Table A2.1Regression results from growth regression

| Variable name      | Description  | Data source   |
|--------------------|--|---|
| CO <sub>2</sub>    | Provincial CO2 emissions<br>constructed using the IPCC<br>guidelines | Zhao and Burnett (2014)   |
| Real GDP           | Real GDP in 2009 prices  | China Data Online   |
| GDP deflator       | GDP deflator   | China Data Online   |
| Employment         | Number of persons employed   | China Data Online   |
| Private employment | Number of persons employed by private enterprises                    | China Statistical Yearbook,<br>various years, Blue book of<br>China's private enterprises |
| Capital            | Real capital stock in 1978 prices                                    | Wu (2009), and own estimations<br>using investment data from Chin<br>Data Online          |
| Manufacturing      | Manufacturing sector (secondary sector) share of GDP                 | China Data Online   |
| Service            | Service sector (tertiary sector) share of GDP                        | China Data Online   |
| РКМ                | Total number of personal km  | China Energy Databook, version 8.0, 2013  |
| ТКМ                | Total number of ton-km   | China Energy Databook, version 8.0, 2013  |
| Coal               | Real coil price deflated using the provincial GDP deflator           | China Energy Databook, version 8.0, 2013  |
| Oil                | Real oil prices deflated using the provincial GDP deflator           | China Energy Databook, version 8.0, 2013  |

# Appendix B: Variable Description and Data Sources

### Lund Economic Studies

| 1. Guy Arvidsson                                | Bidrag till teorin för verkningarna av räntevariationer, 1962   |
|---|---|
| 2. Björn Thalberg                               | A Trade Cycle Analysis. Extensions of the Goodwin Model, 1966   |
| 3. Bengt Höglund                                | Modell och observationer. En studie av empirisk anknytning och aggregation för en linjär produktionsmodell, 1968                        |
| 4. Alf Carling                                  | Industrins struktur och konkurrensförhållanden, 1968  |
| 5. Tony Hagström                                | Kreditmarknadens struktur och funktionssätt, 1968   |
| 6. Göran Skogh                                  | Straffrätt och samhällsekonomi, 1973  |
| 7. Ulf Jakobsson och Göran<br>Norman            | Inkomstbeskattningen i den ekonomiska politiken. En<br>kvantitativ analys av systemet för personlig<br>inkomstbeskattning 1952-71, 1974 |
| 8. Eskil Wadensjö                               | Immigration och samhällsekonomi. Immigrationens ekono-<br>miska orsaker och effekter, 1973  |
| 9. Rögnvaldur Hannesson                         | Economics of Fisheries. Some Problems of Efficiency, 1974   |
| 10. Charles Stuart                              | Search and the Organization of Marketplaces, 1975   |
| 11. S Enone Metuge                              | An Input-Output Study of the Structure and Resource Use in the Cameroon Economy, 1976   |
| 12. Bengt Jönsson                               | Cost-Benefit Analysis in Public Health and Medical Care, 1976   |
| 13. Agneta Kruse och<br>Ann-Charlotte Ståhlberg | Effekter av ATP - en samhällsekonomisk studie, 1977   |
| 14. Krister Hjalte                              | Sjörestaureringens ekonomi, 1977  |
| 15. Lars-Gunnar Svensson                        | Social Justice and Fair Distributions, 1977   |
| 16. Curt Wells                                  | Optimal Fiscal and Monetary Policy - Experiments with an Econometric Model of Sweden, 1978  |
| 17. Karl Lidgren                                | Dryckesförpackningar och miljöpolitik - En studie av styr-<br>medel, 1978   |
| 18. Mats Lundahl                                | Peasants and Poverty. A Study of Haiti, London, 1979  |
| 19. Inga Persson-Tanimura                       | Studier kring arbetsmarknad och information, 1980   |
| 20. Bengt Turner                                | Hyressättning på bostadsmarknaden - Från hyresreglering till bruksvärdesprövning, Stockholm 1979  |
| 21. Ingemar Hansson                             | Market Adjustment and Investment Determination. A Theo-<br>retical Analysis of the Firm and the Industry, Stockholm<br>1981             |
| 22. Daniel Boda Ndlela                          | Dualism in the Rhodesian Colonial Economy, 1981   |

| 23. Tom Alberts                        | Agrarian Reform and Rural Poverty: A Case Study of Peru, 1981   |
|--|---|
| 24. Björn Lindgren                     | Costs of Illness in Sweden 1964-75, 1981  |
| 25. Göte Hansson                       | Social Clauses and International Trade. An Economic Analysis of Labour Standards in Trade Policy, 1981  |
| 26. Noman Kanafani                     | Oil and Development. A Case Study of Iraq, 1982   |
| 27. Jan Ekberg                         | Inkomsteffekter av invandring, 1983   |
| 28. Stefan Hedlund                     | Crisis in Soviet Agriculture?, 1983   |
| 29. Ann-Marie Pålsson                  | Hushållen och kreditpolitiken. En studie av kreditrestrik-<br>tioners effekt på hushållens konsumtion, sparande och<br>konsumtionsmönster, 1983 |
| 30. Lennart Petersson                  | Svensk utrikeshandel, 1871-1980. En studie i den intraindustriella handelns framväxt, 1984  |
| 31. Bengt Assarsson                    | Inflation and Relative Prices in an Open Economy, 1984  |
| 32. Claudio Vedovato                   | Politics, Foreign Trade and Economic Development in the Dominican Republic, 1985  |
| 33. Knut Ödegaard                      | Cash Crop versus Food Crop Production in Tanzania: An<br>Assessment of the Major Post-Colonial Trends, 1985                                     |
| 34. Vassilios Vlachos                  | Temporära lönesubventioner. En studie av ett arbetsmarknadspolitiskt medel, 1985  |
| 35. Stig Tegle                         | Part-Time Employment. An Economic Analysis of Weekly<br>Working Hours in Sweden 1963-1982, 1985   |
| 36. Peter Stenkula                     | Tre studier över resursanvändningen i högskolan, 1985   |
| 37. Carl Hampus Lyttkens               | Swedish Work Environment Policy. An Economic Analysis, 1985   |
| 38. Per-Olof Bjuggren                  | A Transaction Cost Approach to Vertical Integration: The Case of Swedish Pulp and Paper Industry, 1985  |
| 39. Jan Petersson                      | Erik Lindahl och Stockholmsskolans dynamiska metod, 1987  |
| 40. Yves Bourdet                       | International Integration, Market Structure and Prices. A<br>Case Study of the West-European Passenger Car Industry,<br>1987                    |
| 41. Krister Andersson and Erik Norrman | Capital Taxation and Neutrality. A study of tax wedges with special reference to Sweden, 1987   |
| 42. Tohmas Karlsson                    | A Macroeconomic Disequilibrium Model. An Econometric<br>Study of the Swedish Business Sector 1970-84, 1987                                      |
| 43. Rosemary Vargas-<br>Lundius        | Peasants in Distress. Poverty and Unemployment in the Dominican Republic, 1989  |

| 44. Lena Ekelund Axelson    | Structural Changes in the Swedish Marketing of Vegetables, 1991  |
|-----------------------------|--|
| 45. Elias Kazarian          | Finance and Economic Development: Islamic Banking in Egypt, 1991   |
| 46. Anders Danielson        | Public Sector Expansion and Economic Development. The<br>Sources and Consequences of Development Finance in<br>Jamaica 1962-84, 1991 |
| 47. Johan Torstensson       | Factor Endowments, Product Differentiation, and Inter-<br>national Trade, 1992   |
| 48. Tarmo Haavisto          | Money and Economic Activity in Finland, 1866-1985, 1992  |
| 49. Ulf Grönkvist           | Economic Methodology. Patterns of Reasoning and the Structure of Theories, 1992  |
| 50. Evelyne Hangali Maje    | Monetization, Financial Development and the Demand for Money, 1992   |
| 51. Michael Bergman         | Essays on Economic Fluctuations, 1992  |
| 52. Flora Mndeme<br>Musonda | Development Strategy and Manufactured Exports in Tanzania, 1992  |
| 53. Håkan J. Holm           | Complexity in Economic Theory. An Automata Theoretical Approach, 1993  |
| 54. Klas Fregert            | Wage Contracts, Policy Regimes and Business Cycles. A Contractual History of Sweden 1908-90, 1994                                    |
| 55. Per Frennberg           | Essays on Stock Price Behaviour in Sweden, 1994  |
| 56. Lisbeth Hellvin         | Trade and Specialization in Asia, 1994   |
| 57. Sören Höjgård           | Long-term Unemployment in a Full Employment Economy, 1994  |
| 58. Karolina Ekholm         | Multinational Production and Trade in Technological<br>Knowledge, 1995   |
| 59. Fredrik Andersson       | Essays in the Economics of Asymmetric Information, 1995  |
| 60. Rikard Althin           | Essays on the Measurement of Producer Performance, 1995  |
| 61. Lars Nordén             | Empirical Studies of the Market Microstructure on the Swedish Stock Exchange, 1996   |
| 62. Kristian Bolin          | An Economic Analysis of Marriage and Divorce, 1996   |
| 63. Fredrik Sjöholm         | R&D, International Spillovers and Productivity Growth, 1997  |
| 64. Hossein Asgharian       | Essays on Capital Structure, 1997  |
| 65. Hans Falck              | Aid and Economic Performance - The Case of Tanzania, 1997  |
| 66. Bengt Liljas            | The Demand for Health and the Contingent Valuation   |

|                             | Method, 1997   |
|-----------------------------|--|
| 67. Lars Pålsson Syll       | Utility Theory and Structural Analysis, 1997   |
| 68. Richard Henricsson      | Time Varying Parameters in Exchange Rate Models, 1997  |
| 69. Peter Hördahl           | Financial Volatility and Time-Varying Risk Premia, 1997  |
| 70. Lars Nilsson            | Essays on North-South Trade, 1997  |
| 71. Fredrik Berggren        | Essays on the Demand for Alcohol in Sweden - Review and Applied Demand Studies, 1998                     |
| 72. Henrik Braconier        | Essays on R&D, Technology and Growth, 1998   |
| 73. Jerker Lundbäck         | Essays on Trade, Growth and Exchange Rates, 1998   |
| 74. Dan Anderberg           | Essays on Pensions and Information, 1998   |
| 75. P. Göran T. Hägg        | An Institutional Analysis of Insurance Regulation – The Case of Sweden, 1998                             |
| 76. Hans-Peter Bermin       | Essays on Lookback and Barrier Options - A Malliavin Calculus Approach, 1998                             |
| 77. Kristian Nilsson        | Essays on Exchange Rates, Exports and Growth in Developing Countries, 1998                               |
| 78. Peter Jochumzen         | Essays on Econometric Theory, 1998   |
| 79. Lars Behrenz            | Essays on the Employment Service and Employers'<br>Recruitment Behaviour, 1998                           |
| 80. Paul Nystedt            | Economic Aspects of Ageing, 1998   |
| 81. Rasha M. Torstensson    | Empirical Studies in Trade, Integration and Growth, 1999   |
| 82. Mattias Ganslandt       | Games and Markets – Essays on Communication,<br>Coordination and Multi-Market Competition, 1999          |
| 83. Carl-Johan Belfrage     | Essays on Interest Groups and Trade Policy, 1999   |
| 84. Dan-Olof Rooth          | Refugee Immigrants in Sweden - Educational Investments<br>and Labour Market Integration, 1999            |
| 85. Karin Olofsdotter       | Market Structure and Integration: Essays on Trade,<br>Specialisation and Foreign Direct Investment, 1999 |
| 86. Katarina Steen Carlsson | Equality of Access in Health Care, 1999  |
| 87. Peter Martinsson        | Stated preference methods and empirical analyses of equity in health, 2000                               |
| 88. Klas Bergenheim         | Essays on Pharmaceutical R&D, 2000   |
| 89. Hanna Norberg           | Empirical Essays on Regional Specialization and Trade in Sweden, 2000                                    |
| 90. Åsa Hansson             | Limits of Tax Policy, 2000   |
| 91. Hans Byström            | Essays on Financial Markets, 2000  |

| 92. Henrik Amilon  | Essays on Financial Models, 2000   |
|--|--|
| 93. Mattias Lundbäck   | Asymmetric Information and The Production of Health, 2000  |
| 94. Jesper Hansson   | Macroeconometric Studies of Private Consumption,<br>Government Debt and Real Exchange Rates, 2001  |
| 95. Jonas Månsson  | Essays on: Application of Cross Sectional Efficiency<br>Analysis, 2001   |
| 96. Mattias Persson  | Portfolio Selection and the Analysis of Risk and Time Diversification, 2001  |
| 97. Pontus Hansson   | Economic Growth and Fiscal Policy, 2002  |
| 98. Joakim Gullstrand  | Splitting and Measuring Intra-Industry Trade, 2002   |
| 99. Birger Nilsson   | International Asset Pricing, Diversification and Links<br>between National Stock Markets, 2002   |
| 100. Andreas Graflund  | Financial Applications of Markov Chain Monte Carlo Methods, 2002   |
| 101. Therése Hindman<br>Persson  | Economic Analyses of Drinking Water and Sanitation in Developing Countries, 2002   |
| 102. Göran Hjelm   | Macroeconomic Studies on Fiscal Policy and Real<br>Exchange Rates, 2002  |
| 103. Klas Rikner   | Sickness Insurance: Design and Behavior, 2002  |
| 104. Thomas Ericson  | Essays on the Acquisition of Skills in Teams, 2002   |
| 105. Thomas Elger  | Empirical Studies on the Demand for Monetary Services in the UK, 2002  |
| 106. Helena Johansson  | International Competition, Productivity and Regional Spillovers, 2003  |
|  | - F  |
| 107. Fredrik Gallo   | Explorations in the New Economic Geography, 2003   |
| 107. Fredrik Gallo<br>108. Susanna Thede   | • ·  |
|  | Explorations in the New Economic Geography, 2003   |
| 108. Susanna Thede   | Explorations in the New Economic Geography, 2003<br>Essays on Endogenous Trade Policies, 2003<br>Interest Groups and Government Policy, A Political  |
| 108. Susanna Thede<br>109. Fredrik CA Andersson  | Explorations in the New Economic Geography, 2003<br>Essays on Endogenous Trade Policies, 2003<br>Interest Groups and Government Policy, A Political<br>Economy Analysis, 2003  |
| <ul><li>108. Susanna Thede</li><li>109. Fredrik CA Andersson</li><li>110. Petter Lundborg</li></ul>  | Explorations in the New Economic Geography, 2003<br>Essays on Endogenous Trade Policies, 2003<br>Interest Groups and Government Policy, A Political<br>Economy Analysis, 2003<br>Risky Health Behaviour among Adolescents, 2003  |
| <ul><li>108. Susanna Thede</li><li>109. Fredrik CA Andersson</li><li>110. Petter Lundborg</li><li>111. Martin W Johansson</li></ul>                              | Explorations in the New Economic Geography, 2003<br>Essays on Endogenous Trade Policies, 2003<br>Interest Groups and Government Policy, A Political<br>Economy Analysis, 2003<br>Risky Health Behaviour among Adolescents, 2003<br>Essays on Empirical Macroeconomics, 2003  |
| <ul><li>108. Susanna Thede</li><li>109. Fredrik CA Andersson</li><li>110. Petter Lundborg</li><li>111. Martin W Johansson</li><li>112. Joakim Ekstrand</li></ul> | Explorations in the New Economic Geography, 2003<br>Essays on Endogenous Trade Policies, 2003<br>Interest Groups and Government Policy, A Political<br>Economy Analysis, 2003<br>Risky Health Behaviour among Adolescents, 2003<br>Essays on Empirical Macroeconomics, 2003<br>Currency Markets - Equilibrium and Expectations, 2003<br>Central bank power: a matter of coordination rather than |

| 116. Staffan Waldo                  | Efficiency in Education - A Multilevel Analysis, 2003   |
|-------------------------------------|---|
| 117. Mikael Stenkula                | Essays on Network Effects and Money, 2004   |
| 118. Catharina Hjortsberg           | Health care utilisation in a developing country -The case of Zambia, 2004   |
| 119. Henrik Degrér                  | Empirical Essays on Financial Economics, 2004   |
| 120. Mårten Wallette                | Temporary Jobs in Sweden: Incidence, Exit, and On-the-Job<br>Training, 2004   |
| 121. Tommy Andersson                | Essays on Nonlinear Pricing and Welfare, 2004   |
| 122. Kristian Sundström             | Moral Hazard and Insurance: Optimality, Risk and Preferences, 2004  |
| 123. Pär Torstensson                | Essays on Bargaining and Social Choice, 2004  |
| 124. Frederik Lundtofte             | Essays on Incomplete Information in Financial Markets, 2005   |
| 125. Kristian Jönsson               | Essays on Fiscal Policy, Private Consumption and Non-<br>Stationary Panel Data, 2005  |
| 126. Henrik Andersson               | Willingness to Pay for a Reduction in Road Mortality Risk:<br>Evidence from Sweden, 2005  |
| 127. Björn Ekman                    | Essays on International Health Economics: The Role of<br>Health Insurance in Health Care Financing in Low- and<br>Middle-Income Countries, 2005 |
| 128. Ulf G Erlandsson               | Markov Regime Switching in Economic Time Series, 2005   |
| 129. Joakim Westerlund              | Essays on Panel Cointegration, 2005   |
| 130. Lena Hiselius                  | External costs of transports imposed on neighbours and fellow road users, 2005  |
| 131. Ludvig Söderling               | Essays on African Growth, Productivity, and Trade, 2005   |
| 132. Åsa Eriksson                   | Testing and Applying Cointegration Analysis in Macroeconomics, 2005   |
| 133. Fredrik Hansen                 | Explorations in Behavioral Economics: Realism, Ontology and Experiments, 2006   |
| 134. Fadi Zaher                     | Evaluating Asset-Pricing Models in International Financial<br>Markets, 2006   |
| 135. Christoffer Bengtsson          | Applications of Bayesian Econometrics to Financial Economics, 2006  |
| 136. Alfredo Schclarek<br>Curutchet | Essays on Fiscal Policy, Public Debt and Financial Development, 2006  |
| 137. Fredrik Wilhelmsson            | Trade, Competition and Productivity, 2006   |
| 138. Ola Jönsson                    | Option Pricing and Bayesian Learning, 2007  |
|                                     |   |

| 139. Ola Larsson          | Essays on Risk in International Financial Markets, 2007  |
|---------------------------|--|
| 140. Anna Meyer           | Studies on the Swedish Parental Insurance, 2007  |
| 141. Martin Nordin        | Studies in Human Capital, Ability and Migration, 2007  |
| 142. Bolor Naranhuu       | Studies on Poverty in Mongolia, 2007   |
| 143. Margareta Ekbladh    | Essays on Sickness Insurance, Absence Certification and Social Norms, 2007   |
| 144. Erik Wengström       | Communication in Games and Decision Making under Risk, 2007  |
| 145. Robin Rander         | Essays on Auctions, 2008   |
| 146. Ola Andersson        | Bargaining and Communication in Games, 2008  |
| 147. Marcus Larson        | Essays on Realized Volatility and Jumps, 2008  |
| 148. Per Hjertstrand      | Testing for Rationality, Separability and Efficiency, 2008   |
| 149. Fredrik NG Andersson | Wavelet Analysis of Economic Time Series, 2008   |
| 150. Sonnie Karlsson      | Empirical studies of financial asset returns, 2009   |
| 151. Maria Persson        | From Trade Preferences to Trade Facilitation, 2009   |
| 152. Eric Rehn            | Social Insurance, Organization and Hospital Care, 2009   |
| 153. Peter Karpestam      | Economics of Migration, 2009   |
| 154. Marcus Nossman       | Essays on Stochastic Volatility, 2009  |
| 155. Erik Jonasson        | Labor Markets in Transformation: Case Studies of Latin<br>America, 2009  |
| 156. Karl Larsson         | Analytical Approximation of Contingent Claims, 2009  |
| 157. Therese Nilsson      | Inequality, Globalization and Health, 2009   |
| 158. Rikard Green         | Essays on Financial Risks and Derivatives with<br>Applications to Electricity Markets and Credit Markets,<br>2009  |
| 159. Christian Jörgensen  | Deepening Integration in the Food Industry – Prices,<br>Productivity and Export, 2010                              |
| 160. Wolfgang Hess        | The Analysis of Duration and Panel Data in Economics, 2010   |
| 161. Pernilla Johansson   | From debt crisis to debt relief: A study of debt determinants, aid composition and debt relief effectiveness, 2010 |
| 162. Nils Janlöv          | Measuring Efficiency in the<br>Swedish Health Care Sector, 2010  |

| 163. Ai Jun Hou            | Essays on Financial Markets Volatility, 2011   |
|----------------------------|--|
| 164. Alexander Reffgen     | Essays on Strategy-proof Social Choice, 2011   |
| 165. Johan Blomquist       | Testing homogeneity and unit root restrictions in panels, 2012   |
| 166. Karin Bergman         | The Organization of R&D - Sourcing Strategy, Financing and Relation to Trade, 2012   |
| 167. Lu Liu                | Essays on Financial Market Interdependence, 2012   |
| 168. Bujar Huskaj          | Essays on VIX Futures and Options, 2012  |
| 169. Åsa Ljungvall         | Economic perspectives on the obesity epidemic, 2012  |
| 170. Emma Svensson         | Experimenting with Focal Points and<br>Monetary Policy, 2012   |
| 171. Jens Dietrichson      | Designing Public Organizations and Institutions: Essays on<br>Coordination and Incentives, 2013                                      |
| 172. Thomas Eriksson       | Empirical Essays on Health and Human Capital, 2013   |
| 173. Lina Maria Ellegård   | Political Conflicts over Public Policy in Local<br>Governments, 2013   |
| 174. Andreas Hatzigeorgiou | Information, Network and Trust in the Global Economy – Essays on International Trade and Migration, 2013                             |
| 175. Gustav Kjellsson      | Inequality, Health, and Smoking, 2014  |
| 176. Richard Desjardins    | Rewards to skill supply, skill demand and skill match-<br>mismatch – Studies using the Adult Literacy and Lifeskills<br>survey, 2014 |
| 177. Viroj                 | What Drives Exports?   |
| Jienwatcharamongkhol       | - An Empirical Evidence at the Firm Level, 2014  |
| 178. Anton Nilsson         | Health, Skills and Labor Market Success, 2014  |
| 179. Albin Erlanson        | Essays on Mechanism Design, 2014   |
| 180. Daniel Ekeblom        | Essays in Empirical Expectations, 2014   |
| 181. Sofie Gustafsson      | Essays on Human Capital Investments:<br>Pharmaceuticals and Education, 2014  |
| 182. Katarzyna Burzynska   | Essays on Finance, Networks and Institutions, 2015   |

| 183. Mingfa Ding<br>184. Anna Andersson | Corporate Ownership and Liquidity in<br>China's Stock Markets, 2015<br>Vertical Trade, 2015 |
|---|---|
| 185. Cecilia Hammarlund                 | Fish and Trips in the Baltic Sea – Prices, Management and Labor Supply, 2015                |
| 186. Hilda Ralsmark                     | Family, Friend, or Foe?<br>Essays in Empirical Microeconomics, 2015                         |
| 187. Jens Gudmundsson                   | Making Pairs, 2015  |
| 188. Emanuel Alfranseder                | Essays on Financial Risks and the Subprime Crisis, 2015                                     |
| 189. Ida Lovén                          | Education, Health, and Earnings – Type 1 Diabetes in Children and Young Adults, 2015        |
| 190. Caren Yinxia Nielsen               | Essays on Credit Risk, 2015   |
| 191. Usman Khalid                       | Essays on Institutions and Institutional change, 2016                                       |



LUND UNIVERSITY